

Integro-differential models from ecology and evolutionary biology

Sepideh Mirrahimi

▶ To cite this version:

Sepideh Mirrahimi. Integro-differential models from ecology and evolutionary biology. Analysis of PDEs [math.AP]. Université Paul Sabatier (Toulouse 3), 2019. tel-02448102

HAL Id: tel-02448102 https://hal.science/tel-02448102

Submitted on 22 Jan 2020

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Manuscrit

présenté pour l'obtention de

L'HABILITATION À DIRIGER DES RECHERCHES

par

Sepideh MIRRAHIMI

de l'Institut de Mathématiques de Toulouse

Soutenue le 04 Décembre 2019, devant le jury composé de :

Guy Barles	Université de Tours	Examinateur
Henri Berestycki	EHESS	Examinateur
Odo Diekmann	Université d'Utrecht	Rapporteur
Pierre-Emmanuel Jabin	Université du Maryland	Rapporteur
Sylvie Méléard	Ecole Polytechnique	Examinatrice
Benoit Perthame	Sorbonne Université	Examinateur
Ophélie Ronce	CNRS & Université de Montpellier	Examinatrice
Jean-Michel Roquejoffre	Université Paul Sabatier	Parrain
Lenya Ryzhik	Université Stanford	Rapporteur

Remerciements

C'est un grand plaisir pour moi de remercier les personnes qui m'ont accompagnée tout au long de mon parcours scientifique.

Mes premiers mots iront à Jean-Michel Roquejoffre, ne seulement pour avoir accepté de parrainer cette habilitation, mais aussi pour son soutien, ses conseils constructifs et surtout sa bienveillance constante depuis mon arrivée à Toulouse. J'ai beaucoup appris et continue à apprendre à son contact.

Je souhaiterais remercier également Guy Barles et Benoît Perthame, mes anciens directeurs de thèse pour m'avoir fait découvrir le monde de la recherche et pour leur soutien et leur disponibilité pour répondre à toutes les questions que j'ai pu leur poser depuis ces onze dernières années.

Mes sincères remerciements vont à Odo Diekmann, Pierre-Emmanuel Jabin et Lenya Ryzhik pour m'avoir fait l'honneur de rapporter mon manuscrit. Je suis très reconnaissante de leur relecture approfondie et de l'intérêt qu'ils ont porté à mes travaux.

Merci également à Sylvie Méléard d'avoir accepté d'être membre de mon jury mais aussi pour m'avoir accueillie, avec Stéphane Gaubert, en tant que postdoctorante au CMAP. Cette année de postdoctorat a été une source d'inspiration pour moi et a été à l'origine d'un axe important de ma recherche par la suite.

Je remercie chaleureusement Henri Berestycki pour l'intérêt qu'il a porté à mon travail et pour pour avoir accepté d'être membres de mon jury. Mes sincères remerciements vont également à et Ophélie Ronce qui a accepté de jouer le jeu de l'interdisciplinarité. Leur présence me touche et m'honore.

De nombreuse collaborations et échanges avec des chercheurs en mathématiques et en biologie ont enrichi mon travail de recherche. Je remercie tous mes collaborateurs en espérant de n'oublier personne : Guy Barles, Emeric Bouin, Vincent Calvez, Manon Costa, Sylvain Gandon, Chris Henderson, Hélène Leman, Alexander Lorz, Sylvie Méléard, Benoît Perthame, Gaël Raoul, Jean-Michel Roquejoffre, Takis Souganidis, Olga Turanova et Joe-Yuichiro Wakano. Un projet de collaboration avec Sylvain Gandon sur l'application des méthodes que j'ai développé en biologie évolutive, a été particulièrement marquant dans mon travail de recherche et a eu un grand impact sur mes choix de directions de recherche.

Merci égalements à Susely Figueroa Iglesias, Alexis Léculier et Léonard Dekens, pour m'avoir accordé leur confiance pour encadrer ou co-encadrer leurs travaux de thèse. Merci de même à Cristobal Quininao et à Christèle Etchegaray pour leurs travaux en tant que postdoctorants.

J'estime que j'ai une grande chance de travailler dans un environnement scientifique très riche et agréable. Un grand merci à mes collègues de l'institut de mathématiques de Toulouse (un merci particulier à ma collègue de bureau Manon Costa), pour les discussions et les échanges scientifiques et non-scientifiques et les moments agréables partagés. Ces excellentes conditions sont aussi garanties par la qualité et le dévouement du personnel d'appui à la recherche et je tiens particulièrement à remercier Janani Chandran pour son travail précieux. Merci également à Sophia Abir qui s'est occupe de l'organisation de ma soutenance de HDR.

Mes derniers remerciement vont à mes amis et ma famille. Merci à mes amis pour leur soutien et pour tous les moments agréables partagés avec eux. Votre présence est très précieuse. Merci à mes parents pour leur soutien inconditionnel. Et merci à Pascal pour sa patience, son soutien et son encouragement tout au long de ces années. Et une pensée pour Soren que s'il n'a pas beaucoup aidé à la sortie de ce mémoire, il embellit chaque jour ma vie.

Contents

G	eneral presentation	vii
Li	ist of publications	xi
1	A Hamilton-Jacobi approach for models from evolutionary biology 1.1 Introduction	1 1 2 5 6 7 9
2	Fractional reaction-diffusion equations 2.1 Introduction	13 13 14 18 24
3	Evolutionary adaptation in space-heterogeneous environments3.1Introduction	 29 29 32 36 38 42
4	 Evolutionary adaptation in time-heterogeneous environments 4.1 Introduction	49 49 51 58 70

Perspectives

 $\mathbf{75}$

General presentation

In this dissertation, I will present a large part of the research I did after my PhD on elliptic and parabolic integro-differential equations and Hamilton-Jacobi equations coming from ecology and evolutionary biology. This report is composed of four chapters. The first two chapters are devoted to the development of the basis of an asymptotic method based on Hamilton-Jacobi equations to study models from evolutionary biology and ecology. The last two chapters concern the development and the application of the method in the case of heterogeneous environments. At the end of the manuscript I will present comments and open questions for future research.

In Chapter 1, I will present the works [LMP11, MR15b, MR16] which concern the well-posedness of the Hamilton-Jacobi approach for a typical selection-mutation model. This approach has been introduced in [DJMP05] and has been developed during the last decade to study models from evolutionary biology (see [BP07, PB08] for the first results within this framework). A main part of my PhD thesis has also been devoted to the development of the basis of this approach [Mir11]. The main objective in such works is to provide an asymptotic analysis of the phenotypic density of a population which solves a parabolic (or elliptic) integro-differential equation, considering small diffusion term (modeling small effects of mutations). The solution to such equation concentrates, in the limit of vanishing diffusion and in long time, as a sum of Dirac masses corresponding to the dominant traits. The asymptotic analysis of the equation, via a logarithmic transformation of the solution, leads indeed usually to a Hamilton-Jacobi equation with constraint. The description of the phenotypic density is then reduced to the description of the solution to the corresponding Hamilton-Jacobi equation. While the derivation of such Hamilton-Jacobi equations from selection-mutation models is rather well-developed (see for instance [BP07, PB08, BMP09]), the asymptotic description of the phenotypic density, based on the analysis of the Hamilton-Jacobi equation with constraint, is less understood. In Chapter 1, I present a framework where such analysis of the Hamilton-Jacobi equation is provided and precise results on the phenotypic density are obtained. A main contribution in this chapter is the uniqueness of the viscosity solution to the Hamilton-Jacobi equation with constraint derived from a typical selection-mutation model. The uniqueness of the viscosity solution to such equation was an open question since the seminal work of Perthame and Barles [PB08]. In addition to the uniqueness property, within this framework, we show that the solution to the Hamilton-Jacobi equation is indeed smooth and classical and it has a unique maximum point. This allows us to describe the asymptotic phenotypic density as a single Dirac mass which evolves with time. Finally, this analysis leads to error estimates and an asymptotic expansion of the solution providing more quantitative results for biological applications.

In Chapter 2, I will focus on the asymptotic study of fractional reaction-diffusion equations [MM15, Mir, LMR]. A main objective in this chapter is to develop the Hamilton-Jacobi approach in the case of selection-mutation models where the mutation term is modeled by a frac-

tional laplacian. The choice of a fractional laplacian rather than a classical laplacian or an integral kernel with thin tails, allows to take into account large mutation jumps with a high rate. The presence of such large mutations leads to drastically faster dynamics and the previous scalings are not adapted to the study of such problems. Note that a related problem is to provide an asymptotic analysis of the fractional Fisher-KPP equation which leads to propagation phenomena with exponential speed [CR13]. A variant of the asymptotic approach based on Hamilton-Jacobi equations has indeed been used in the 80's to study propagation phenomena in local reaction-diffusion equations [Fre85a, Fre85b, ES89, BES90]. As a first step, we first extend these results to the case of fractional Fisher-KPP equation [MM15]. By introducing an unconventional rescaling we derive indeed, via an asymptotic analysis, the exponential speed of propagation established by Cabré and Roquejoffre [CR13] (see also Section 2.4 where we extend this result to the case of periodically hostile environments in the framework of the PhD project of A. Léculier, a student that I supervise together with J.-M. Roquejoffre). Next, being inspired by such analysis we introduce a rescaling for the fractional selection-mutation model which leads to the concentration of the phenotypic density as a single Dirac mass, similarly to the case of classical selection-mutation models [MM15, Mir]. However, significant difficulties comparing to previous works arise from the fractional laplacian term which lead to considerably different analysis. In particular, the logarithmic transform of the phenotypic density does not converge anymore to a viscosity solution of a Hamilton-Jacobi equation, but to a viscosity supersolution which is minimal in a certain class of functions.

Chapter 3 is devoted to the study of the evolutionary dynamics of a population in a heterogeneous environment with a discrete spatial structure [Mir13, GM17, Mir17, MG]. In this chapter we characterize the phenotypic density of a population at equilibrium between selection, mutation and migration and provide analytic approximations for the moments of the phenotypic distribution. Note that such analytic approximations are closely related to an asymptotic expansion of the solution discussed above, at the end of the paragraph on the first chapter. A main part of the work presented in this chapter has been done in the framework of a collaboration project with S. Gandon who is a biologist. The objective of this project was to show, via the particular model considered in this work, how the Hamilton-Jacobi approach can be used to bring insight into current biological questions. This analysis improves the previous results in the biological literature and allows to consider less restrictive assumptions (see for instance [MCG97, Day00, YG09, DRG13, DYG15]). Furthermore, our work makes a connection between the theory of adaptive dynamics and moment based approaches in quantitative genetics. Adaptive dynamics and quantitate genetics, which focus respectively on the evolution of discrete and continuous characters (as a result, respectively, of rare and frequent mutations), are two theories which are widely used in theoretical evolutionary biology (see for instance [MHM⁺96, B00, Die04, Ric04]). Finally, our method allows to go beyond the Gaussian approximation commonly used in the theory of quantitative genetics.

In Chapter 4, we study the evolutionary dynamics of populations in time varying environments [MPS15, FIM18, FIM, CEM]. While the understanding of the effect of the time variations of the environment, due for instance to seasonal changes, climate change or a time administration of drugs, on the adaptation of populations is an important problem in evolutionary biology [CLM10, PMLJ⁺13, KM14, BPnMGI17], the outcome of such heterogeneities is very rich and depends strongly on the speed of the environmental change, the size of the mutational effects, and on how the variations act on the growth rate of individuals. A main difficulty in the study of such problems comes from the combination of the nonlocal competition term with the time dependency of the growth rate. In chapter 4 we will study several types of time variations of the

environment leading to different types of behaviors: time periodic environments [MPS15, FIM18], shifting and oscillating environments [FIM] and a piecewise constant environment with slow switch [CEM]. Several challenging mathematical questions emerge in the development of the Hamilton-Jacobi approach for these models. We apply our method to several illuminating biological examples leading to different outcomes on the phenotypic density. In particular, we show how our method can be used to provide a possible explanation for the outcome of a biological experiment [KMZ⁺13, FIM18]. The articles [FIM18, FIM] are prepared in the framework of the PhD project of my student S. Figueroa Iglesisas. The result in [CEM] is obtained in the framework of the postdoctoral project of C. Etchegaray that I mentored together with M. Costa.

Some of the results that I have obtained after my PhD will not be addressed in this manuscript to maintain some homogeneity:

- The works [BCM⁺12, LMM15, BM15, MP15, CHM⁺] on evolutionary dynamics of populations in space-heterogeneous environments with continuous space structure. These works are closely related to the presented work in this dissertation. While I believe these results correspond to important contributions in this field, these articles are prepared less as a continuous project but more as independent contributions. This is why I chose not to present them in this dissertation.
- The work in [MPW14] where we derive, with B. Perthame and J. Wakano, a selectionmutation model, known as direct competition model, from a model taking into account the interaction of the population with resources.
- The work in [MS13] where we study, with P. Souganidis, a model for the motion of motor proteins.

List of publications

Here is the list of my publications. Papers marked with \bullet are presented in this dissertation. Papers marked with \diamond are part of my PhD thesis. Those marked with * were completed after my PhD but are not presented in this report.

- [CEM] M. COSTA, C. ETCHEGARAY, AND S. MIRRAHIMI. Survival criterion for a population subject to selection and mutations; application to temporally piecewise constant environments. Preprint.
- [FIM] S. FIGUEROA IGLESIAS AND S. MIRRAHIMI. Selection and mutation in a shifting and fluctuating environment. Preprint.
- [LMR] A. LÉCULIER, S. MIRRAHIMI, AND J.-M. ROQUEJOFFRE. Propagation in a fractional reaction-diffusion equation in a periodically hostile environment. Preprint.
- * [CHM⁺] V. CALVEZ, C. HENDERSON, S. MIRRAHIMI, O. TURANOVA AND T. DUMONT, Non-local competition slows down front acceleration during dispersal evolution, Preprint.
- [Mir] S. MIRRAHIMI, Singular limits for models of selection and mutations with heavy tails, accepted in J. Math. Pures Appl.
- [MG] S. MIRRAHIMI AND S. GANDON, Evolution of specialization in heterogeneous environments: equilibrium between selection, mutation and migration, accepted in Genetics.
- [FIM18] S. FIGUEROA IGLESIAS AND S. MIRRAHIMI, Long time evolutionary dynamics of phenotypically structured populations in time-periodic environments, SIAM J. Math. Anal, 50(5):5537–5568, 2018.
- [Mir17] S. MIRRAHIMI, A Hamilton-Jacobi approach to characterize the evolutionary equilibria in heterogeneous environments, Mathematical Models and Methods in Applied Sciences, 27(13):2425–2460, 2017.
- [GM17] S. GANDON AND S. MIRRAHIMI, A Hamilton-Jacobi method to describe the evolutionary equilibria in heterogeneous environments and with non-vanishing effects of mutations, Comptes Rendus Mathématique, 355(2):155–160, 2016.
- [MR16] S. MIRRAHIMI AND J.-M. ROQUEJOFFRE, A class of Hamilton-Jacobi equations with constraint: uniqueness and constructive approach, Journal of differential equations, 250(5):4717–4738, 2016.
- [MR15b] S. MIRRAHIMI AND J.-M. ROQUEJOFFRE, Uniqueness in a class of Hamilton-Jacobi equations with constraints, Comptes Rendus Mathematique, 353:489–494, 2015.
- * [MP15] S. MIRRAHIMI AND B. PERTHAME, Asymptotic analysis of a selection model with space, J. Math. Pures Appl., 104:1108–1118, 2015.

- [MM15] S. MÉLÉARD AND S. MIRRAHIMI, Singular limits for reaction-diffusion equations with fractional Laplacian and local or nonlocal nonlinearity, Communications in Partial Differential Equations, 40(5):957–993, 2015.
- * [BM15] E. BOUIN AND S. MIRRAHIMI, A Hamilton-Jacobi limit for a model of population structured by space and trait, Comm. Math. Sci., Vol. 13(6):1431–1452, 2015.
- * [LMM15] H. LEMAN, S. MÉLÉARD AND S. MIRRAHIMI, Influence of a spatial structure on the long time behavior of a competitive Lotka-Volterra type system, Discrete Contin. Dyn. Syst. Ser. B, 20(2):469–493, 2015.
- [MPS15] S. MIRRAHIMI, B. PERTHAME, P. E. SOUGANIDIS, *Time fluctuations in a population model of adaptive dynamics*, Annales de l'Institut Henri Poincaré (C) Non Linear Analysis, 32(1):41–58, 2015.
- * [MPW14] S. MIRRAHIMI, B. PERTHAME AND J. Y. WAKANO, *Direct competition re*sults from strong competition for limited resource, Journal of mathematical biology, Vol. 68(4):931–949, 2014.
- * [MS13] S. MIRRAHIMI AND P. E. SOUGANIDIS, A homogenization approach for the motion of motor proteins, Nonlin. Diff. Eq. Appl. (NoDEA), 20(1):129–147, 2013.
- [Mir13] S. MIRRAHIMI, Adaptation and migration of a population between patches, Discrete Contin. Dyn. Syst. Ser. B, 18(3):753-768, 2013.
- ◊ [MR13] S. MIRRAHIMI AND G. RAOUL, Population structured by a space variable and a phenotypical trait, Theoretical Population Biology, 84:87–103, 2013.
- * [BCM⁺12] E. BOUIN, V. CALVEZ, N. MEUNIER, S. MIRRAHIMI, B. PERTHAME, G. RAOUL, AND R. VOITURIEZ, *Invasion fronts with variable motility: phenotype selection, spatial sorting and wave acceleration*, Comptes rendus Mathématique 350:761-766, 2012.
- ◊ [MBPS12] S. MIRRAHIMI, G. BARLES, B. PERTHAME AND P. E. SOUGANIDIS, Singular Hamilton-Jacobi equation for the tail problem, SIAM J. Math. Anal. 44(6):4297–4319, 2012.
- ◊ [MPW12] S. MIRRAHIMI, B. PERTHAME AND J. Y. WAKANO, Evolution of species trait through resource competition, J. Math. biol., 64(7):1189 –1223, 2012.
- [LMP11] A. LORZ, S. MIRRAHIMI, AND B. PERTHAME, Dirac mass dynamics in multidimensional nonlocal parabolic equations, Communications in Partial Differential Equations (CPDE), 36(6):1071-1098, 2011.
- ◊ [BMP09] G. BARLES, S. MIRRAHIMI AND B. PERTHAME, Concentration in Lotka-Volterra parabolic or integral equations: a general convergence result. Methods and Applications of Analysis (MAA) 16(3):321–340, 2009.

Chapter 1

A Hamilton-Jacobi approach for models from evolutionary biology

1.1 Introduction

In this chapter we present the main elements of an approach based on Hamilton-Jacobi equations with constraint, and some recent results on the well-posedness of this approach, to study the evolutionary dynamics of populations structured by quantitative traits.

The ability of an individual to survive and to reproduce depends indeed on genetic or phenotypic parameters called traits. Several mechanisms contribute to the evolution of the living organisms: heredity, i.e. vertical transmission of the ancestral trait to the offspring, mutation which generates variability in the trait values, selection which results from the interaction of individuals with their environment, and horizontal gene transfer, i.e. horizontal exchange of genetic information between individuals during their life time. Is it possible to predict the survival or extinction of a population which is subject to such mechanisms? Can we characterize the phenotypic distribution of such population? Such types of questions emerge for instance in the study of the impact of an environmental change, e. g. climate change, on a population or in the investigation of an efficient therapy avoiding resistance of bacteria or cancer cells to medications [LMR10, CLM10, KM14, BPnMGI17].

Several mathematical formalisms are used to study the Darwinian evolution of phenotypically structured populations. Game theory is one of the first approaches to study evolution and remains one of the important tools in this field [MS74, HS88]. Adaptive dynamics, which is based on the study of the stability of differential systems, allows to study evolution under rare mutations (see for instance [MHM⁺96, GKMM98, Die04]). Integro-differential models (usually elliptic, parabolic or kinetic equations) have also contributed a lot to the understanding of evolutionary dynamics of large populations (see for instance [MW00, CC04, DJMP05, DJMR08, JR11]). Finally, probabilistic approaches allow to study finite populations (see for instance [CL07]) and also to justify the above frameworks from individual based models by proper choices of scaling on the size of the population, birth, death and mutations (see for instance [CFM06, CFM08]).

Here, we are interested in the qualitative behaviour of parabolic Lotka-Volterra type integrodifferential models describing populations with asexual reproduction. Such integro-differential equations have the property that in the limit of small diffusion, representing the mutations, and in

2

long time the solution concentrates on one or several evolving points corresponding to dominant traits. A logarithmic transformation of the solutions to such equations lead to Hamilton-Jacobi equations with constraint. In this chapter, we present the main elements of an asymptotic approach based on the study of such Hamilton-Jacobi equations with constraint. This approach was first suggested in [DJMP05] and the first general results were provided in [BP07, PB08]. An important part of my work has also been devoted to the development of this approach for general selection-mutation models with possible heterogeneity. Note that related tools were already used to study the propagation phenomena for local reaction-diffusion equations (see for instance [Fre85a, ES89]).

A main contribution presented in this chapter corresponds to a uniqueness result for a Hamilton-Jacobi equation with constraint derived from a typical selection-mutation model. While the derivation of such Hamilton-Jacobi equations was obtained earlier in [PB08, BMP09], the uniqueness of the solution, being a more challenging question, was proved in a later work with J.-M. Roquejoffre in [MR16], under concavity assumptions on the growth rate. This result, in addition to the well-posedness of the Hamilton-Jacobi equation with constraint, provides a constructive existence result which was not available previously. In addition to the uniqueness property, within this framework, we show that the solution to the Hamilton-Jacobi equation with constraint is indeed smooth and classical and it has a unique maximum point. This allows us to describe the asymptotic phenotypic density as a single Dirac mass which evolves with time [LMP11]. Finally, this analysis leads to error estimates and an asymptotic expansion of the solution providing more quantitative results for biological applications [MR15b].

In Section 1.2 we introduce a typical selection-mutation model and our assumptions. In Section 1.3 we show how the asymptotic study of such selection-mutation equations leads to Hamilton-Jacobi equations with constraint. In Section 1.4 we introduce a particular framework, with concavity assumptions, where one can obtain concentration of n as a Dirac mass, using the limiting Hamilton-Jacobi equation with constraint [LMP11]. In Section 1.5 we present our uniqueness result for the Hamilton-Jacbi equation with constraint. In Section 1.6 we provide the main elements of the proof of the uniqueness result. These two sections are based on the works in [MR15b, MR16].

1.2 Model and assumptions

In this section we present a typical selection-mutation equation modeling populations structured by quantitative traits in a homogeneous environment. We study the following equation:

$$\begin{cases} \partial_t n - \sigma \Delta n = n R(z, I), & (t, z) \in \mathbb{R}^+ \times \mathbb{R}^d, \\ I(t) = \int_{\mathbb{R}^d} n(t, y) \psi(y) dy, & (1.1) \\ n(0, z) = n_0(z). \end{cases}$$

Here, t corresponds to time and z corresponds to a phenotypic trait. The function n represents the phenotypic density of a population. The term ψ models the consumption rate of the resources and I(t) corresponds to the total consumption of the population. The growth rate of the individuals is denoted by R(z, I) which is decreasing with respect to the total consumption I, taking into account in this way competition between the individuals. The Laplace term models the mutations and the term σ measures the effect of mutations. This type of model was first suggested by Kimura [Kim65] and was later derived from stochastic individual based models [CFM06, CFM08].

A simple example with above type of growth rate is the chemostat model [Die04, DJMP05, MPBM11]. Chemostat is a laboratory device to culture micro-organisms like bacteria or algae. A standard equation for the chemostat is written

$$\begin{cases} \frac{d}{dt}S(t) = d(S_{\rm in} - S(t)) - S(t) \int_{\mathbb{R}} \psi(y)n(t, y)dy, \\ \frac{\partial}{\partial t}n(t, z) - \sigma\Delta n(t, z) = -dn(t, z) + S(t)\psi(z)n(t, z). \end{cases}$$
(1.2)

Here we consider a population of micro-organisms, characterized by a parameter $z \in \mathbb{R}$ (it can be thought of as the logarithm of their size), which live in a bath containing a nutrient which is continuously renewed with a rate d > 0. The nutrient concentration is denoted by S(t) and the fresh nutrient by $S_{in} > 0$. The population density is denoted by n(t, z) and uptake rate for individuals of trait z is $\psi(z) > 0$. Finally, we model the mutations by a diffusion term.

We next suppose that the nutrients have fast dynamics compared to the time scale of the dynamics of the population and they reach quickly an equilibrium. Then, S(t) is given by

$$S(t) = \frac{dS_{\text{in}}}{d + \int_{\mathbb{R}} \psi(y) n(t, y) dy}$$

Replacing this in (1.2) we obtain an equation of type (1.1) with

$$R(z, I) = \frac{d\psi(z)S_{\rm in}}{d + I(t)} - d.$$

Different variants of model (1.1) are used to model selection and mutation mechanisms. A wellknown equation is the direct competition model where the growth rate R(z, I) is replaced by a term $a(z) - \int c(z, y)n(t, y)dy$ (see for instance [BP97, DL00, CFM08]). Such growth rate considers more important competition between individuals with closer traits. In [DJMR08, JR11, CJR10] the long time behavior of the solutions to equations with such competition term, but without the diffusion term, are investigated. See also [LCS15, CJL15] for numerical analysis of such equations. [Rao11, CCDR13] provide also asymptotic analysis of the direct competition model, considering small rate of mutations, where the mutations are modeled via an integral kernel instead of a Laplace term. The Hamilton-Jacobi approach may also be applied to such models [BMP09, LMP11]. In the particular case where $a(x) \equiv 1$, the model is known as the nonlocal Fisher-KPP equation. For such model, the existence of traveling waves, pulsating waves and wave trains are widely investigated (see [GVA06, BNPR09, NTP10, NRRP13, HR14, FH15] and references therein). Models considering several resources are also studied [CJ11, CJM14]. In the case where $R(z,I) = \psi(z) - I$, such equation is known as the replicator-mutator equation [Kim65]. Note that in this case, the total population size is conserved (equal to 1). Then, considering particular forms of ψ , one can solve explicitly either the equation [AC14] or an equation verified by the cumulant generating function of the phenotypic distribution [GHGR17].

We are interested in the study of (1.1) in the case where the mutations have small effects. To this end, we take $\sigma = \varepsilon^2$, with ε a small parameter. We also make a change of variable in time:

$$t\mapsto \frac{t}{\varepsilon},$$

which accelerates the dynamics in order to observe the effect of small mutations. We hence define

$$n_{\varepsilon}(t,z) = n(\frac{t}{\varepsilon},z),$$

which solves the following equation

$$\begin{cases} \varepsilon \partial_t n_\varepsilon - \varepsilon^2 \Delta n_\varepsilon = n_\varepsilon R(z, I_\varepsilon), & (t, z) \in \mathbb{R}^+ \times \mathbb{R}^d, \\ I_\varepsilon(t) = \int_{\mathbb{R}^d} n_\varepsilon(t, y) \psi(y) dy, & \\ n_\varepsilon(0, z) = n_{\varepsilon,0}(z). \end{cases}$$
(1.3)

The selection-mutation models have the property that, in the limit of small diffusion and in long time, the solution concentrates on one or several evolving points corresponding to dominant traits. In the case of (1.3) we expect that the solution concentrates as a single Dirac mass since there is a single limiting factor (the nutrient modeled via the term I_{ε}) in the model. The principle of competitive exclusion [Lev70, Sch74] in evolutionary biology states indeed that when there are k limiting factors for the population, no more than k distinct traits may coexist. See Figure 1.1 for an illustration of the solution of (1.3) for a choice of parameters, which is compatible with our assumptions below.



Figure 1.1: The dynamics of the phenotypic density. The colors represent different values of the phenotypic density $n_{\varepsilon}(t, z)$, the solution to (1.3). The population concentrates on a dominant trait which evolves with time. Here we consider the following parameters $R(z, I) = 7 - z^2 - I$, $\psi(z) \equiv 1$, $n_{\varepsilon,0}(z) = \frac{1}{\sqrt{\varepsilon\pi}} \exp(-\frac{(z+2)^2}{\varepsilon})$ and $\varepsilon = 0.02$.

Note that the main difficulties in the study of such problems come from the non-local term I_{ε} . Due to this term (1.3) does not admit a comparison principle. Moreover, as $\varepsilon \to 0$, the limit of I_{ε} is only of bounded variation and can be potentially discontinuous.

Assumptions:

To state more precise results on equation (1.3) we first present some assumptions. We assume that the growth rate R is a strictly decreasing function of the competition term I:

$$-K_1 \le \frac{\partial}{\partial I} R(z, I) \le -K_2.$$
(1.4)

We also assume that there is a certain threshold for the competition term I above which the growth rate is negative everywhere:

$$\max_{z \in \mathbb{R}^d} R(z, I_M) = 0.$$
(1.5)

This assumption means that the resources are limited. We also make the following technical assumptions on R:

$$R(\cdot, I) \in W^{2,\infty}_{\text{loc}}(\mathbb{R}^d), \quad -K_4(1+|z|^2) \le R(z, I), \quad \text{for all } z \in \mathbb{R}^d \text{ and } 0 \le I \le 2I_M, \quad (1.6)$$

$$-K_3 \leq \Delta(\psi(z)R(z,I)), \quad \text{for all } z \in \mathbb{R}^d \text{ and } 0 \leq I \leq 2I_M.$$
 (1.7)

We assume that the consumption rate ψ is a smooth function such that, for some positive constants ψ_m and ψ_M , we have

$$\psi_m \le \psi(z) \le \psi_M, \qquad \psi \in W^{2,\infty}(\mathbb{R}^d).$$
 (1.8)

We then assume that the initial condition is well-prepared. We suppose indeed that there exist positive constants A_i such that

$$n_{\varepsilon,0} = \frac{1}{(2\pi\varepsilon)^{\frac{d}{2}}} \exp\left(\frac{u_{\varepsilon,0}}{\varepsilon}\right), \quad \text{with} \quad -A_1 - A_2|z|^2 \le u_{\varepsilon,0}(z) \le A_3 - A_4|z| \quad \text{for all } z \in \mathbb{R}^d.$$
(1.9)

The left hand side inequality above is a technical assumption and may be relaxed. However, we keep it to simplify the presentation of the result. The right hand side inequality means that there is small size of population at infinity.

We also assume that the initial size of the population is non-negligeable:

$$I_{\varepsilon}(0) \to I_0 > 0, \qquad \text{as } \varepsilon \to 0.$$
 (1.10)

Finally, we assume that the initial population is not maladapted to the environment so that we can guarantee the persistence of the population:

$$\frac{1}{\varepsilon} \int_{\mathbb{R}^d} \psi(z) R(z, I_{\varepsilon}(0)) n_{\varepsilon,0}(z) dz \ge o(1), \quad \text{as } \varepsilon \to 0.$$
(1.11)

1.3 Convergence to a Hamilton-Jacobi equation with constraint

In this section, we show that a Hamilton-Jacobi equation with constraint can be derived from (1.3). In order to study the limit of n_{ε} , as $\varepsilon \to 0$, we first introduce the following Hopf-Cole transformation

$$n_{\varepsilon} = \frac{1}{(2\pi\varepsilon)^{d/2}} \exp\left(\frac{u_{\varepsilon}}{\varepsilon}\right). \tag{1.12}$$

The reason why we introduce such transformation is that the solution n_{ε} to (1.3) will naturally have this form. While we expect that the limit of (n_{ε}) , as $\varepsilon \to 0$, would be a measure, the limit of (u_{ε}) will be a continuous function that solves a Hamilton-Jacobi equation. The main idea of the approach is to first study u_{ε} and next obtain information on n_{ε} from the properties of u_{ε} .

Replacing (1.12) in (1.3) we obtain the following equation on u_{ε} :

$$\partial_t u_{\varepsilon} - \varepsilon \Delta u_{\varepsilon} = |\nabla u_{\varepsilon}|^2 + R(z, I_{\varepsilon})$$

We can then pass to the limit as $\varepsilon \to 0$ and obtain the following result:

Theorem 1.1 (Convergence to a Hamilton-Jacobi equation with constraint [PB08, BMP09]). Assume (1.4)–(1.11). Let n_{ε} be the solution of (1.3), and u_{ε} be given by (1.12). Then, as $\varepsilon \to 0$ and along subsequences, (I_{ε}) converges a.e. to $I \in BV(\mathbb{R}^+)$ and (u_{ε}) converges locally uniformly to a function $u \in C((0, \infty) \times \mathbb{R}^d)$, which solves the following equation in the viscosity sense

$$\begin{cases} \partial_t u = |\nabla u|^2 + R(z, I), \quad (t, z) \in \mathbb{R}^+ \times \mathbb{R}^d, \\ \max_{z \in \mathbb{R}^d} u(t, z) = 0, \qquad t \in \mathbb{R}^+. \end{cases}$$

If additionally $(u_{\varepsilon,0})$ is a sequence of uniformly continuous functions which converges locally uniformly to u_0 then $u \in C([0,\infty) \times \mathbb{R}^d)$ and $u(0,z) = u_0(z)$ in \mathbb{R}^d .

Here, I may be thought of as a sort of regulator, or a sort of Lagrange multiplier, to maintain the maximum of u equal to 0. Note that I is only of bounded variation and can be potentially discontinuous. The above theorem was proved in [PB08] with the uniform continuity assumption on the initial condition $(u_{\varepsilon,0})_{\varepsilon}$ and without such assumption in [BMP09].

Once we have identified the limit of (u_{ε}) , we can then obtain some information on the limit phenotypic density n and in some cases to characterize n completely.

Theorem 1.2 (Convergence of (n_{ε}) [PB08]). Assume (1.4)–(1.11). Then, as $\varepsilon \to 0$ and along subsequences, n_{ε} converges in L^{∞} (w* $(0, \infty)$; $\mathcal{M}^1(\mathbb{R}^d)$) to a measure n such that, for a.e. t,

$$\int \psi(z)n(t,z)dz = I(t) \quad a.e.,$$

and

$$\operatorname{supp} n(t, z) \subset \{ z \, | \, u(t, z) = 0 \} \subset \{ z \, | \, R(z, I(t)) = 0 \}.$$
(1.13)

Note in particular that while $z \in \mathbb{R}$ and R is a monotonic function with respect to z, then (1.13) implies that n is a single Dirac mass., in the following form

$$n(t,z) = \rho(t)\delta(z - \overline{z}(t)),$$

with (\overline{z}, ρ) uniquely determined by

$$R(\overline{z}(t), I(t)) = 0, \qquad \rho(t) = \frac{I(t)}{\psi(\overline{z}(t))}$$

1.4 A concave framework and concentration as a Dirac mass

In this section, we introduce a particular framework, with concavity assumptions, where one can obtain concentration of n as a Dirac mass in the multi-d case. To this end, additionally to the above assumptions we also make the following concavity assumptions, for all $z \in \mathbb{R}^d$ and $0 \leq I \leq 2I_M$,

$$-2B_1 \le D^2 R(z,I) \le -2B_2, \quad -\underline{B}_0 - B_1 |z|^2 \le R(z,I) \le -\overline{B}_0 - B_2 |z|^2, \tag{1.14}$$

$$-2D_1 \le D^2 u_{\varepsilon,0}(z) \le -2D_2, \quad -\underline{D}_0 - D_1 |z|^2 \le u_{\varepsilon,0}(z) \le -\overline{D}_0 - D_2 |z|^2.$$
(1.15)

Here Df represents the derivative of the function f with respect to z. Under such assumptions, the limit function u has strong regularity properties and it is strictly concave with respect to z, and hence it has a unique maximum point. The property (1.13) then implies that n is indeed a single Dirac mass.

More precisely, we have

Theorem 1.3 (Convergence to a Dirac mass in the concave framework [LMP11]). Assume (1.4)-(1.11) and (1.14)-(1.15). Then, we have

$$-E_1 \leq D^2 u_{\varepsilon}(t,z) \leq -E_2, \quad for \ all \ (t,z) \in \mathbb{R} \times \mathbb{R}^d,$$

for some positive constants E_1 and E_2 . Moreover, the limit function u belongs to $L^{\infty}_{\text{loc}}\left(\mathbb{R}^+; W^{2,\infty}_{loc}(\mathbb{R}^d)\right) \cap W^{1,\infty}_{\text{loc}}\left(\mathbb{R}^+; L^{\infty}_{loc}(\mathbb{R}^d)\right)$, and it is strictly concave with respect to z. As a consequence, we can identify n as follows

$$n(t,z) = \rho(t)\delta(z-\overline{z}(t)), \qquad \rho(t) = \frac{I(t)}{\psi(\overline{z}(t))},$$

with $\overline{z}(t) \in \mathbb{R}^d$ the unique maximum point of $u(t, \cdot)$.

If we make the following additional regularity assumptions,

$$\|D^3 R\|_{L^{\infty}(\mathbb{R}^d \times [0, 2I_M])} \le B_3, \tag{1.16}$$

$$\|D^3 u_{\varepsilon,0}\|_{L^{\infty}(\mathbb{R}^d)} \le D_3, \tag{1.17}$$

we can then also provide a differential equation for the dominant trait $\overline{z}(t)$:

Theorem 1.4 (The canonical equation [LMP11]). Assume (1.4)-(1.11) and (1.14)-(1.17). Then, we have the following regularity properties

$$(u,I) \in L^{\infty}_{\text{loc}}\left(\mathbb{R}^+; W^{3,\infty}_{loc}(\mathbb{R}^d)\right) \cap C^1(\mathbb{R}^+ \times \mathbb{R}^d) \times C^1(\mathbb{R}^+).$$

Moreover, $\bar{z}(\cdot)$ belongs to $C^1(\mathbb{R}^+;\mathbb{R}^d)$ and satisfies

$$\dot{\bar{z}}(t) = \left(-D^2 u\bigl(t, \bar{z}(t)\bigr)\bigr)^{-1} \cdot \nabla_z R\bigl(\bar{z}(t), I(t)\bigr).$$
(1.18)

1.5 A uniqueness result for the Hamilton-Jacobi equation with constraint

Together with Jean-Michel Roquejoffre in [MR15b, MR16] we provided a uniqueness result for the viscosity solution to the Hamilton-Jacobi equation with constraint

$$\begin{cases} \partial_t u = |\nabla u|^2 + R(z, I), \quad (t, z) \in \mathbb{R}^+ \times \mathbb{R}^d \\ \max_{z \in \mathbb{R}^d} u(t, z) = 0, \quad t \in \mathbb{R}^+, \\ u(0, z) = u_0(z), \end{cases}$$
(1.19)

in the concave framework presented in the previous section, and showed that this result leads indeed to an asymptotic expansion for u_{ε} and hence to a more quantitative result for biological applications. Note that the uniqueness of the solution to (1.19) is a nonstandard problem because of the constraint $\max_{z \in \mathbb{R}^d} u(t, z) = 0$. The function I has to be chosen in a way such that this constraint holds. Such uniqueness property was indeed a missing element of the Hamilton-Jacobi approach for evolutionary biology since the seminal work of Perthame and Barles [PB08]. Our result allows to guarantee the well-posedness of this theory in the concave framework.

To present our main result we state the assumptions on u_0 , which are analogous to the assumptions above on $u_{\varepsilon,0}$, and an additional regularity assumption on R.

Assumptions on u_0 :

We assume the existence of positive constants $\overline{D}_0, \overline{D}_1, \overline{D}_2$ and \overline{D}_3 such that

$$-\overline{D}_0 - \overline{D}_1 |z|^2 \le u_0(z) \le \overline{D}_0 - \overline{D}_2 |z|^2, \qquad (1.20)$$

$$-2\overline{D}_1 \le D^2 u_0 \le -2\overline{D}_2,\tag{1.21}$$

$$\|D^3 u_0\|_{L^{\infty}(\mathbb{R}^d)} \le \overline{D}_3, \tag{1.22}$$

$$\max_{z} u_0(z) = u_0(\overline{z}_0) = 0, \qquad R(\overline{z}_0, I_0) = 0.$$
(1.23)

Additional regularity assumption on R:

We also assume that there exists a positive constant B_4 such that

$$\left|\frac{\partial^2 R}{\partial I \partial z_i}(z,I)\right| + \left|\frac{\partial^3 R}{\partial I \partial z_i \partial z_j}(z,I)\right| \le B_4, \quad \text{for } 0 \le I \le I_M, \text{ and } i, j = 1, 2, \cdots, d. \quad (1.24)$$

Here is our main result:

Theorem 1.5 (Uniqueness for the Hamiton-Jacobi equation with constraint [MR16]). Assume (1.4)-(1.6), (1.14), (1.16) and (1.20)-(1.24). Then, the Hamilton-Jacobi equation with constraint (1.19) has a unique viscosity solution $(u, I) \in C(\mathbb{R}^+ \times \mathbb{R}^d) \times C((0, +\infty))$, which is indeed a classical solution. Moreover we have

$$(u,I) \in L^{\infty}_{loc}(\mathbb{R}^+; W^{3,\infty}_{loc}(\mathbb{R}^d)) \cap C^1(\mathbb{R}^+ \times \mathbb{R}^d) \times C^1(\mathbb{R}^+) \quad and \quad \nabla u \in C^1(\mathbb{R}^+ \times \mathbb{R}^d).$$

The above theorem on the one hand provides the uniqueness of the solution to the Hamilton-Jacobi equation with constraint, in the concave framework. On the other hand it provides a constructive existence result while the previous existence result obtained by the vanishing viscosity limit of the viscous problem was not constructive. Moreover, this result implies that the convergence of (u_{ε}) to u is not only along subsequences and the whole sequence converges to the same limit.

The method used to prove such result is rather robust and can be generalized to study related models where for instance the competition term does not only depend on an integral with a weight of the phenotypic density n_{ε} but on a convolution term, as for the direct competition model:

$$\varepsilon \partial_t n_\varepsilon - \varepsilon^2 \Delta n_\varepsilon = n_\varepsilon \big(r(z) - \int n(t, y) C(z, y) dy \big),$$

under some conditions presented in [LMP11].

Note also that such uniqueness result also allows to overcome some technical difficulties in the study of more complex heterogeneous models. In models with heterogeneity it is often harder to obtain regularity estimates in order to obtain the asymptotic convergence of the viscous problem. In such cases, one can indeed use a uniqueness result to pass to the limit (see for instance [MP15]).

The most important consequence of the uniqueness result is that it leads indeed to error estimates and an asymptotic expansion of u_{ε} in terms of ε ([MR15b] and work in progress with Jean-Michel Roquejoffre):

$$u_{\varepsilon} = u + \varepsilon v + o(\varepsilon), \qquad I_{\varepsilon} = I + \varepsilon J + o(\varepsilon).$$

The above expansion allows to obtain an interesting approximation of the phenotypic density n_{ε} :

$$n_{\varepsilon}(t,z) = \frac{1}{(2\pi\varepsilon)^{d/2}} \exp\left(\frac{u(t,z)}{\varepsilon} + v(t,z) + o(1)\right).$$
(1.25)

Such result goes further than the usual approximations used by biologists in the theory of quantitative genetics which is a theory in evolutionary biology that investigates the evolution of continuously varying traits (see for instance [Ric04]–chapter 7). Such approximations are usually based on an a priori assumption that the phenotypic distribution is a Gaussian.

We can go further in the approximations and use (1.25) and the Laplace's method of integration to obtain first order asymptotic expansions, in terms of ε , for the moments of the phenotypic distribution in terms of the derivatives of the functions u and v at the point $(t, \overline{z}(t))$, where $\overline{z}(t)$ is the unique maximum point of $u(t, \cdot)$. In particular, we can compute the covariance matrix of the phenotypic distribution as follows

$$\mathbf{V}_{\varepsilon,t} = \varepsilon (-D^2 u)^{-1} (t, \overline{z}(t)) + o(\varepsilon),$$

where $\mathbf{V}_{\varepsilon,t} = (v_{i,j}(t))$ with $v_{i,j} = \int z_i z_j \frac{n_\varepsilon(t,z)}{\rho_\varepsilon(t)} dx - (\int z_i \frac{n_\varepsilon(t,z)}{\rho_\varepsilon(t)} dz)(\int z_j \frac{n_\varepsilon(t,z)}{\rho_\varepsilon(t)} dz)$. Note that here we find again the term $(-D^2 u)^{-1}(t, \overline{z}(t))$ which also appeared in equation (1.18). This property allows to provide a biological interpretation of this equation and in particular to compare it to the so-called canonical equation in the theory of adaptive dynamics [Die04, DL96] or to Lande's equation in the theory of quantitative genetics [Lus37, Lan79, LA83]. In these equations, which are very related equations under different formalisms, the change in the dominant/average trait is given by the product of the gradient of the fitness and a term that scales the rate of evolutionary change (proportional to mutational variance or genetic variance respectively in adaptive dynamics and quantitative genetics). In (1.18), the dynamics of the dominant trait $\dot{\overline{z}}(t)$ is also given by the product of the gradient of the fitness $\nabla_z R$ and the term $(-D^2 u(t, \bar{z}(t)))^{-1}$ which approximates well the phenotypic covariance matrix. In this way, (1.18) may be seen as a generalization of the canonical equation or Lande's equation to a case where the mutations are not assumed to be very rare (on the contrary to adaptive dynamics) and such that the evolution of the genetic variance is included in the dynamics.

In chapters 3 and 4 we will present some models with heterogeneous environments where such analytic approximations allow to bring insight into biological questions.

More recently, the uniqueness of the viscosity solutions of (1.19) in a more general setting was proved in [CL]. However, this more general result does not allow to describe the structure of the solutions, nor to prove the concentration of n on a single point. Furthermore this work does not lead to an asymptotic expansion of u_{ε} .

1.6 The main elements of the proof of Theorem 1.5

A first main element in the proof of Theorem 1.5 is the fact that under the assumptions of the theorem, any viscosity solution u to (1.19) is indeed smooth and strictly concave. We prove indeed such result for an unconstrained Hamilton-Jacobi equation where R(z, I) is replaced by R(t, z):

$$\begin{cases} u_t = |\nabla u|^2 + R(t, z) & (t > 0, z \in \mathbb{R}^d), \\ u(0, z) = u_0(z). \end{cases}$$
(1.26)

To present our results on (1.26) we provide our assumptions on R(t, z) which are closely related to the ones on R(z, I).

Assumptions on R(t, z):

We choose R to be continuous in t and to have first and second derivatives with respect to z,

that are continuous both with respect to t and z. We suppose that

$$-\overline{B}_0 - \overline{B}_1 |z|^2 \le R(t, z) \le \overline{B}_0 - \overline{B}_1 |z|^2, \quad \text{for } t \in \mathbb{R}^+,$$
(1.27)

$$-2\overline{B}_1 \le D^2 R(t,z) \le -2\overline{B}_2 < 0 \text{ as symmetric matrices}, \tag{1.28}$$

$$\|D^3 R(t, \cdot)\|_{L^{\infty}(\mathbb{R}^d)} \le \overline{B}_3, \qquad \text{for } t \in \mathbb{R}^+.$$
(1.29)

Theorem 1.6 ([MR16]). Assume (1.20)–(1.22) and (1.27)–(1.29). Equation (1.26) has a unique viscosity solution u that is bounded from above. Moreover, it is a classical solution: $u \in L^{\infty}_{loc}(\mathbb{R}^+; W^{3,\infty}_{loc}(\mathbb{R}^d)) \cap C^1(\mathbb{R}^+ \times \mathbb{R}^d), \nabla u \in C^1(\mathbb{R}^+ \times \mathbb{R}^d),$

$$-\max(2\overline{D}_1,\sqrt{\overline{B}_1}) \le D^2 u \le -\min(2\overline{D}_2,\sqrt{\overline{B}_2}) \quad and \quad \|D^3 u\|_{L^{\infty}([0,T]\times\mathbb{R}^d)} \le C(T),$$

where C(T) is a positive constant depending on $\overline{D}_1, \overline{B}_1, \overline{D}_3, \overline{B}_3$ and T.

The existence and uniqueness of viscosity solution to (1.26) is rather classical. The main ingredient to prove the regularity properties above is to use the fact that the viscosity solution to (1.26) is given by the dynamic programming principle:

$$u(t,z) = \sup_{\substack{(\gamma(s),s) \in \mathbb{R}^d \times [0,t]\\\gamma(t)=x}} \left\{ F(\gamma) : \gamma \in C^1([0,t];\mathbb{R}^d) \right\},$$

with

$$F(\gamma) := u_0(\gamma(0)) + \int_0^t \left(-\frac{|\dot{\gamma}|^2}{4}(s) + R(s,\gamma(s)) \right) ds,$$

and the fact that u_0 and R are strictly concave functions.

The second step of the proof is to show that (1.19) is equivalent with the following slightly nonstandard differential system:

$$\begin{cases} R\left(\overline{z}(t), I(t)\right) = 0, & \text{for } t \in \mathbb{R}^+, \\ \overline{z}(t) = \left(-D^2 u\left(t, \overline{z}(t)\right)\right)^{-1} \nabla R\left(\overline{z}(t), I(t)\right), & \text{for } t \in \mathbb{R}^+, \\ \partial_t u = |\nabla u|^2 + R(z, I), & \text{in } \mathbb{R}^+ \times \mathbb{R}^d, \end{cases}$$
(1.30)

with initial conditions

$$I(0) = I_0, \qquad u(0, \cdot) = u_0(\cdot), \qquad \overline{z}(0) = \overline{z}_0,$$

such that $\max_z u_0(z) = u_0(\overline{z}_0) = 0$ and $R(\overline{z}_0, I_0) = 0.$ (1.31)

Note that (1.30) is really a differential system because the assumptions on R imply that I(t) can implicitly be expressed in terms of $\bar{z}(t)$. And it is slightly nonstandard because \bar{z} solves an ODE whose nonlinearity depends on u. The precise statement is the following

Theorem 1.7 ([MR16]). Solving the constrained problem (1.19) is equivalent to solving the initial value ODE-PDE problem (1.30)–(1.31).

Finally the last step is to prove that (1.30)-(1.31) has a unique solution. This is obtained using a Banach fixed point argument, by defining an adapted mapping and proving, thanks to some technical computations, that such mapping is a strict contraction.

Chapter 2

Fractional reaction-diffusion equations

2.1 Introduction

In this chapter, we are interested in the study of the following selection-mutation model where the mutations are modeled by a fractional laplacian:

$$\begin{cases} \partial_t n + \sigma(-\Delta)^{\alpha} n = n R(z, \rho(t)), & (t, z) \in \mathbb{R}^+ \times \mathbb{R}^d \\ n(t=0, \cdot) = n^0(\cdot), & \rho(t) = \int_{\mathbb{R}^d} n(t, z) dz, \end{cases}$$
(2.1)

with $0 < \alpha < 1$ and

$$(-\Delta)^{\alpha} n(t,z) = \text{p.v.} \int_{R^d} \left(n(t,z) - n(t,z+h) \right) \frac{dh}{|h|^{d+2\alpha}}.$$

Here, as in Chapter 1, z corresponds to a phenotypic trait and n represents the phenotypic density. The total size of the population is denoted by ρ . The function R represents the growth rate of individuals with trait z, and it is a decreasing function with respect to the total size of the population ρ , taking into account in this way competition between the individuals. The difference with (1.1) is that the mutations are modeled by a fractional laplacian instead of a classical laplacian. The choice of a fractional laplacian rather than a classical laplacian or an integral kernel with thin tails, allows to take into account large mutation jumps with a high rate. Such equation has been derived from stochastic individual based models, in the case where the mutation distribution has algebraic tails [JMW12].

We are interested in an asymptotic analysis of the above equation to understand the effect of large mutation jumps on the dynamics of the phenotypic density. Is there still a rescaling under which one would observe the concentration of the phenotypic density as a Dirac mass? Can the Hamilton-Jacobi approach be extended to study mutation integral kernels with thick tails? Models closely related to (2.1) but considering regular integral kernels with thin tails or with compact support, instead of the fractional laplacian, have been previously studied (see for instance [BMP09, Rao11, GHGR17]). However, not much is known for the models with heavy tailed integral kernels. The presence of large mutations with a high probabilty leads to drastically faster dynamics and the previous scalings are not adapted to the study of such problems. With S. Méléard, in order to obtain a relevant rescaling and to perform an asymptotic analysis for the fractional selection-mutation equation, we first studied a fractional Fisher-KPP equation:

$$\partial_t n + (-\Delta)^{\alpha} n = n(1-n).$$

The level sets of the solution to such equation propagate with a speed that is exponential in time [CCR12, CR13]. Note that such propagation is much faster compared to the classical Fisher-KPP equation where the propagation has constant speed [Fis37, KPP37, AW78]. In [MM15] we provided an unconventional rescaling that allowed us to provide an asymptotic analysis of this equation and capture the exponential speed of propagation associated to this equation. This work extends previous works on the asymptotic analysis of propagation phenomena, related to large deviation principles [FW98], in classical reaction-diffusion equations to the case of diffusion kernels with fat tails (see for instance [Fre85a, Fre86, ES89]).

Next, being inspired by such analysis we introduced a rescaling for the fractional selectionmutation model which leads to the concentration of the phenotypic density as a single Dirac mass, similarly to the case of classical selection-mutation models [MM15, Mir]. However, significant difficulties comparing to previous works arise from the fractional laplacian term which lead to considerably different analysis. In particular, the logarithmic transform of the phenotypic density does not converge anymore to a viscosity solution of a Hamilton-Jacobi equation, but to a viscosity supersolution which is minimal in a certain class of functions. Moreover, the limiting Hamilton-Jacobi equation can take infinite values.

The result in [MM15] on the fractional KPP equation has been extended in several directions. With J.-M. Roquejoffre, we supervise a PhD student, A. Léculier, on reaction-diffusion equations with nonlocal diffusion operators. In a first work [L19], he has extended the result in [MM15] on the fractional Fisher-KPP equation to more general stable operators and considering periodic media (see also [CCR12, ST19]). Together with A. Léculier and J.-M. Roquejoffre we also have expanded this approach to study fractional KPP type equations in a periodically hostile environment. Such model describes the nonlocal dispersion of a population in a domain with periodic presence of obstacles.

In [BGHP18], Bouin et al. extended also this method to other dispersion kernels without singularity but with other types of decay. It is expected that the more recent work [Mir] on the fractional selection-mutation model can also be extended to such integral kernels following the same type of adaptation.

In Section 2.2 we present the asymptotic analysis of the fractional Fisher-KPP equation, which is based on the work in [MM15]. In Section 2.3 we go back to the case of the fractional selectionmutation model (2.1) and provide our results on the derivation of a Hamilton-Jacobi equation with constraint and the concentration of n_{ε} as a Dirac mass. This part is based on the work in [Mir]. Finally, in Section 2.4 we present a result on the fractional Fisher-KPP equation in a periodically hostile environment [LMR].

2.2 Fractional Fisher-KPP equation

In this section we study the asymptotic behavior of the fractional Fisher-KPP equation:

$$\begin{cases} \partial_t n(t,x) + (-\Delta)^{\alpha} n = n (1-n), \\ n(0,x) = n^0(x), \ x \in \mathbb{R}^d, \end{cases}$$

$$(2.2)$$

with $\alpha \in (0, 1)$. Note that, here x may correspond to a spatial position and n to a spatial density. Then (2.2) describes the spatial dispersion of a population under logistic growth. The fractional diffusion describes the situations where the individuals can jump (move rapidly) from one point to another with a high probability, for instance because of the wind for seeds or human transportation for animals.

It has been proved in [CCR12, CR13] that the level sets of n, the solution to (2.2), propagate with a speed that is exponential in time (see also [Eng10, BRR11, Gar11, CR12, RT17] for related works on the speed of propagation for reaction-diffusion equations with fractional laplacian or a diffusion term with heavy tails). In particular, in [CR13] it is proved that for any initial data such that

$$0 \le n_0(x) \le C \frac{1}{1+|x|^{d+2\alpha}},\tag{2.3}$$

we have

$$\begin{cases} n(t,x) \to 0, & \text{uniformly in } \mathcal{A}_{\sigma} = \{ |x| \ge e^{\sigma t} \}, \text{ if } \sigma > \frac{1}{d+2\alpha}, \text{ as } t \to \infty, \\ n(t,x) \to 1, & \text{uniformly in } \mathcal{B}_{\sigma} = \{ |x| \le e^{\sigma t} \}, \text{ if } \sigma < \frac{1}{d+2\alpha}, \text{ as } t \to \infty. \end{cases}$$
(2.4)

Our objective is to understand this behavior using singular limits as for the KPP equation with a Laplace term (cf. [Fre85a, BES90]). The idea is to rescale the equation and to perform an asymptotic limit so that we forget the full and detailed behavior and capture only this propagation. In the case of the classical Fisher-KPP equation, where the speed of propagation is known to be constant [AW78], to study the asymptotic behavior of the solutions one should use the following rescaling [Fre85a, Fre85b, ES89, BES90]

$$x \mapsto \frac{x}{\varepsilon}, \qquad t \mapsto \frac{t}{\varepsilon}, \quad \text{and} \quad n_{\varepsilon}(t,x) = n(\frac{t}{\varepsilon},\frac{x}{\varepsilon}).$$

In the case of the fractional Fisher-KPP equation, being inspired by (2.4) we use the following long-range/long-time rescaling

$$x \mapsto r^{\frac{1}{\varepsilon}}\nu, \qquad t \mapsto \frac{t}{\varepsilon}, \qquad \text{with } r = |x| \text{ and } \nu = \frac{x}{|x|}.$$
 (2.5)

Note that with such rescaling the sets \mathcal{A}_{σ} and \mathcal{B}_{σ} remain unchanged. We then define

$$n_{\varepsilon}(t,x) = n(\frac{t}{\varepsilon}, r^{\frac{1}{\varepsilon}}\nu), \quad \text{with } r = |x| \text{ and } \nu = \frac{x}{|x|}$$

which solves the following rescaled equation

$$\begin{cases} \varepsilon \partial_t n_\varepsilon = \mathcal{L}_\varepsilon n_\varepsilon + n_\varepsilon (1 - n_\varepsilon), \\ \mathcal{L}_\varepsilon n_\varepsilon (t, x) = \text{p.v.} \int_{\mathbb{R}^d} \left(n_\varepsilon \left(t, \left| r^{\frac{1}{\varepsilon}} \nu + h \right|^\varepsilon \frac{r^{\frac{1}{\varepsilon}} \nu + h}{|r^{\frac{1}{\varepsilon}} \nu + h|} \right) - n_\varepsilon (t, x) \right) \frac{dh}{|h|^{d+2\alpha}}, \tag{2.6}$$
$$n_\varepsilon (0, x) = n_\varepsilon^0 (x).$$

With this rescaling we are able to capture the exponential speed of propagation and obtain a result which is equivalent with the one given in [CR13]. To this end, we make the following assumption

$$0 \le n_{\varepsilon}(0, x) \le \frac{C}{1 + |x|^{\frac{d+2\alpha}{\varepsilon}}},\tag{2.7}$$

which is indeed equivalent with (2.3) after rescaling (2.5). Here is our main result: **Theorem 2.1** (Exponential speed of propagation [MM15]). Let n_{ε} be the solution of (2.6). Under assumption (2.7), as $\varepsilon \to 0$, we have

$$\begin{cases} n_{\varepsilon} \to 0, & \text{locally uniformly in } \mathcal{A} = \{(t, x) \in (0, \infty) \times \mathbb{R}^d \mid t < (d + 2\alpha) \log |x|\}, \\ n_{\varepsilon} \to 1, & \text{locally uniformly in } \mathcal{B} = \{(t, x) \in (0, \infty) \times \mathbb{R}^d \mid t > (d + 2\alpha) \log |x|\}. \end{cases}$$

Note that this result is equivalent with (2.4) proven in [CR13].

One may wonder why the rescaling (2.5) does not depend on α . Finding a relevant rescaling that would capture such exponential dynamics is indeed a major contribution of our work in [MM15]. Understanding the transition from exponential speed of propagation to constant speed of propagation as $\alpha \to 1$ is indeed very helpful to understand why our rescaling does not depend on α . In [CR12], A.-C. Coulon and J.-M. Roquejoffre studied such transition as $\alpha \to 1$. They proved that when $\alpha < 1$, the position of the front scales ultimately as $e^{\frac{t}{d+2\alpha}}$ for $t > t_{\alpha}$, with t_{α} a positive constant depending on α . At intermediate times, the position of the front scales as $2t^{1/\alpha}$. As $\alpha \to 1$, t_{α} tends to infinity and the front will ultimately move as 2t.

With our asymptotic analysis, we fix α and we let $\varepsilon \to 0$. That is we study the large time behavior and in particular at the limit we only capture the dynamics corresponding to $t > t_{\alpha}$. This means that we capture only the exponential speed of propagation and our method does not allow to capture the transition from exponential speed to constant speed of propagation as $\alpha \to 1$.

In Subsection 2.2.1 we provide the main ingredients of the proof of our result. We also show that at the limit $\varepsilon = 0$ the only impact of the fractional laplacian is at the initial time when it determines the thickness of the tails of the solution. At positive times, the fractional laplacian does not have any impact on the dynamics of the solution which are determined uniquely by the reaction term. This is an important difference with the classical Fisher-KPP equation, where the Laplace term not only forces the solution to have an exponential tail but also it still influences in positive times the dynamics and modifies the speed of propagation.

2.2.1 The main ingredients of the proof and the impact of the fractional laplacian on the dynamics

A main ingredient in the proof of Theorem 2.1 is the fact that with the rescaling (2.5) the term $(\mathcal{L}_{\varepsilon}n_{\varepsilon})n_{\varepsilon}^{-1}$ vanishes as $\varepsilon \to 0$. More precisely we find sub and supersolutions to (2.6) which are indeed sub and supersolutions to a perturbation of an ordinary differential equation derived from (2.6) by omitting the term coming from the fractional laplacian. They also have the property that when one applies the operator $(\mathcal{L}_{\varepsilon}f)f^{-1}$ to such functions, the outcome is very small.

To prove such results, we use the Hopf-Cole transformation. To show why such transformation is still relevant, we recall the following bounds on the transition probability function p associated with the fractional laplacian with coefficient α (see, e.g., Sato [Sat99] p.89 and p.202):

$$\frac{B_m}{t^{\frac{d}{2\alpha}}(1+|t^{-\frac{1}{2\alpha}}x|^{d+2\alpha})} \le p(t,x) \le \frac{B_M}{t^{\frac{d}{2\alpha}}(1+|t^{-\frac{1}{2\alpha}}x|^{d+2\alpha})}.$$
(2.8)

Note that the solution v to the following equation

$$\begin{cases} \partial_t v + (-\Delta)^{\alpha} v = 0, & \text{in } \mathbb{R} \times \mathbb{R}^+, \\ v(x,0) = v^0, & \text{in } \mathbb{R}, \end{cases}$$

satisfies

$$v(t,x) = \int_{\mathbb{R}} p(y,t)v^0(x-y)dy.$$

The inequality (2.8) is written after the rescaling (2.5) as

$$\frac{B_m}{(\frac{t}{\varepsilon})^{\frac{d}{2\alpha}}(1+|(\frac{t}{\varepsilon})^{\frac{-1}{2\alpha}}|x|^{\frac{1}{\varepsilon}}|^{d+2\alpha})} \le p_{\varepsilon}(|x|,t) = p(|x|^{\frac{1}{\varepsilon}},\frac{t}{\varepsilon}) \le \frac{B_M}{(\frac{t}{\varepsilon})^{\frac{d}{2\alpha}}(1+|(\frac{t}{\varepsilon})^{\frac{-1}{2\alpha}}|x|^{\frac{1}{\varepsilon}}|^{d+2\alpha})},$$
(2.9)

Being inspired now by (2.9) we use the classical Hopf-Cole transformation

$$n_{\varepsilon} = \exp\left(\frac{u_{\varepsilon}}{\varepsilon}\right),\tag{2.10}$$

and make the following assumption

$$\frac{C_m}{1+|x|^{\frac{d+2\alpha}{\varepsilon}}} \le n_{\varepsilon}(x,0) \le \frac{C_M}{1+|x|^{\frac{d+2\alpha}{\varepsilon}}}, \quad \text{with } C_m < 1 < C_M.$$
(2.11)

Note that this property can be obtained thanks to assumption (2.7), the estimates (2.8) on the heat kernel, and a translation in time. We then prove the following.

Theorem 2.2. Let n_{ε} be the solution of (2.6) and u_{ε} be given by (2.10). Under assumption (2.11), as $\varepsilon \to 0$, (u_{ε}) converges locally uniformly to u, which is the unique solution to

$$\begin{cases} \max(\partial_t u - 1, u) = 0, \\ u(0, x) = \min(0, -(1 + \alpha) \log |x|). \end{cases}$$
(2.12)

Therefore, u is given by

$$u(t, x) = \min(0, -(1 + \alpha)\log|x| + t).$$

This result is a main ingredient in the proof of Theorem 2.1. Note that it is immediate from (2.10) that in the set $\{u < 0\}$, $n_{\varepsilon} \to 0$, as $\varepsilon \to 0$. We also prove, using an appropriate viscosity test function, that in the set $\{u = 0\}$, $n_{\varepsilon} \to 1$, as $\varepsilon \to 0$.

In view of (2.12), we notice that at the limit $\varepsilon = 0$, the fractional laplacian does not have any impact on the dynamics of u and the dynamics are determined only by the reaction term. The only role of the fractional laplacian in the limit is at the first initial time where the tail of the solution is forced to satisfy (2.11). The exponential propagation is hence derived only from the form of the solution at the initial layer. This is an important difference with the KPP equation with the classical laplacian where, the Laplace term not only forces the solution to have an exponential tail but also it still influences in positive times the dynamics and modifies the speed of propagation. To observe this property consider the following equation

$$\begin{cases} \partial_t m - \delta \Delta m = m(1-m), & \delta \in \{0,1\}, \\ m(x,0) = \exp\left(-\frac{x^2}{2}\right). \end{cases}$$

It is easy to verify that in long time the invasion front scales as $x \sim \sqrt{2t}$ for $\delta = 0$, while for $\delta = 1$ the invasion front scales as $x \sim 2t$. Therefore, the diffusion term speeds up the propagation. Next we consider an analogous equation but with fractional laplacian:

$$\begin{cases} \partial_t m + \delta(-\Delta)^{\alpha} m = m(1-m), & \delta \in \{0,1\} \quad \alpha \in [0,1], \\ m(x,0) = m^0(x), & \text{with } \frac{C_m}{1+|x|^{d+2\alpha}} \le m^0 \le \frac{C_M}{1+|x|^{d+2\alpha}}. \end{cases}$$

Then following the computations above, in long time and for both cases $\delta = 0, 1$, the invasion front scales as $x \sim e^{\frac{t}{d+2\alpha}}$.

2.3 Fractional selection-mutation equation

We now go back to the case of fractional selection-mutation model (2.1). In Subsection 2.3.1 we compare the behavior of (2.1) with classical selection-mutation models and discuss the choice of the rescaling, being inspired from the rescaling (2.5) for the fractional Fisher-KPP equation. In Subsection 2.3.2 we provide our assumptions and in Subsection 2.3.3 we present our main results. Finally in Subsection 2.3.4 we provide the main ideas of the proof.

2.3.1 The choice of the rescaling

Before providing our rescaling of the fractional selection-mutation model, we first compare the behavior of such equation with classical selection-mutation equations. To have an idea of the difference of selection-mutation models with fractional or classical Laplace terms, we compare the numerical resolutions of (2.1) and (1.1) in Figure 2.1, considering the following parameters

$$R(z,\rho) = 1 + \exp(2 - \frac{z^2}{5}) - \rho, \qquad \sigma = 1, \qquad n(0,z) = \exp\left(-\frac{(z+2)^2}{0.5}\right).$$
(2.13)



Figure 2.1: Left: the numerical resolution of (1.1) with the parameters given in (2.13) and with $I = \rho$. The colors represent the different values of the phenotypic density n. Right: the numerical resolution of (2.1) with the same parameters. We observe that in both cases the initial solution has a distribution centered around the point z = -2, and that this distribution moves gradually to the right and it will ultimately be centered around the point z = 0 which is the optimal trait corresponding to the growth rate given in (2.13). However, in the case with the fractional diffusion, i.e. the figure at the right hand side, the evolution of the phenotypic distribution is much faster and almost immediately the distribution becomes centered around the point z = 0. Moreover, this distribution is much wider because of the presence of large jumps with a high rate.

Note that here we have considered $\psi \equiv 1$ such that $\rho = I$. Moreover, with the above choice of R, assumptions (1.4)–(1.7) are satisfied.

We observe that in both cases the initial solution has a distribution centered around the point z = -2, and that this distribution moves gradually to the right and it will ultimately be centered around the point z = 0 which is the optimal trait corresponding to the growth rate given in (2.13). However, in the case with the fractional diffusion, the evolution of the phenotypic distribution is much faster and almost immediately the distribution becomes centered around the point z = 0.

Moreover, this distribution is much wider because of the presence of large jumps with a high rate.

It is also interesting to compare (2.1) with selection-mutation models where the mutations are described by an integral kernel with thin tails:

$$\begin{cases} \partial_t n = \int_{\mathbb{R}^d} \left(n(t, z+h) - n(t, z) \right) b(z) K(h) dh + R(z, \rho), \\ \rho(t) = \int_{\mathbb{R}^d} n(t, z) dz. \end{cases}$$
(2.14)

Here K corresponds to a mutation law, i.e. K(h) is the probability of a mutation jump with size h. The other terms are as above.

Such selection-mutation model has been derived from stochastic individual based models, in the limit of large populations in [CFM06, CFM08]. In the case where K has thin tails, i.e. the decay of K is faster than exponential, an asymptotic study of (2.14), analogous to the one presented in Chapter 1, has been provided in [BMP09]. Such analysis, which assumes again small effect of mutations and long time, is based on the following rescaling

$$t \to \frac{t}{\varepsilon}, \qquad K(h) \to K(\frac{h}{\varepsilon}) \frac{dh}{\varepsilon},$$

such that (2.14) becomes

$$\begin{cases} \varepsilon \partial_t n_{\varepsilon} = \int_{\mathbb{R}^d} \left(n_{\varepsilon}(t, z + \varepsilon h) - n_{\varepsilon}(t, z) \right) b(z) K(h) dh + R(z, \rho_{\varepsilon}), \\ \rho_{\varepsilon}(t) = \int_{\mathbb{R}^d} n_{\varepsilon}(t, z) dz. \end{cases}$$
(2.15)

With this scaling, which is closely related to the one in (1.3), a Hamilton-Jacobi equation with constrained is derived which allows to capture a concentration phenomenon for n_{ε} , similarly to the one obtained for (1.3) in Chapter 1. However, this rescaling would not be adapted to study the fractional selection-mutation model (2.1). In particular, the limiting Hamilton-Jacobi equation derived in [BMP09], that is

$$\partial_t u(t,z) = \int e^{\nabla u(t,z) \cdot h} K(h) dh + R(z,\rho(t)),$$

will not be well-defined here, because of the algebraic decay of the integral kernel.

The above observations show that in order to capture concentration phenomena in fractional selection-mutations models a new rescaling is required. In [MM15], being inspired by the rescaling for the fractional Fisher-KPP equation introduced in Section 2.2, we provide a rescaling for (2.1) leading to the concentration of the phenotypic density as a Dirac mass. In what follows we show how such rescaling is derived.

Although, for the classical Fisher-KPP equation, the long range and long time rescaling coincides with the one with small diffusion steps and long time, this is not the case for the fractional Fisher-KPP equation. To understand this better, we rewrite (2.6) in a simple case, that is for d = 1, $x \neq 0$ and assuming that the solution n is symmetric such that n(x) = n(-x):

$$\begin{split} \varepsilon \partial_t n_{\varepsilon}(t,x) &= \int_0^\infty \left(n_{\varepsilon}(t,||x|^{1/\varepsilon} + h|^{\varepsilon}) - n_{\varepsilon}(t,||x|^{1/\varepsilon} - h|^{\varepsilon}) - 2n_{\varepsilon}(t,x) \right) \frac{dh}{|h|^{1+2\alpha}} \\ &+ n_{\varepsilon}(t,x) \left(1 - n_{\varepsilon}(x,t) \right) \\ &= |x|^{\frac{-2\alpha}{\varepsilon}} \int_0^\infty \left(n_{\varepsilon} \left(t,|x| + |x|((1+r)^{\varepsilon} - 1)) + n_{\varepsilon} \left(t,|x| + |x|(|1-r|^{\varepsilon} - 1)) - 2n_{\varepsilon}(t,x) \right) \frac{dr}{r^{1+2\alpha}} \\ &+ n_{\varepsilon}(t,x) \left(1 - n_{\varepsilon}(t,x) \right). \end{split}$$

Here we have used the following change of variable:

$$h = |x|^{\frac{1}{\varepsilon}}r$$
, so that $|x|^{\frac{1}{\varepsilon}} \pm h = |x|^{\frac{1}{\varepsilon}}(1\pm r)$.

We observe that this rescaling is heterogeneous in x, and the diffusion steps are rescaled differently at different points.

Since in the context of selection-mutation models, the rescaling is on the size of the mutations and not on the variable z to consider the long range limit, the non homogeneity in the mutation kernel induced by (2.5) is not realistic. Therefore, in this case, we rescale the mutation step instead of the point z but following the same structure of rescaling. More precisely, we consider the following rescaling where the mutation kernel remains independent of z:

$$\varepsilon \partial_t n_{\varepsilon}(t,z) = \int_0^\infty \int_{\nu \in S^{d-1}} \left(n_{\varepsilon}(t,z + ((1+r)^{\varepsilon} - 1)\nu) - n_{\varepsilon}(t,z) \right) \frac{dSdr}{r^{1+2\alpha}} + n_{\varepsilon}(t,z) R(z,\rho_{\varepsilon}(t)).$$
(2.16)

Here dS is the sphere element for the unit sphere. Note that the power of r in the above integral is now $1+2\alpha$ and does not depend on the dimension d anymore because of the change of variable. With this rescaling, where ε is a small parameter, we consider much smaller mutation steps than the scaling considered for the classical case (2.15). Here, the parameter ε appears as a power for the size of mutations, instead of a multiplicative constant. However, the mutation distribution has still algebraic law, but with a large power. Moreover, while the mutation distribution did not have a variance before rescaling, the rescaled mutation distribution has a small variance of order $O(\varepsilon^2)$, similarly to the case of classical selection-mutation (2.15). This property is an important element to obtain the concentration of the phenotypic density with a similar behavior to the one obtained for (2.14) or (1.1). We present in Figure 2.2 the numerical resolution of (2.16) with $\varepsilon = .01$ and compare it to the numerical resolution of (1.3) using the following parameters:

$$R(z,\rho) = 1 + \exp(2 - \frac{z^2}{5}) - \rho, \qquad \varepsilon = 0.01, \qquad n(0,z) = \exp\left(-\frac{(z+2)^2}{0.5\varepsilon}\right). \tag{2.17}$$

We observe that the dynamics of the solutions are qualitatively very close. The solution now concentrates at a dominant trait which evolves gradually and converges ultimately to the optimal trait z = 0.

For our analysis, it is helpful to rewrite the above equation with a change of variable:

 $r = e^k - 1,$

which leads to

$$\begin{cases} \varepsilon \partial_t n_{\varepsilon}(t,z) = \int_0^\infty \int_{\nu \in S^{d-1}} \left(n_{\varepsilon}(t,z+(e^{\varepsilon k}-1)\nu) - n_{\varepsilon}(t,z) \right) \frac{e^k}{|e^k-1|^{1+2\alpha}} dS dk + n_{\varepsilon}(t,z) R(z,\rho_{\varepsilon}(t)) \\ \rho_{\varepsilon}(t) = \int_{\mathbb{R}} n_{\varepsilon}(t,z) dz, \\ n_e(x,0) = n_{\varepsilon}^0(x). \end{cases}$$

$$(2.18)$$

2.3.2 Assumptions

Our analysis is still based on the Hopf-Cole transformation

$$n_{\varepsilon} = \exp\left(\frac{u_{\varepsilon}}{\varepsilon}\right). \tag{2.19}$$



Figure 2.2: Left: the numerical resolution of (1.3) with the parameters given in (2.17) and with $I = \rho$. The colors represent the different values of the phenotypic density n. Right: the numerical resolution of (2.16) with the same parameters. In both figures, the population concentrates at a dominant trait which evolves with time.

Our assumptions are analogous to the ones used in [BMP09] for the model with thin tailed mutation kernels:

We assume that there are two constants $0 < \rho_m < \rho_M < \infty$ such that

$$\min_{z \in \mathbb{R}^d} R(z, \rho_m) = 0, \qquad \max_{z \in \mathbb{R}^d} R(z, \rho_M) = 0.$$
(2.20)

The l.h.s. condition above means that all the traits are viable such that the growth rate of all individuals become positive when the competition is not strong. The r.h.s. condition means that the resources are limited such that the growth rate of all individuals become negative if the competition is too strong.

We also assume that the growth rate is a decreasing function of the competition term, that is there exists a constants $K_1 > 0$ such that, for any $z \in \mathbb{R}^d$, $\rho \in \mathbb{R}$,

$$-K_1 \le \frac{\partial R}{\partial \rho}(z,\rho) \le -K_1^{-1} < 0.$$
(2.21)

We also make the following technical assumption

$$\sup_{\frac{\rho_m}{2} \le \rho \le 2\rho_M} \| R(\cdot, \rho) \|_{W^{2,\infty}(\mathbb{R}^d)} < K_2.$$

$$(2.22)$$

Moreover, we make the following assumptions on the initial data:

 $(u_{\varepsilon}^{0})_{\varepsilon}$ is a sequence of continuous functions which converge in $C_{\text{loc}}(\mathbb{R}^{d})$ to u^{0} , as $\varepsilon \to 0$, (2.23)

and there exists a constant $A < \alpha$ such that

$$n_{\varepsilon}^{0}(z) \le \frac{C_{0}}{\left(C_{1}(1+|z|)\right)^{\frac{2A}{\varepsilon}}},$$
(2.24)

$$\rho_m \le \int_{\mathbb{R}^d} n_{\varepsilon}^0(z) dz \le \rho_M.$$
(2.25)
2.3.3 Main results

Our main result is the following.

Theorem 2.3. Let n_{ε} be the solution of (2.18) and $u_{\varepsilon} = \varepsilon \log n_{\varepsilon}$. Assume (2.20)–(2.25). Then, along subsequences as $\varepsilon \to 0$, (ρ_{ε}) converges a.e. to ρ and (u_{ε}) converges locally uniformly to a function u which is Lipschitz continuous with respect to z and continuous in t, such that

 $\|D_z u\|_{L^{\infty}(\mathbb{R}^d \times \mathbb{R}^+)} \le 2\alpha, \qquad u(t, z+h) - u(t, z) \le 2\alpha \log(1+|h|), \qquad \text{for all } z, h \in \mathbb{R}^d.$ (2.26)

Moreover, u is a viscosity supersolution to the following equation

$$\begin{cases} \partial_t u - \int_0^\infty \int_{\nu \in S^{d-1}} \left(e^{kD_z u \cdot \nu} - 1 \right) \frac{e^k dS dk}{|e^k - 1|^{1+2\alpha}} = R(z, \rho), \\ u(z, 0) = u^0(z). \end{cases}$$
(2.27)

For fixed ρ , u is indeed the minimal viscosity supersolution of (2.27) satisfying (2.26). Moreover, u satisfies the following constraint

$$\max_{z \in \mathbb{R}^d} u(t, z) = 0, \qquad \text{for all } t > 0.$$
(2.28)

It is also a viscosity subsolution of (2.27) in the following weak sense. Let's $\varphi \in C^2(\mathbb{R}^+ \times \mathbb{R}^d)$ be a test function such that $u - \varphi$ takes a maximum at (t_0, z_0) and

$$\varphi(t, z+h) - \varphi(t, z) \le (2\alpha - \xi) \log(1+|h|), \quad \text{for all } (t, z) \in B_r(t_0, z_0) \text{ and } h \in \mathbb{R}^d,$$

with r and ξ positive constants. Then, we have

$$\partial_t \varphi(t_0, z_0) - \int_0^\infty \int_{\nu \in S^{d-1}} \left(e^{k D_z \varphi(t_0, z_0) \cdot \nu} - 1 \right) \frac{e^k dS dk}{|e^k - 1|^{1 + 2\alpha}} \le \limsup_{s \to t_0} R(z, \rho(s)).$$

A main difficulty in this convergence result is that the Hamiltonian in the above Hamilton-Jacobi equation can take infinite values and that the limit is not necessarily a viscosity solution to (2.27). Another difficulty comes from the fact that the term $\rho(t)$ is only BV and potentially discontinuous.

The above Hamilton-Jacobi equation can indeed take infinite values when $|D_z u| \geq 2\alpha$. This property leads to the Lipschitz regularity result in (2.26). The second regularity result in (2.26), which is stronger than the first one, is not a characteristic of a solution to (2.27) and it is not generally satisfied by a Hamilton-Jacobi equation of this type. This property comes from the original problem with ε . These regularity properties do not hold necessarily at the initial time, which means that a strong regularizing effect is associated with this problem.

Note that in Theorem 2.3 we do not characterize the limit u as a viscosity solution to a Hamilton-Jacobi equation with constraint, as was the case in the previous results on such selection-mutation models (see for instance [DJMP05, PB08, BMP09]). We only prove that u is the minimal viscosity supersolution to (2.27) satisfying (2.26) and a viscosity subsolution in a weak sense. One can wonder if u is indeed a viscosity solution to (2.27). We do not expect this assertion to be true in general. In [Mir], we provide an example of a Hamilton-Jacobi equation of similar type which has a solution that does not satisfy the second inequality in (2.26). Existence of such solutions together with the uniqueness of viscosity solutions to Hamilton-Jacobi equations of type (2.27), with fixed ρ , (see [BC13], Section 6) indicates that u might not be a viscosity solution of (2.27) in general. Note that, of course, thanks to the comparison principle for fixed ρ , u is always greater than (or equal to) the unique viscosity solution of (2.27).

The information obtained in Theorem 2.3 still allows to obtain the concentration of the population's density as Dirac masses, analogously to the previous works [PB08, BMP09]:

Theorem 2.4. Let n_{ε} be the solution of (2.18). Assume (2.20)–(2.25). Then, along subsequences as $\varepsilon \to 0$, n_{ε} converges in $L^{\infty}(w*(0,\infty); \mathcal{M}^{1}(\mathbb{R}^{d}))$ to a measure n, such that,

$$supp n(t, \cdot) \subset \{z \mid u(t, z) = 0\}, \quad for \ a.e. \ t.$$

Moreover, for all continuous points of $\rho(t)$, we have

$$\{z \mid u(t,z) = 0\} \subset \{z \mid R(z,\rho(t)) = 0\}.$$

In particular, if $z \in \mathbb{R}$ and R is monotonic with respect to z, then for a.e. t > 0,

$$n(t,z) = \rho(t) \,\delta(z - \overline{z}(t)).$$

2.3.4 The main elements of the proof

We replace the Hopf-Cole transformation (2.19) in (2.18):

$$\partial_t u_{\varepsilon}(t,z) = \int_0^\infty \int_{\nu \in S^{d-1}} (e^{\frac{u_{\varepsilon}(t,z+(e^{\varepsilon k}-1)\nu)-u_{\varepsilon}(t,z)}{\varepsilon}} - 1) \frac{e^k}{|e^k - 1|^{1+2\alpha}} dS dk + R(z,\rho_{\varepsilon}(t)).$$

This equation converges formally to (2.27). The following properties would allow to obtain the convergence of the integral term:

$$\|D_z u_{\varepsilon}\|_{L^{\infty}(\mathbb{R}\times\mathbb{R}^+)} < 2\alpha, \qquad u_{\varepsilon}(t,z+h) - u_{\varepsilon}(t,z) < 2\alpha \log(1+|h|)$$

We prove these properties (with non-strict inequalities as given by (2.26)) and the convergence of (u_{ε}) simultaneously.

To prove the convergence of (u_{ε}) we use the method of semi-relaxed limits [BP88] in the theory of viscosity solutions. However, since the Hamiltonian in (2.27) takes infinite values and since the limit u is not in general a viscosity solution of (2.27), we cannot use this method in a classical manner and further work is required. This issue is indeed closely related to the work in [BC13] where a large deviation type result has been obtained for a Lévy type nonlocal operator where the integral kernel has at most exponential tails. In the case where the integral kernel has exponential tails, a Hamilton-Jacobi equation close to (2.27), without the growth term, is obtained at the limit. However, in that case the function obtained at the limit does not satisfy necessarily the second regularity result in (2.26) and it is indeed a viscosity solution to the Hamilton-Jacobi equation.

To provide a main idea in the proof, we first recall the classical method of semi-relaxed limits. We first define the semi-relaxed limits:

$$\overline{u}(t,x) = \limsup_{\substack{(s,y) \to (t,x) \\ \varepsilon \to 0}} u_{\varepsilon}(s,y), \qquad \underline{u}(t,x) = \liminf_{\substack{(s,y) \to (t,x) \\ \varepsilon \to 0}} u_{\varepsilon}(s,y).$$

With the classical method, one would prove that \overline{u} and \underline{u} are respectively sub and supersolution to the limit Hamilton-Jacobi equation (that is (2.27) here). Such properties usually hold because

of the stability properties of the viscosity solutions. Then, if the problem admits a strong comparison principle (for discontinuous solutions), we can deduce that

 $\overline{u} \leq \underline{u}.$

However, by definition, we also have

 $\underline{u} \leq \overline{u}.$

Combining the above inequalities we deduce that $\underline{u} = \overline{u}$ and (u_{ε}) converges locally uniformly to $u := \underline{u} = \overline{u}$.

Here, since the Hamiltonian takes infinite values and since the limit u is not necessarily a subsolution to (2.27) we cannot expect to prove that \overline{u} is a subsolution to (2.27). However, we can prove that \underline{u} is a viscosity supersolution to (2.27) and it satisfies (2.26). We also prove that \overline{u} is a viscosity subsolution to (2.27) in the weak sense as defined in Theorem 2.4. We then modify \underline{u} to make it smooth and an admissible test function such that we can compare it with \overline{u} to obtain that $\overline{u} \leq \underline{u}$.

2.4 Fractional Fisher-KPP equation in a periodically hostile environment

2.4.1 Model and motivations

We study the following equation

$$\begin{cases} \partial_t n + (-\Delta)^{\alpha} n = n(1-n) & \text{in } (0,\infty) \times \Omega, \\ n = 0 & \text{in } (0,\infty) \times \Omega^c, \\ n(0,x) = n_0(x), \end{cases}$$
(2.29)

with A a positive constant and

$$\Omega = \bigcup_{k \in \mathbb{Z}} (2kA, (2k+1)A).$$

Equation (2.29) models the growth and the invasion of a species subject to a non-local dispersion in a periodically hostile environment. The function n stands for the density of the population. The fractional Laplacian describes the motions of individuals, it takes into account the possibility of "jump" (move rapidly) of individuals from one point to another with a high probability, for instance because of the wind for seeds and polens or human transportation for animals. The logistic term n(1-n) represents the growth rate of the population. The originality of this model is the following, we forbid our species to invade some periodic patches and thus the reachable areas are disconnected. This is for instance to consider the presence of obstacles as sees, lakes, or urban zones. Here, we assume that the regions where the species can develop itself are homogeneous. Thanks to the non-local diffusion (which models the "jumps"), the species will invade all the "good" patches and the solution will converge to a non-null stable stationary state with a speed which grows exponentially fast.

Many works deal with the case of a standard diffusion with homogenous or heterogeneous environment (see [Fis37, KPP37, AW78, GF79]). More close to this work, Guo and Hamel in [GH12]

treat a Fisher-KPP equation with periodically hostile regions and a standard diffusion. The authors prove that the stable state will invade the unstable state in the connected component of the initial data. In our work, thanks to the non-local character of the fractional Laplacian, contrary to what happens in [GH12], we show that there exists a unique non-trivial positive bounded stationary state. Moreover this stationary state invades the unstable state 0 everywhere and not only on the connected component of the initial data. Existence and uniqueness of steady solutions to fractional KPP type equations in periodic media, without obstacles, was stablished in [BRR11]. Next, in [CCR12] it was proved that such steady solution invades the unstable state 0 with an exponential speed (see also [L19, ST19] for alternative proofs). The work in this section extends those results to the case of periodically hostile environments.

2.4.2 Assumptions and main results

We assume that

$$A \ge A_0,\tag{2.30}$$

where A_0 is such that for all $A \ge A_0$ the principal eigenvalue λ_A of the Dirichlet operator $(-\Delta)^{\alpha} - Id$ in (0, A) is negative (the existence of such A_0 is provided by [BRR11]). In other words, there exists a function φ_A and a constant λ_A such that

$$\begin{cases} ((-\Delta)^{\alpha}\varphi_A - \varphi_A = \lambda_A \varphi_A, & \text{in } (0, A), \\ \varphi_A = 0, & \text{in } \mathbb{R} \setminus (0, A), \\ \varphi_A \ge 0, \lambda_A < 0. \end{cases}$$

We also define λ_0 to be the principal eigenvalue to the problem in Ω :

$$\begin{cases} ((-\Delta)^{\alpha}\varphi_0 - \varphi_0 = \lambda_0\varphi_0, & \text{in } \Omega, \\ \varphi_0 = 0, & \text{in } \Omega^c, \\ \varphi_0 \ge 0. \end{cases}$$

We prove the following.

Theorem 2.5. Under the assumption (2.30), there exists a unique positive and bounded stationary state n_+ to (2.29). Moreover, we have $0 \le n_+ \le 1$ and n_+ is 2A-periodic.

The existence is due to the negativity of the principal eigenvalue of the Dirichlet operator $(-\Delta)^{\alpha}$ – Id in (0, A) which allows to construct by an iterative method a stationary state. To prove the uniqueness, we first prove that thanks to the non-local character of the fractional Laplacian, all the bounded stationary states behave like

$$\delta(x) = \operatorname{dist}(x, \partial \Omega)^{\alpha} \mathbf{1}_{\Omega}(x).$$

Hence, thanks to the maximum principle and the fractional Hopf Lemma (see [GS16]), we obtain the result. We should underline that the uniqueness is clearly due to the non-local diffusion operator and it does not hold in the case of a standard diffusion term ($\alpha = 1$).

We next focus on the propagation phenomena. Let's first recall the state of the art on related problems. The question of propagation in a Fisher-KPP type equation involving a fractional Laplacian in a constant environment was first treated by X. Cabré and J.-M. Roquejoffre in [CR13]. They proved that the front position is exponential in time. Next, X. Cabré, A. Coulon and J.-M. Roquejoffre proved in [CCR12] the convergence to a stationary state with an exponential speed in a periodic heterogeneous environment. Thanks to the asymptotic approach introduced in Subsection 2.2, in [MM15] with S. Méléard we proved the result of an exponential speed of propagation in a constant environment. More recently, [BGHP18, ST19, L19] extended this approach to derive the speed of propagation for different non-local operators in the case of a homogeneous environment for [BGHP18] and a periodic environment for [ST19, L19].

To study the present model we followed the general idea introduced in [MM15] and Section 2.2. We expect that in large time the propagation front is located in

 $\mathcal{B} = \{(t, x) \in \mathbb{R}^+ \times \mathbb{R} \mid (1 + 2\alpha) \log |x| < \lambda_0 t\}.$

We then perform the long time-long range rescaling (2.5) to capture the effective behavior of the solution. We thus define

$$n_{\varepsilon}(t,x) = n(\frac{t}{\varepsilon}, |x|^{1/\varepsilon} \frac{x}{|x|}), \qquad n_{\varepsilon,+}(t,x) = n_{+}(\frac{t}{\varepsilon}, |x|^{1/\varepsilon} \frac{x}{|x|}).$$

To present our result we introduce the following notations, for $\nu, \varepsilon > 0$,

$$\Omega_{\nu} = \{ x \in \Omega \, | \, \operatorname{dist}(x, \partial \Omega) > \nu \}, \qquad \Omega^{\varepsilon} = \{ x \in \mathbb{R} \, | \, |x|^{1/\varepsilon} \frac{x}{|x|} \in \Omega \}.$$

We also make the following assumption on the initial condition

$$n_0 \in C^0(\mathbb{R}^+ \times \mathbb{R}) \cap C_c^\infty(\mathbb{R}^+ \times \Omega), \qquad n_0 \neq 0.$$
(2.31)

This is our main result:

Theorem 2.6. Assume (2.30) and (2.31). Then, as $\varepsilon \to 0$, (i) n_{ε} converges locally uniformly to 0 in $\{(t,x) \in \mathbb{R}^+ \times \mathbb{R} \mid (1+2\alpha) \log |x| > \lambda_0 t\}$. (ii) $\frac{n_{\varepsilon}}{n_{\varepsilon,+}}$ converges locally uniformly to 1 in $((0,\infty) \times \Omega^{\varepsilon}) \cap B$.

A direct consequence of this result is the invasion of the unstable state by the stable state:

Corollary 2.7. Assume (2.30) and (2.31). Then, for all
$$\nu > 0$$
,
(i) $\lim_{t \to +\infty} n(t,x) = 0$ uniformly in $\{(t,x) \mid |x| \ge e^{Ct}\}$, for all $C > \frac{|\lambda_0|}{1+2\alpha}$.
(ii) $\lim_{t \to +\infty} n(t,x) = n_+(x)$ uniformly in $\{(t,x) \mid |x| \le e^{ct}\} \cap (0,\infty) \times \Omega_{\nu}$, for all $c < \frac{|\lambda_0|}{1+2\alpha}$.

Note that the main new difficulty to study this problem comes from the Dirichlet boundary condition. We modify the previous sub and supersolutions introduced for the KPP equation in Section 2.2 carefully and in particular by multiplying by the principal eigenfunction associated to the operator $(-\Delta)^{\alpha} - Id$ in $\Omega_{\pm\nu}$. Note that these sub and supersolutions have algebraic decay at infinity. Obviously, we cannot put the subsolution below n_0 . However, using heat kernel estimates, we are able to put the subsolution below the solution n for $t \geq 1$. Finally, to prove the convergence of n_{ε} , we use the method of perturbed test functions from the theory of viscosity solutions and homogenization (introduced by Evans [Eva89, Eva92]).

Chapter 3

Evolutionary adaptation in space-heterogeneous environments

3.1 Introduction

This chapter is devoted to the study of the evolutionary dynamics of a population in a heterogeneous environment with a discrete spatial structure. Spatially heterogeneous selection is ubiquitous and constitutes a potent evolutionary force that promotes the emergence and the maintenance of biodiversity. Spatial variation in selection can yield adaptation to local environmental conditions, however, other evolutionary forces like migration and mutation tend to homogenize the spatial patterns of differentiation and thus to impede the build up of local adaptation. Understanding the balance between these contrasted evolutionary forces is a major objective of evolutionary biology theory [Sla78, Whi15, SLM13]) and could lead to a better understanding of the speciation process and the evolutionary response to global change [DD03, LDD08].

In the previous chapters we saw that an approach based on Hamilton-Jacobi equations with constraint allows to describe the evolutionary dynamics of phenotypically structured populations in homogeneous environments. Is it possible to extend this approach to study heterogeneous environments? Are the limiting solutions still in the form of sums of Dirac masses? Does the heterogeneity lead to polymorphic situations? In this chapter we extend the Hamilton-Jacobi approach to the case of spatially heterogeneous environments. In chapter 4 we will focus on time heterogeneous environments.

A main part of the work presented in this chapter has been done in the framework of a collaboration project with S. Gandon who is a biologist. The objective of this project was to show, via the particular model considered in this work, how the Hamilton-Jacobi approach can be used to bring insight into current biological questions.

We study the following system, for $z \in \mathbb{R}^d$,

$$\begin{cases} -\varepsilon^2 \Delta n_{\varepsilon,1}(z) = n_{\varepsilon,1}(z) R_1(z,\rho_{\varepsilon,1}) + m_2 n_{\varepsilon,2}(z) - m_1 n_{\varepsilon,1}(z), \\ -\varepsilon^2 \Delta n_{\varepsilon,2}(z) = n_{\varepsilon,2}(z) R_2(z,\rho_{\varepsilon,2}) + m_1 n_{\varepsilon,1}(z) - m_2 n_{\varepsilon,2}(z), \\ \rho_{\varepsilon,i} = \int_{\mathbb{R}^d} n_{\varepsilon,i}(z) dz, \quad \text{for } i = 1, 2. \end{cases}$$

$$(3.1)$$

This system represents the equilibrium of a population that is structured by a phenotypic trait

30

z, and which is subject to selection, mutation and migration between two habitats. We denote by $n_{\varepsilon,i}(z)$ the phenotypic density, and by $\rho_{\varepsilon,i}$ the total population size in habitat i. The term $R_i(z, \rho_{\varepsilon,i})$ represents the growth rate of the individuals with trait z in habitat i and it is a decreasing function of the total population size $\rho_{\varepsilon,i}$ taking into account in this way a local competition between individuals. The nonnegative constants m_i are the migration rates between the habitats. The diffusion terms correspond to the mutations and ε is an effective size of the mutations.

In this work, we study the asymptotic behavior of (3.1), as $\varepsilon \to 0$. Using the Hamilton-Jacobi approach, we characterize the solution of (3.1) for ε small. we show that the population's phenotypical distribution has at most two peaks and we give explicit conditions under which the population will be monomorphic (unimodal distribution) or dimorphic (bimodal distribution). More importantly, we determine the dominant terms of the population's distribution in each case. This analysis improves the previous results in the biological literature and allows to consider less restrictive assumptions (see for instance [MCG97, Day00, YG09, DRG13, DYG15]). Furthermore, our work makes a link between the theory of adaptive dynamics and moment based approaches in quantitative genetics. Adaptive dynamics and quantitate genetics, which focus respectively on the evolution of discrete and continuous characters, are two theories which are widely used in theoretical evolutionary biology (see for instance [MHM⁺96, BÖ0, Die04, Ric04]). Finally, our method allows to go beyond the Gaussian approximation commonly used in the theory of quantitative genetics.

Closely related models describing the balance between selection, mutation and migration between habitats have indeed been studied using the theories of adaptive dynamics (see for instance [MCG97, Day00, FMP⁺]) and quantitative genetics (see for instance [HDT01, RK01]). In the adaptive dynamics approach, one considers that the mutations are very rare such that the population has time to attain its equilibrium between two mutations and hence the population's distribution has discrete support (one or two points in a two habitats model). The quantitative genetics approach allows more frequent mutations and does not separate the evolutionary and the ecological time scales so that the population's distribution is continuous (see [Ric04]–chapter 7). In this class of works, models closely related to (3.1) are studied based on a main assumption that one considers that the population's distribution is a gaussian [HDT01, RK01] or, to take into account the possibility of dimorphic populations, a sum of one or two gaussian distributions [YG09, DRG13].

In our work, as in the quantitative genetics framework, we also consider continuous phenotypic distributions. However, we don't assume any a priori gaussian assumption. We compute directly the population's distribution and in this way we correct the previous approximations. The asymptotic analysis of (3.1) is closely related to the identification of the evolutionary stable strategy (ESS) (see Section 3.2.1 for the definition) in adaptive dynamics. To perform such asymptotic analysis, we indeed provide also some new results in the framework of adaptive dynamics and in particular, we generalize previous results on the identification of the ESS to the case of non-symmetric habitats.

In [Mir13] a general convergence result (allowing also more than two habitats) has been provided and a Hamilton-Jacobi equation has been derived at the limit considering general form of growth rate R. However, to provide more precise results, in [Mir17, MG] we restricted the study to $z \in \mathbb{R}$ and to the following particular form of growth rate:

$$R_i(z,\rho_i) = r_i - g_i(z-\theta_i)^2 - \kappa_i \rho_i, \quad \text{with } \theta_1 = -\theta \text{ and } \theta_2 = \theta.$$
(3.2)

Here, r_i represents the maximum intrinsic growth rate, the positive constant g_i is the strength of the selection, θ_i is the optimal trait in habitat *i* and the positive constant κ_i represents the intensity of the competition.

In this chapter we will present the results for this particular form of growth rate to provide the most precise results.

Assumptions:

To guarantee that the population does not get extinct, we assume that

$$\max(r_1 - m_1, r_2 - m_2) > 0. \tag{3.3}$$

Moreover, in the first part of this work, we assume that there is positive migration rate in both directions, i.e.

$$m_i > 0, \qquad i=1,2.$$
 (3.4)

The source and sink case, where for instance $m_2 = 0$, will also be analyzed.

The main elements of the method:

To describe the solutions $n_{\varepsilon,i}(z)$ we use a WKB transformation

$$n_{\varepsilon,i}(z) = \frac{1}{\sqrt{2\pi\varepsilon}} \exp\left(\frac{u_{\varepsilon,i}(z)}{\varepsilon}\right).$$
(3.5)

Note that a first approximation that is commonly used in the theory of 'quantitative genetics', is a gaussian distribution of the following form

$$n_{\varepsilon,i}(z) = \frac{\rho_i}{\sqrt{2\pi\varepsilon\sigma}} \exp\left(\frac{-(z-z^*)^2}{\varepsilon\sigma^2}\right) = \frac{1}{\sqrt{2\pi\varepsilon}} \exp\left(\frac{-\frac{1}{2\sigma^2}(z-z^*)^2 + \varepsilon\log\frac{\rho_i}{\sigma}}{\varepsilon}\right).$$

Here, we try to go further than this a priori gaussian assumption and to approximate directly $u_{\varepsilon,i}$. To this end, we write an expansion for $u_{\varepsilon,i}$ in terms of ε :

$$u_{\varepsilon,i} = u_i + \varepsilon v_i + O(\varepsilon^2). \tag{3.6}$$

By replacing (3.6) and (3.5) in (3.1), we first prove that $u_1 = u_2 = u$ is the unique viscosity solution to a Hamilton-Jacobi equation with constraint which can be computed explicitly. The uniqueness of solution of such Hamilton-Jacobi equation with constraint is related to the uniqueness of the ESS and to the weak KAM theory ([Lio82a], Chapter 5 and [Fat16]). Such function u indeed satisfies

$$\max_{m} u(z) = 0$$

with the maximum points attained at one or two points corresponding to the ESS points of the problem. We next compute formally the next order terms v_i . Note that once u and v_i are known we have already a good approximation of the population's distribution with an error of order $O(\sqrt{\varepsilon})$. We then notice that, while u(z) < 0, $n_{\varepsilon,i}(z)$ is exponentially small. Therefore, to estimate the moments of the population's distribution only the values of u and v_i at the points which are close to the zero level set of u matter, i.e. the ESS points. Using the Laplace's method of integration and the Taylor expansions of the terms u and v_i around the ESS points we provide indeed some analytic approximations of the moments of the population's distribution with an error of order ε^2 as $\varepsilon \to 0$. Using the method presented in this chapter, a detailed biological study of this model has been provided in [MG]. This analysis allows to observe new features. In particular, the computation of the correctors, v_i is important to estimate the mean phenotypic trait. This computation allows indeed to capture a gap between the mean phenotypic traits in the two habitats in presence of mutations, in a case where in absence of the mutations the population is monomorphic, i.e. the distribution concentrates on a same single point in the two habitats (see Subsection 3.5 for more details). Moreover, the analytical approximation of the moments is of particular interest in the biological community and there have been several attempts to provide such approximations (see for instance [YG09, DRG13, DYG15]). Our work provides a more mathematically rigorous method for such computations and improves some previous results. In particular, we precise the impact of the migration rate, and other parameters of the model, on the variance of the population's distribution (see [MG] and Subsection 3.5). This is specially interesting since in classical quantitative genetics models, it is not clear how to include the influence of migration in the variance of the population's distribution that one generally assumes to be fixed. In such models the variance is usually under estimated (see for instance [DRG13]).

Note that the part of the method where one proves the convergence of $(u_{\varepsilon,i})_{\varepsilon}$ to a viscosity solution of a Hamilton-Jacobi equation is very robust and can be generalized to the case of several habitats with multi-dimensional trait and general growth rates R_i (see [Mir13]). However, we use the particular structure of the model to obtain the uniqueness and the identification of the ESS. This is the reason why in [Mir17, MG] we consider a particular growth rate (3.2), so that we can highlight the usefulness of the Hamilton-Jacobi approach for the cases where the adaptive dynamics framework is well-understood. Note however that although with this choice of model, in absence of migration, the asymptotic analysis in each habitat is rather simple (in each habitat the population will concentrate on the fittest trait in the patch), the presence of the migration makes the qualitative study non-trivial. The migration can indeed lead to dimorphism in each of the habitats. In particular a Gaussian approximation would not be adapted.

In Section 3.2 we present our main results in the case where there is migration in both directions, that is under assumption (3.4). In Section 3.3 we provide our main results in a source and sink scenario, that is with $m_2 = 0$. The works presented in Section 3.2 and in Section 3.3 are based on the results in [Mir13, Mir17]. In Section 3.4 we provide some heuristic computations that allow to understand our main results but also to compute formally the corrector terms and to provide analytic approximations of the moments of the phenotypic distribution. Finally in Section 3.5 we provide our analytic approximations of the moments of the phenotypic distribution for two biological examples and discuss the biological results [MG].

3.2 Migration in both directions

In this section, we study the case where there is migration in both directions, that is we assume (3.4). In Subsection 3.2.1 we provide some notions from the theory of adaptive dynamics. Note that our objective is not to study the framework of adaptive dynamics where the mutations are

assumed to be very rare. However, these notions appear naturally from our asymptotic computations. Our analysis of the integro-differential model (3.1) is indeed based on a preliminary analysis in the adaptive dynamics framework. In Subsection 3.2.2 we present the main results in the adaptive dynamics framework. Finally, in Subsection 3.2.3 we provide our main results on the derivation of the zero order term u.

3.2.1 Some notions from the theory of adaptive dynamics

In this subsection, we introduce some notions from the theory of adaptive dynamics [MCG97] that we will be using in the next subsections.

Effective fitness: The effective fitness $W(z; \rho_1, \rho_2)$ is the largest eigenvalue of the following matrix:

$$\mathcal{A}(z;\rho_1,\rho_2) = \begin{pmatrix} R_1(z;\rho_1) - m_1 & m_2\\ m_1 & R_2(z;\rho_2) - m_2 \end{pmatrix},$$
(3.7)

that is

$$W(z;\rho_1,\rho_2) = \frac{1}{2} \Big[(R_1(z;\rho_1) + R_2(z;\rho_2) - m_1 - m_2) \\ + \sqrt{(R_1(z;\rho_1) - R_2(z;\rho_2) - m_1 + m_2)^2 + 4m_1m_2)} \Big].$$

This indeed corresponds to the *effective* growth rate associated with trait z in the whole metapopulation when the total population sizes are given by (ρ_1, ρ_2) .

Demographic equilibrium: Consider a set of points $\Omega = \{z_1, \dots, z_m\}$. The demographic equilibrium corresponding to this set is given by $(n_1(z), n_2(z))$, with the total population sizes (ρ_1, ρ_2) , such that

$$n_i(z) = \sum_{j=1}^m \alpha_{i,j} \delta(z - z_j), \quad \rho_i = \sum_{j=1}^m \alpha_{i,j}, \quad W(z_j, \rho_1, \rho_2) = 0, \quad i = 1, 2, \ j = 1, \cdots, m_i$$

and such that $(\alpha_{1,j}, \alpha_{2,j})^T$ is the right eigenvector associated with the largest eigenvalue $W(z_j, \rho_1, \rho_2) = 0$ of $\mathcal{A}(z_j; \rho_1, \rho_2)$.

Invasibility: We say that a mutant trait z_m can invade a resident strategy $\{\overline{z}\}$ at its demographic equilibrium $(\overline{\rho}_1, \overline{\rho}_2)$ if $W(z_m, \overline{\rho}_1, \overline{\rho}_2) > 0$.

Evolutionary stable strategy: A set of points $\Omega^* = \{z_1^*, \dots, z_m^*\}$ is called an evolutionary stable strategy (ESS) if

$$W(z, \rho_1^*, \rho_2^*) = 0$$
, for $z \in \mathcal{A}$ and, $W(z, \rho_1^*, \rho_2^*) \le 0$, for $z \notin \mathcal{A}$,

where ρ_1^* and ρ_2^* are the total population sizes corresponding to the demographic equilibrium associated with the set Ω^* . Note that this definition corresponds indeed to globally evolutionary stable strategies. Here, we are interested only in such global ESS (see for instance [Die04] for the definition of local ESS).

Notation: We will use the star sign * whenever we talk about an evolutionary stable strategy Ω^* (and similarly for the corresponding demographic equilibrium (n_1^*, n_2^*) and the total population sizes (ρ_1^*, ρ_2^*)). We add an index M when the strategy is monomorphic (a set of a single

trait $\{z^{M*}\}$ with the corresponding demographic equilibrium (n_1^{M*}, n_2^{M*}) , and the total population sizes $(\rho_1^{M*}, \rho_2^{M*}))$ and an index D when the strategy is dimorphic (a set of two traits $\{z_1^{D*}, z_{II}^{D*}\}$ with the corresponding demographic equilibrium (n_1^{D*}, n_2^{D*}) , and the total population sizes $(\rho_1^{D*}, \rho_2^{D*}))$.

3.2.2 The main results in the adaptive dynamics framework

Our main result in the adaptive dynamics framework is that there exists a unique ESS which is whether monomorphic (a single Dirac mass) or dimorphic (a sum of two Dirac masses). We determine indeed under which conditions the ESS is monomorphic or dimorphic. To state our result, we first define

$$z^{D*} = \sqrt{\theta^2 - \frac{m_1 m_2}{4\theta^2 g_1 g_2}}, \qquad \rho_1^{D*} = \frac{\frac{m_1 m_2}{4\theta^2 g_2} + r_1 - m_1}{\kappa_1}, \qquad \rho_2^{D*} = \frac{\frac{m_1 m_2}{4\theta^2 g_1} + r_2 - m_2}{\kappa_2}$$

Theorem 3.1. Assume (3.3)–(3.4). Then, there exists a unique set of points Ω^* which is an ESS.

(i) The ESS is dimorphic if and only if

$$\frac{m_1 m_2}{4g_1 g_2 \theta^4} < 1, \tag{3.8}$$

$$0 < m_2 \rho_2^{D*} + (R_1(-z^{D*};\rho_1^{D*}) - m_1)\rho_1^{D*}, \qquad (3.9)$$

and

$$0 < m_1 \rho_1^{D*} + (R_2(z^{D*}; \rho_2^{D*}) - m_2)\rho_2^{D*}.$$
(3.10)

Then the dimorphic equilibrium is given by

$$n_i^{D*} = \nu_{\mathrm{I},i}\delta(z+z^{D*}) + \nu_{\mathrm{II},i}\delta(z-z^{D*}), \quad \nu_{\mathrm{I},i} + \nu_{\mathrm{II},i} = \rho_i^{D*}, \quad i = 1, 2.$$
(3.11)

(ii) If the above conditions are not satisfied then the ESS is monomorphic. In the case where condition (3.8) is verified but the r.h.s. of (3.9) (respectively (3.10)) is negative, the fittest trait belongs to the interval $(-\theta, -z^{D*})$ (respectively (z^{D*}, θ)). If (3.8) is satisfied but (3.9) (respectively (3.10)) is an equality then the monomorphic ESS is given by $\{-z^{D*}\}$ (respectively $\{z^{D*}\}$).

Note that one can compute the weights $\nu_{k,i}$, for k = I, II and i = 1, 2:

$$\begin{pmatrix} \nu_{\mathrm{I},1} \\ \nu_{\mathrm{I},2} \end{pmatrix} = \frac{m_1 \rho_1^{D^*} + (R_2(z^{D^*};\rho_2^{D^*}) - m_2)\rho_2^{D^*}}{m_1 m_2 - \left(R_1(-z^{D^*};\rho_1^{D^*}) - m_1\right) \left(R_2(z^{D^*};\rho_2^{D^*}) - m_2\right)} \begin{pmatrix} m_2 \\ -R_1(-z^{D^*};\rho_1^{D^*}) + m_1 \end{pmatrix}, \\ \begin{pmatrix} \nu_{\mathrm{II},1} \\ \nu_{\mathrm{II},2} \end{pmatrix} = \frac{m_2 \rho_2^{D^*} + (R_1(-z^{D^*};\rho_1^{D^*}) - m_1)\rho_1^{D^*}}{m_1 m_2 - \left(R_1(-z^{D^*};\rho_1^{D^*}) - m_1\right) \left(R_2(z^{D^*};\rho_2^{D^*}) - m_2\right)} \begin{pmatrix} -R_2(z^{D^*};\rho_2^{D^*}) + m_2 \\ m_1 \end{pmatrix}.$$

The above theorem leads to the following corollary which will be helpful thereafter.

Corollary 3.2. Assume that

$$m_2\rho_2^{D*} + (R_1(-z^{D*};\rho_1^{D*}) - m_1)\rho_1^{D*} \neq 0, \qquad m_1\rho_1^{D*} + (R_2(z^{D*};\rho_2^{D*}) - m_2)\rho_2^{D*} \neq 0, \quad (3.12)$$

and let the set Ω^* be the unique ESS of the model and (ρ_1^*, ρ_2^*) be the total population sizes at the demographic equilibrium of this ESS. Then,

$$W(z, \rho_1^*, \rho_2^*) < 0, \quad \text{for all } z \in \mathbb{R} \setminus \Omega^*.$$

Note also that when the habitats are symmetric, then conditions (3.9) and (3.10) always hold under condition (3.8), and hence

Corollary 3.3. Assume that the habitats are symmetric:

$$r = r_1 = r_2, \quad g = g_1 = g_2, \quad \kappa = \kappa_1 = \kappa_2, \quad m = m_1 = m_2.$$
 (3.13)

(i) Then the unique ESS is dimorphic if and only if

$$\frac{m}{2g} < \theta^2. \tag{3.14}$$

The dimorphic ESS is determined by (3.11).

(ii) When condition (3.14) is not satisfied, then the ESS is monomorphic and the corresponding monomorphic equilibrium is given by

$$n_1^{M*}(z) = n_2^{M*}(z) = N^{M*} \delta(z), \quad with \ N^{M*} = \frac{1}{\kappa} \left(r - g\theta^2 \right).$$

The next proposition gives an interpretation of conditions (3.9) and (3.10).

Proposition 3.4. Assume that condition (3.8) is satisfied and that $r_i - m_i > 0$, for i = 1, 2. Then,

(i) condition (3.9) holds if and only if a mutant trait of type z^{D*} can invade a monomorphic resident population of type $-z^{D*}$ which is at it's demographic equilibrium.

(ii) condition (3.10) holds if and only if a mutant trait of type $-z^{D*}$ can invade a monomorphic resident population of type z^{D*} which is at it's demographic equilibrium.

One can indeed rewrite conditions (3.9) and (3.10) respectively as below

$$C_1 < \alpha_2 r_2 - \alpha_1 r_1, \qquad C_2 < \beta_1 r_1 - \beta_2 r_2,$$

with C_i , α_i and β_i constants depending on m_1 , m_2 , g_1 , g_2 , κ_1 , κ_2 and θ . These conditions are indeed a measure of asymmetry between the habitats. They appear from the fact that even if condition (3.8), which is the only condition for dimorphism in symmetric habitats, is satisfied, while the quality of the habitats are very different, the ESS cannot be dimorphic. In this case, the population will be able to adapt only to one of the habitats and it will be maladapted to the other one.

3.2.3 Derivation of the zero order terms u_i

The identification of the zero order terms u_i is based on the following result.

Theorem 3.5. Assume (3.3)-(3.4).

(i) As $\varepsilon \to 0$, $(n_{\varepsilon,1}, n_{\varepsilon,2})$ converges to (n_1^*, n_2^*) , the demographic equilibrium of the unique ESS of the model. Moreover, as $\varepsilon \to 0$, $\rho_{\varepsilon,i}$ converges to ρ_i^* , the total population size in habitat i corresponding to this demographic equilibrium.

(ii) As $\varepsilon \to 0$, both sequences $(u_{\varepsilon,i})_{\varepsilon}$, for i = 1, 2, converge along subsequences and locally uniformly in \mathbb{R} to a continuous function $u \in C(\mathbb{R})$, such that u is a viscosity solution to the following equation

$$\begin{cases} -|u'(z)|^2 = W(z, \rho_1^*, \rho_2^*), & \text{in } \mathbb{R}, \\ \max_{z \in \mathbb{R}} u(z) = 0. \end{cases}$$
(3.15)

Moreover, we have the following condition on the zero level set of u:

$$\operatorname{supp} n_1^* = \operatorname{supp} n_2^* \subset \{ z \mid u(z) = 0 \} \subset \{ z \mid W(z, \rho_1^*, \rho_2^*) = 0 \}.$$

(iii) Under condition (3.12) we have supp $n_1^* = \text{supp } n_2^* = \{z \mid W(z, \rho_1^*, \rho_2^*) = 0\}$ and hence

$$\{z \mid u(z) = 0\} = \{z \mid W(z, \rho_1^*, \rho_2^*) = 0\}.$$
(3.16)

The solution of (3.15)–(3.16) is indeed unique and hence the whole sequence $(u_{\varepsilon,i})_{\varepsilon}$ converge locally uniformly in \mathbb{R} to u.

Note that statement (iii) above is a consequence of the results in the adaptive dynamics framework and in particular of Corollary (3.2). Hamilton-Jacobi equations of type (3.15) in general might admit several viscosity solutions. Here, the uniqueness is obtained thanks to (3.16) and a property from the weak KAM theory, which is the fact that the viscosity solutions are completely determined by one value taken on each static class of the Aubry set ([Lio82a], Chapter 5 and [Con01]). Here, this set corresponds to the maximum points of W. In what follows we assume that (3.12) and hence (3.16) always hold. We then give an explicit formula for u considering two cases (one can indeed verify easily that the functions below are viscosity solutions to (3.15)– (3.16)):

(i) **Monomorphic ESS**: We consider the case where there exists a unique monomorphic ESS z^{M*} such that the corresponding demographic equilibrium is given by $(\rho_1^{M*}\delta(z^*), \rho_2^{M*}\delta(z^{M*}))$. Then u is given by

$$u(z) = - \big| \int_{z^{M*}}^{z} \sqrt{-W(x;\rho_1^{M*},\rho_2^{M*})} dx \big|.$$

(ii) **Dimorphic ESS**: We next consider the case where there exists a unique dimorphic ESS $(z_{I}^{D*}, z_{II}^{D*})$ with the demographic equilibrium: $n_i = \nu_{I,i}\delta(z-z_{I}^{D*}) + \nu_{II,i}\delta(z-z_{II}^{D*})$, and $\nu_{I,i} + \nu_{II,i} = \rho_i^{D*}$. Then u is given by

$$u(z) = \max\Big(-|\int_{z_1^*}^z \sqrt{-W(x;\rho_1^{D*},\rho_2^{D*})} dx|, -|\int_{z_{11}^*}^z \sqrt{-W(x;\rho_1^{D*},\rho_2^{D*})} dx|\Big).$$

3.3 A source and sink case

In this section, we consider a particular case where there is migration only from one habitat to the other, that is

$$m_1 > 0, \qquad m_2 = 0.$$
 (3.17)

In this case, one population (the source) does not receive any migrant from the second population (the sink). We also assume that

$$r_1 - m_1 > 0. (3.18)$$

Following similar arguments to the case of migration in both directions, one can characterize the mutation, selection and migration equilibria. However, since the migration is only in one direction, we should study the equilibria in the two habitats separately. Note that since $m_2 = 0$, there is no influence of the second habitat on the first habitat. One can indeed compute explicitly $n_{\varepsilon,1}$:

$$n_{\varepsilon,1}(z) = \frac{g_1^{\frac{1}{4}}\rho_{\varepsilon,1}}{\sqrt{2\pi\varepsilon}} \exp\Big(-\frac{\sqrt{g_1}(z+\theta)^2}{2\varepsilon}\Big), \qquad \rho_{\varepsilon,1} = \frac{r_1 - m_1 - \varepsilon\sqrt{g_1}}{\kappa_1}.$$

Note that as $\varepsilon \to 0$, $n_{\varepsilon,1}$ converges in the sense of measures to n_1^{M*} with

$$n_1^{M*}(z) = \rho_1^{M*} \delta(z+\theta), \qquad \rho_1^{M*} = \frac{r_1 - m_1}{\kappa_1}.$$

Here, $\{-\theta\}$ is indeed the unique ESS in the first habitat and n_1^* corresponds to the demographic equilibrium at the ESS.

In the second habitat however, there is an influence of the population coming from the first habitat. The natural quantity that appears in this case as the effective fitness in the second habitat is still the principal eigenvalue of (3.7) which is, in this case, given by

$$W(z,\rho_2) = \max(r_1 - g_1(z+\theta)^2 - \kappa_1 \rho_1^{M*} - m_1, r_2 - g_2(z-\theta)^2 - \kappa_2 \rho_2) = \max(-g_1(z+\theta)^2, r_2 - g_2(z-\theta)^2 - \kappa_2 \rho_2).$$

Then one can introduce the notion of the ESS for this habitat similarly to Section 3.2.1.

In Subsection 3.3.1 we present our results for the source and sink model in the adaptive dynamics framework. In Subsection 3.3.2 we provide our main result on the derivation of the zero order term u_2 .

3.3.1 The results in the adaptive dynamics framework

We can indeed always identify the unique ESS:

Theorem 3.6. Assume (3.17)–(3.18). In each habitat there exists a unique ESS. In habitat 1 the ESS is always monomorphic and it is given by $\{-\theta\}$ with the following demographic equilibrium:

$$n_1^{M*} = \rho_1^{M*} \,\delta(z+\theta), \quad \rho_1^{M*} = \frac{r_1 - m_1}{\kappa_1}.$$

In habitat 2 there are two possibilities:

(i) the ESS is dimorphic if and only if

$$\frac{m_1(r_1 - m_1)}{\kappa_1} < \frac{4g_2\theta^2 r_2}{\kappa_2}.$$
(3.19)

The dimorphic ESS is given by $\{-\theta, \theta\}$ with the following demographic equilibrium:

$$n_2^{D*} = \alpha \delta(z+\theta) + \beta \delta(z-\theta), \quad \rho_2^{D*} = \alpha + \beta = \frac{r_2}{\kappa_2}, \quad \alpha = \frac{m_1(r_1 - m_1)}{4g_2 \theta^2 \kappa_1}, \quad \beta = \frac{r_2}{\kappa_2} - \frac{m_1(r_1 - m_1)}{4g_2 \theta^2 \kappa_1}.$$

(ii) If condition (3.19) is not satisfied then the ESS in the second habitat is monomorphic. The ESS is given by $\{-\theta\}$ with the following demographic equilibrium:

$$n_2^{M*} = \rho_2^{M*} \,\delta(z+\theta), \quad \rho_2^{M*} = \frac{1}{2\kappa_2} \Big(r_2 - 4g_2\theta^2 + \sqrt{(r_2 - 4g_2\theta^2)^2 + 4\frac{\kappa_2}{\kappa_1}m_1(r_1 - m_1)} \Big).$$

3.3.2 The computation of the zero order term u_2

We next identify the zero order term u_2 (introduced in (3.5)–(3.6)):

Theorem 3.7. Assume (3.17)–(3.18).

(i) As $\varepsilon \to 0$, $(n_{\varepsilon,1}, n_{\varepsilon,2})$ converges to (n_1^{M*}, n_2^*) , the demographic equilibrium of the unique ESS of the metapopulation, given by Theorem 3.6. Moreover, as $\varepsilon \to 0$, $(\rho_{\varepsilon,1}, \rho_{\varepsilon,2})$ converges to (ρ_1^{M*}, ρ_2^*) , the total population sizes in habitats 1 and 2 corresponding to this demographic equilibrium.

(ii) As $\varepsilon \to 0$, $(u_{\varepsilon,1})_{\varepsilon}$ converges locally uniformly in \mathbb{R} to $u_1(z) = -\frac{\sqrt{g_1}}{2}(z+\theta)^2$. As $\varepsilon \to 0$, $(u_{\varepsilon,2})_{\varepsilon}$ converges along subsequences and locally uniformly in \mathbb{R} to a function $u_2 \in C(\mathbb{R})$ which satisfies

$$-|u_2'|^2 \le \max(R_1(z,\rho_1^{M*}) - m_1, R_2(z,\rho_2^*)), \quad -|u_2'|^2 \ge R_2(z,\rho_2^*), \quad u_1(z) \le u_2(z), \quad \max_{z \in \mathbb{R}} u_2(z) = 0$$

where the first two inequalities are in the viscosity sense. Moreover, we have the following condition on the zero level set of u_2 :

$$\operatorname{supp} n_2^* \subset \{ z \mid u_2(z) = 0 \} \subset \{ z \mid \max(R_1(z, \rho_1^{M*}) - m_1, R_2(z, \rho_2^*)) = 0 \}.$$

Theorem 3.7 allows us to identify u in a neighborhood of the ESS points:

Proposition 3.8. (i) There exists a connected and open set $\gamma_{I} \subset \mathbb{R}$, with $-\theta \in \gamma_{I}$, such that

$$u_2(z) = -\frac{\sqrt{g_1}}{2}(z+\theta)^2.$$

(ii) Assume that (3.19) holds. Then, there exists a connected and open set $\gamma_{II} \subset \mathbb{R}$, with $\theta \in \gamma_{II}$, such that

$$u_2(z) = -\frac{\sqrt{g_2}}{2}(z-\theta)^2$$

(iii) Assume that

$$\frac{4g_2\theta^2 r_2}{\kappa_2} < \frac{m_1(r_1 - m_1)}{\kappa_1}$$

Then $u_2(\theta) < 0$.

3.4 Heuristics and approximation of the moments of the phenotypic distribution

In order to apply our results to the study of some biological case studies, which will be done in the next section, we will provide two analytic approximations. Our *first approximation* is provided by the computation of the global values of the terms u and v_i and using (3.5) and (3.6). In order to provide more explicit formula for the moments of order $k \ge 1$ of the population's distribution in terms of the parameters of the model, we also provide a *second approximation*. This *second approximation*, instead of using the values of u and v_i in the whole domain, is based on the computation of Taylor expansions of u and v_i around the ESS points.

Our *second approximation* is by definition less accurate than the first one. It still provides convincing results when the parameters are such that we are far from the transition zone from monomorphic to dimorphic distribution (see for instance Figure 3.2). This approximation is

indeed based on an integral approximation (see Subsection 3.4.2) which is relevant only when the population's distribution is relatively sharp around the ESS points. This is not the case in the transition zone unless the effect of the mutations, i.e. ε , is very small.

In this section, we show how to derive these approximations in the case where $m_i > 0$ (see [MG] for the derivations in the source and sink scenario). First, we provide some heuristic computations which allow us to derive terms u and v_i and hence our *first approximation* (subsection 3.4.1). In particular, such computations allow us to derive the Hamilton-Jacobi equation (3.15) . Next, we show how to derive analytic approximations for the moments of the phenotypic distribution, that is our *second approximation* (subsection 3.4.2).

3.4.1 Derivation of our *first approximation*

In this subsection we can provide the elements to obtain an approximation of the population's total density $\rho_{\varepsilon,i}$ and the phenotypic density $n_{\varepsilon,i}(z)$ in the following form

$$\rho_{\varepsilon,i} \approx \rho_i + \varepsilon K_i, \qquad n_{\varepsilon,i}(z) \propto \frac{1}{\sqrt{2\pi\varepsilon}} \exp\left(\frac{u_i(z) + \varepsilon v_i(z)}{\varepsilon}\right).$$

Computation of the zero order term u_i :

We first provide some heuristic arguments leading to the Hamilton-Jacobi equation (3.15). Note that the equilibrium $(n_{\varepsilon,1}, n_{\varepsilon,2})$ solves

$$\begin{cases} 0 = \varepsilon^2 \frac{\partial^2 n_{\varepsilon,1}(z)}{\partial z^2} + n_{\varepsilon,1}(z) R_1(z;\rho_{\varepsilon,1}) + m_2 n_{\varepsilon,2}(z) - m_1 n_{\varepsilon,1}(z), \\ 0 = \varepsilon^2 \frac{\partial^2 n_{\varepsilon,2}(z)}{\partial z^2} + n_{\varepsilon,2}(z) R_2(z;\rho_{\varepsilon,2}) + m_1 n_{\varepsilon,1}(z) - m_2 n_{\varepsilon,2}(z). \end{cases}$$
(3.20)

We first let $\varepsilon \to 0$ in the above equation to obtain formally that $n_{\varepsilon,i}(z) \to n_i(z)$ and $\rho_{\varepsilon,i} \to \rho_i$ with

$$\begin{cases} 0 = n_1(z)R_1(z;\rho_1) + m_2n_2(z) - m_1n_1(z), \\ 0 = n_2(z)R_2(z;\rho_2) + m_1n_1(z) - m_2n_2(z), \\ \rho_i = \int_{\mathbb{R}} n_i(z)dz, \end{cases}$$

which is equivalent with

$$\mathcal{A}(z,\rho_1,\rho_2)\left(\begin{array}{c}n_1(z)\\n_2(z)\end{array}\right)=0,\qquad \rho_i=\int_{\mathbb{R}}n_i(z)dz,$$

with $\mathcal{A}(z, \rho_1, \rho_2)$ given by (3.7). This means that (ρ_1, ρ_2) corresponds to the sizes of the populations 1 and 2 at the demographic equilibrium $(n_1(z), n_2(z))$, in absence of mutations. We will show that this equilibrium corresponds indeed to an evolutionary stable strategy. To this end, we replace (3.5) in (3.20) and obtain

$$\begin{cases} 0 = \varepsilon \frac{\partial^2 u_{\varepsilon,1}(z)}{\partial z^2} + |\frac{\partial}{\partial z} u_{\varepsilon,1}(z)|^2 + R_1(z;\rho_1) + m_2 \exp(\frac{u_{\varepsilon,2}(z) - u_{\varepsilon,1}(z)}{\varepsilon}) - m_1, \\ 0 = \varepsilon \frac{\partial^2 u_{\varepsilon,2}(z)}{\partial z^2} + |\frac{\partial}{\partial z} u_{\varepsilon,2}(z)|^2 + R_2(z;\rho_2) + m_1 \exp(\frac{u_{\varepsilon,1}(z) - u_{\varepsilon,2}(z)}{\varepsilon}) - m_2. \end{cases}$$
(3.21)

We can determine $u_i(z)$, $v_i(z)$ from the above equation and (3.6).

Note that the exponential terms in (3.21) suggest that, when $m_i > 0$ for i = 1, 2, as $\varepsilon \to 0$ $u_{\varepsilon,1}(z)$ and $u_{\varepsilon,2}(z)$ converge to the same limit u(z). Otherwise, one of these exponential terms tends to infinity while the other terms are bounded. Keeping the zero order terms we obtain

$$\begin{cases} 0 = |\frac{\partial}{\partial z}u(z)|^2 + R_1(z;\rho_1) + m_2 \exp(v_2(z) - v_1(z)) - m_1, \\ 0 = |\frac{\partial}{\partial z}u(z)|^2 + R_2(z;\rho_2) + m_1 \exp(v_1(z) - v_2(z)) - m_2. \end{cases}$$
(3.22)

We then multiply the first line by $\exp(v_1(z))$ and the second line by $\exp(v_2(z))$ and write the system in the matrix form to obtain

$$\begin{pmatrix} R_1(z;\rho_1)-m_1 & m_2\\ m_1 & R_2(z;\rho_2)-m_2 \end{pmatrix} \begin{pmatrix} \exp(v_1(z))\\ \exp(v_2(z)) \end{pmatrix} = -|\frac{\partial}{\partial z}u(z)|^2 \begin{pmatrix} \exp(v_1(z))\\ \exp(v_2(z)) \end{pmatrix}.$$

Note that the matrix in the l.h.s. is nothing but $\mathcal{A}(z,\rho_1,\rho_2)$ given by (3.7). The equality above means that $-|\frac{\partial}{\partial z}u(z)|^2$ is indeed the principal eigenvalue of $\mathcal{A}(z;\rho_1,\rho_2)$, that is (see Section 3.2.1):

$$-|\frac{\partial}{\partial z}u(z)|^2 = W(z;\rho_1,\rho_2).$$

Moreover, since the total population size is bounded, we have

$$\max_{z \in \mathbb{R}} u(z) = 0.$$

Similarly we obtain

$$upp n_i \subset \{ z \,|\, u(z) = 0 \}. \tag{3.23}$$

Let \bar{z} be such that $u(\bar{z}) = 0$ which means that it is a maximum point of u(z). Then, $\frac{\partial}{\partial z}u(\bar{z}) = 0$ and hence

$$W(\bar{z};\rho_1,\rho_2)=0$$

Moreover in all the points $z \in \mathbb{R}$, we have

$$W(z;\rho_1,\rho_2) = -|\frac{\partial}{\partial z}u(z)|^2 \le 0$$

This implies that

if
$$z \in \operatorname{supp} n_1 = \operatorname{supp} n_2$$
 then $W(z; \rho_1, \rho_2) = 0$,
if $z \notin \operatorname{supp} n_1 = \operatorname{supp} n_2$ then $W(z; \rho_1, \rho_2) \le 0$.

In other words $(n_1(z), n_2(z))$ corresponds to the demographic equilibrium corresponding to the global ESS and hence $n_i(z) = n_i^*(z)$ and $\rho_i = \rho_i^*$, with $n_i^*(z)$ and ρ_i^* given in Subsection 3.2.2. We gather the informations that we obtained on u(z):

$$\begin{cases} -|\frac{\partial}{\partial z}u(z)|^2 = W(z;\rho_1^*,\rho_2^*),\\ \max_{z\in\mathbb{R}}u(z) = 0, \end{cases}$$

which is nothing but (3.15).

Such Hamilton-Jacobi equation may admit several viscosity solutions in general. However, any nonpositive viscosity solution of (3.15) can be determined by its values at the maximum points of W ([Lio82a], Chapter 5 and [Con01]). The maximum points of W are given by the set $\{z \in \mathbb{R} \mid W(z; \rho_1^*, \rho_2^*) = 0\}$. Under condition (3.12), we obtain that

if
$$z \notin \operatorname{supp} n_1 = \operatorname{supp} n_2$$
 then $W(z; \rho_1^*, \rho_2^*) < 0$,

which implies that

supp
$$n_1 = \text{supp } n_2 = \{ z \in \mathbb{R} \, | \, W(z; \rho_1^*, \rho_2^*) \}.$$

Combining this equality with (3.23) we deduce that u = 0 at the maximum points of W. This property leads to the uniqueness of u and allows us to derive the explicit formula for u, given in Section 3.2.3.

Computation of the next order terms $v_i(z)$: Next, we provide the main elements to compute formally the next order term $v_i(z)$.

From (3.22) and (3.15) we can compute $v_2(z) - v_1(z)$ thanks to the following formula

$$v_2(z) - v_1(z) = \log\left(\frac{1}{m_2}\left(W(z,\rho_1^*,\rho_2^*) - R_1(z,\rho_1^*) + m_1\right)\right).$$

We next keep the first order terms in (3.21). To do so, we need to go further in the approximation of $u_{\varepsilon,i}(z)$ in (3.6) and also keep the term of order ε^2 :

$$u_{\varepsilon,i}(z) = u(z) + \varepsilon v_i(z) + \varepsilon^2 l_i(z) + O(\varepsilon^3).$$

Then, keeping the first order terms in (3.21) we obtain

$$\begin{cases} 0 = \frac{\partial^2}{\partial z^2} u(z) + 2\frac{\partial}{\partial z} u(z) \frac{\partial}{\partial z} v_1(z) - \kappa_1 K_1 + m_2 \exp(v_2(z) - v_1(z))(l_2(z) - l_1(z)), \\ 0 = \frac{\partial^2}{\partial z^2} u(z) + 2\frac{\partial}{\partial z} u(z) \frac{\partial}{\partial z} v_2(z) - \kappa_2 K_2 + m_2 \exp(v_1(z) - v_2(z))(l_1(z) - l_2(z)). \end{cases}$$

Using the above equalities and by evaluating them at the ESS points we can compute $v_i(z)$ and K_i for i = 1, 2. See [Mir17]–Section 3.3 for the details of such computations.

3.4.2 Derivation of our second approximation

In this subsection, we prove the main elements to derive analytic approximations for the moments of the phenotypic distribution. We only present the method in the case of monomorphic population. See [MG] for the case of dimorphic population.

Let's suppose that we are in the case where the ESS is monomorphic and given by z^{M*} In order to provide an explicit approximation of the moments of the population's distribution, we compute the third order approximation of u(z) around z^{M*} :

$$u(z) = -\frac{A}{2}(z - z^{M*})^2 + B(z - z^{M*})^3 + O(z - z^{M*})^4.$$
(3.24)

and the first order approximation of $v_i(z)$ around z^{M*} :

$$v_i(z) = \log(\sqrt{A}\rho_i^{M*}) + D_i(z - z^{M*}) + O(z - z^{M*})^2.$$
(3.25)

To obtain the zero order term in the expansion for $v_i(z)$ we use the fact that, as the mutation's variance vanishes $(\varepsilon \to 0)$, the total population size $\rho_{\varepsilon,i}$ tends to ρ_i^{M*} which corresponds to the demographic equilibrium at the ESS.

The above expansions allow us to estimate the moments of the population's distribution:

$$\begin{cases} \mu_{\varepsilon,i} = \frac{1}{\rho_{\varepsilon,i}} \int z n_{\varepsilon,i}(z) dz = z^{M*} + \varepsilon (\frac{3B}{A^2} + \frac{D_i}{A}) + O(\varepsilon^2) \\ \sigma_{\varepsilon,i}^2 = \frac{1}{\rho_{\varepsilon,i}} \int (z - \mu_{\varepsilon,i})^2 n_{\varepsilon,i}(z) dz = \frac{\varepsilon}{A} + O(\varepsilon^2), \\ \psi_{\varepsilon,i} = \frac{1}{\rho_{\varepsilon,i}} \int (z - \mu_{\varepsilon,i})^3 n_{\varepsilon,i}(z) dz = \frac{6B}{A^3} \varepsilon^2 + O(\varepsilon^3). \end{cases}$$

We next show how to compute such approximations. We can indeed use the expressions in (3.24) and (3.25) to compute for any integer $k \ge 1$,

$$\int (z - z^{M*})^k n_{\varepsilon,i}(z) dz = \frac{\varepsilon^{\frac{\kappa}{2}} \sqrt{A\rho_i^{M*}}}{\sqrt{2\pi}} \int_{\mathbb{R}} \left(y^k e^{-\frac{A}{2}y^2} \left(1 + \sqrt{\varepsilon} (By^3 + D_i y) + O(\varepsilon) \right) dy \right)$$
$$= \varepsilon^{\frac{k}{2}} \rho_i^{M*} \left(\omega_k(\frac{1}{A}) + \sqrt{\varepsilon} (B\omega_{k+3}(\frac{1}{A}) + D_i \omega_{k+1}(\frac{1}{A})) \right) + O(\varepsilon^{\frac{k+2}{2}}),$$

where $\omega_k(\sigma^2)$ corresponds to the k-th order central moment of a Gaussian distribution with variance σ^2 . Note that to compute the integral terms above we have performed a change of variable $z - z^{M*} = \sqrt{\varepsilon}y$, therefore each term $z - z^{M*}$ can be considered as of order $\sqrt{\varepsilon}$ in the integrations. The above integrations are the main ingredients to obtain the approximations of the moments given above, i.e. our *second approximation*.

3.5 Some examples and biological applications

In this section, we consider two particular case studies: a first case with symmetric habitats (that is under assumption (3.13)), and a second case with source and sink scenario (that is with $m_2 = 0$). For both examples, we will compare the numerical value of the phenotypic density $n_{\varepsilon,i}$ and its moments (obtained by the numerical approximation of the solution to (3.1)) with our two analytic approximations and discuss the biological applications of our results.

See [MG] for more detailed results and discussion on the biological application of this method.

3.5.1 Symmetric fitness landscapes

We focus first on a symmetric scenario where, apart from the position of the optimum, the two habitats are identical: $m_1 = m_2 = m$, $\kappa_1 = \kappa_2 = \kappa$, $g_1 = g_2 = g$, $r_1 = r_2 = r$. In this special case it is possible to fully characterize the evolutionary equilibrium.

When migration rate is higher than critical migration threshold $m > m_c = 2g\theta^2$ migration prevents the differentiation of the trait between the two habitats (see Corollary 3.3). The only evolutionary equilibrium, when the mutation rate is vanishingly small, is monomorphic and satisfies $z^{M*} = 0$ and $n_1^{M*}(z) = n_2^{M*}(z) = \rho^{M*}\delta(z)$, where $\delta(.)$ is the Dirac delta function and $\rho^{M*} = \frac{1}{\kappa} (r - g\theta^2)$.

Monomorphic case: Let's suppose that $m_c = 2g\theta^2 \leq m$. Then $z^{M*} = 0$ is the only ESS and $\rho^{M*} = \frac{1}{\kappa} (r - g\theta^2)$. Then, we can provide our *first approximation* of the phenotypic density $n_{\varepsilon,i}(z)$ following the method introduced above (Figure 3.1). Moreover, defining $\phi = \sqrt{1 - 2g\theta^2/m}$, we can use the *second approximation* to obtain analytic formula for the

moments of the stationary state

$$\begin{cases} \rho_{\varepsilon,1} = \rho_{\varepsilon,2} = \int n_{\varepsilon,i}(z)dz = \frac{1}{\kappa} \left(r - g\theta^2\right) - \varepsilon \frac{\sqrt{g}\phi}{\kappa} + O(\varepsilon^2), \\ \mu_{\varepsilon,1} = \frac{1}{\rho_{\varepsilon,1}} \int z n_{\varepsilon,1}(z)dz = -\varepsilon \frac{\sqrt{g}\theta}{m\phi} + O(\varepsilon^2), \\ \mu_{\varepsilon,2} = \frac{1}{\rho_{\varepsilon,2}} \int z n_{\varepsilon,2}(z)dz = \varepsilon \frac{\sqrt{g}\theta}{m\phi} + O(\varepsilon^2), \\ \sigma_{\varepsilon,1}^2 = \sigma_{\varepsilon,2}^2 = \frac{1}{\rho_{\varepsilon,i}} \int (z - \mu_{\varepsilon,i})^2 n_{\varepsilon,i}(z)dz = \frac{\varepsilon}{\sqrt{g}\phi} + O(\varepsilon^2), \\ \psi_{\varepsilon,i} = \frac{1}{\rho_{\varepsilon,i}} \int (z - \mu_{\varepsilon,i})^3 n_{\varepsilon,i}(z)dz = O(\varepsilon^3). \end{cases}$$

In what follows 'differentiation' refers to the following quantity:

$$D_{\varepsilon} = \mu_{\varepsilon,2} - \mu_{\varepsilon,1}.$$

Note that the variance $\sigma_{\varepsilon,i}^2$ is larger than the case with no heterogeneity between the habitats, where we recover the well-known equilibrium variance $\frac{\varepsilon}{\sqrt{g}}$ of Quantitative Genetics [Lan75, BÖ0, Ric04]. This increase in the equilibrium variance comes from ϕ which depends on dispersion and the heterogeneity between the two habitats. The variance of the distribution increases as ϕ decreases. When $\phi = 0$ the approximation for the variance becomes infinitely large. Indeed, this corresponds to the threshold value of migration below which the above approximation collapses because the distribution becomes bimodal. In this case we have to switch to the analysis of the dimorphic case. Note that the differentiation between habitats depends also on ϕ . Some differentiation start to emerge even when the migration rate is above the critical migration rate, m_c . In Figure 3.2 we provide a comparison of the results from the *first* and the *second approximations*. Our *second approximation* provides convincing results when the parameters are such that we are far from the transition zone from monomorphic to dimorphic distribution as expected.

Dimorphic case: When $m < m_c$, the only stable evolutionary equilibrium is dimorphic which yields the following ESS: $\{z_{\rm I}^{D*}, z_{\rm II}^{D*}\}$ with $z_{\rm I}^{D*} = -z_{\rm II}^{D*} = -z^{D*}$ and $z^{D*} = \frac{\sqrt{4g^2\theta^4 - m^2}}{2g\theta}$. When $\varepsilon = 0$ this yields the following phenotypic densities at equilibrium : $n_i^{D*}(z) = \nu_{\rm I,i}\delta(z - z_{\rm I}^{D*}) + \nu_{\rm II,i}\delta(z - z_{\rm II}^{D*})$ (analytic expressions for $\nu_{\rm I,j}$ and $\nu_{\rm II,j}$ are given in [MG]). When $\varepsilon > 0$ we can use the *second approximation* to obtain analytic expressions for the local moments of the stationary distribution in each habitat (see [MG]).

3.5.2 The source-sink scenario

We next focus on the source-sink scenario, assuming that $m_1 > 0$ and $m_2 = 0$. In this case, we are not able to compute the first order corrector v_2 in the whole domain. However, a local approximation of v_2 gives already convincing results (Figure 3.3).

The evolutionary outcome in the first habitat is obvious because it depends only on selection acting in habitat 1: the ESS is $-\theta$ and

$$\rho_1^* = \frac{r_1 - m_1}{\kappa_1}.\tag{3.26}$$

Moreover, the population's phenotypic density $n_{\varepsilon,1}$ can be computed explicitly: $n_{\varepsilon,1} = \rho_{\varepsilon,1} f_{\varepsilon}$, where $\rho_{\varepsilon,1} = \frac{r_1 - m_1 - \varepsilon \sqrt{g_1}}{\kappa_1}$ and f_{ε} is the probability density of a normal distribution $\mathcal{N}(-\theta, \frac{\varepsilon}{\sqrt{g_1}})$.



Figure 3.1: Selection-mutation-migration equilibrium of the phenotypic densities $n_{\varepsilon,i}(z)$ in the two habitats in a symmetric scenario. We plot the exact phenotypic densities at equilibrium obtained from numerical computations (blue dots) together with our *first approximation* (full black line) in a case where the distribution is unimodal in each habitat. We also plot the approximation given in [DRG13] (red dashed line). Note that our approximation captures the emergence of some differentiation even though we are above the critical migration rate leading to the evolution of a dimorphic population. In presence of mutations, the population's distribution is indeed shifted to the left(respectively right) in the first(respectively second) habitat, while [DRG13] provided the same approximation for both habitats. Our calculation yields also better approximations for the variance of the distribution in each habitat ([DRG13] underestimates this variance). In this figure and in the following ones, to compute numerically the equilibrium, we have solved numerically a time-dependent version of equation (3.1) and kept the solution obtained after long time when the equilibrium has been reached. Parameter values: m = 1.5, r = 3, g = 2; $\theta = 0.5$, $\kappa = 1$, U = 1, $\varepsilon = 0.1$.

In habitat 2, the evolutionary outcome results from the balance between migration from habitat 1 and local selection. Interestingly, migration has a non-monotonous effect on adaptation in the sink (See Figure 3.3). Indeed, Figure 3.3A shows that the population size in the sink is maximized for intermediate values of migration. More migration from the source has a beneficial effect on the demography of the sink but it prevents local adaptation. Yet, when migration from the source becomes very large it limits the size of the population in the source (see (3.26)). This limits the influence of the source on the sink and can even promote adaptation to the sink. In fact it is worth noting that differentiation between the two habitats can actually increase with migration (Figure 3.3B). The level of migration from the source that prevents local adaptation in the sink is given by the condition:

$$\frac{4g_2\theta^2 r_2}{\kappa_2} \le \frac{m_1(r_1 - m_1)}{\kappa_1},\tag{3.27}$$

Indeed, when condition (3.27) is verified, the migration from the source overwhelms local selection and the evolutionary stable strategy in the sink is $z^* = -\theta$. In contrast, when condition (3.27) does not hold, the effective growth rate of the optimal trait θ in the sink habitat is high enough



Figure 3.2: Effects of migration in a symmetric scenario on (A) the total population size $(\rho_{\varepsilon,1})$ in habitat 1, (B) the differentiation between habitats $(\mu_{\varepsilon,2} - \mu_{\varepsilon,1})$, (C) the variance $(\sigma_{\varepsilon,1}^2)$ and (D) the third central moment of the phenotypic distribution $(\psi_{\varepsilon,1})$ in habitat 1. The dots refer to the numerical resolutions of the problem with $\varepsilon = 0.05$, the red line indicates the case where $\varepsilon = 0$ while the lines in black refer to our two approximations when $\varepsilon = 0.05$ (the dashed line for the *first approximation* and the full line for the *second approximation*). The vertical gray line indicates the critical migration rate below which dimorphism can evolve in the Adaptive Dynamics scenario. Other parameter values: r = 1, g = 2, $\theta = 0.5$, $\kappa = 1$.

to compete with the trait $-\theta$ coming from the source, allowing coexistence between the two strategies. Note again that our two approximations (see [MG] for more details) provide very good predictions for the moments of the phenotypic distribution in the sink (Figure 3.3).



Figure 3.3: Effects of migration in a source-sink scenario on (A) the total population size in the sink habitat, (B) the differentiation between habitats, (C) the variance and (D) the third central moment of the phenotypic distribution in sink. The dots refer to exact numerical computations when $\varepsilon = 0.05$, the red line indicates the case where $\varepsilon = 0$ while the lines in black refer to our two approximations when $\varepsilon = 0.05$ (dashed line for the *first approximation* and the full line for the second approximation). The vertical gray lines, at m = 1 and m = 2, indicate the critical migration rates where transition occurs between monomorphism and dimorphism in the Adaptive Dynamics framework (see condition (3.27)). Other parameter values: $r_1 = 3$, $r_2 = 1$, $g_1 = 3$, $g_2 = 2$, $\kappa_1 = \kappa_2 = 1$, $\theta = 0.5$.

Chapter 4

Evolutionary adaptation in time-heterogeneous environments

4.1 Introduction

In this chapter, we study the evolutionary dynamics of populations in time varying environments. Natural populations experience environmental change due for instance to seasonal changes, climate change and interaction with other species. Cancer tumors and bacterial infections, when treated, are subject to a time heterogeneous selection thanks to the time dependent administration of the medications. Several questions arise with regard to time heterogeneous selection. How the time variations of an environment modify the evolutionary dynamics of a population? Can a population survive under the environmental change? How the size of the population would be affected? What would be the impact of time heterogeneous selection on the phenotypic distribution of the population? The better understanding of such phenomena is a major topic in the evolutionary biology theory and may help the understanding of several important questions [LMR10, CLM10, PMLJ⁺13, KM14, BPnMGI17]. What is the role of adaptation in the preservation of the biodiversity in the context of climate change? Is it possible to find optimal drug delivery schedules or optimal ways of combining drugs to avoid resistance in the treatment of cancer and bacterial infections?

An example of biological experiment addressing the above questions is given in [KMZ⁺13]. In this experiment, initially genetically identical populations of bacterial pathogen Serratia marcescens were kept in two different environments with constant or fluctuating temperatures for several weeks. Then, these populations were sampled and their growth rates were measured in different environments. The authors compared the performance of these two populations in different environments and observed in particular that the population evolved in periodically fluctuating temperature (daily variation between 24°C and 38°C, mean 31°C) outperformed the strains that evolved in constant temperature (31°C), when both strains were allowed to compete in a constant environment with temperature 31° C (see Figure 4.1). This is surprising at the first glance, since one could expect that the population evolved in a constant environment would select for the best traits in such environment, while the population evolved in a changing environment would have better performance when it is exposed to different types of environments but not necessarily at temperature 31° C which is the original environment of the first population. In Section 4.3 we will investigate, based on a mathematical model, in which situations such phenomenon may arise



and provide some possible explanation for the outcome of this experiment.

Figure 4.1: The comparison of the performances of the populations of bacterial pathogen Serratia marcescens evolved in environments with constant or fluctuating temperature, when they are placed at different environments. Figure from [KMZ⁺13].

A typical selection-mutation model in a time heterogeneous environment is written

$$\begin{cases} \frac{\partial}{\partial t}n(t,z) - \sigma\Delta n(t,z) = n(t,z)R(e(t),z,I(t)), & (t,z) \in \mathbb{R}^+ \times \mathbb{R}^d, \\ I(t) = \int_{\mathbb{R}^d} n(t,y)\psi(y)dy, & (4.1)\\ n(0,z) = n_0(z). \end{cases}$$

Here, as in the above chapters, z represents a phenotypic trait and n(t, z) corresponds to the density of individuals with phenotype z at time t. The diffusion term corresponds to the mutations. We denote by $\psi(z)$ the resource consumption rate of individuals with trait z and by I(t) the total resource consumption of the population. The main difference with the previous models is that the growth rate R depends, in addition to the trait z and the environmental feedback I, on the environmental state e(t). A main difficulty in the study of such problems comes from the combination of the nonlocal competition term with the time dependency of the growth rate.

The outcome of such models is very rich. We will observe in the next sections that, depending on how the environment variations act on the growth rate and depending on the speed of the environment change, different types of behaviors may be observed. In Section 4.2 we will study the dynamics of the phenotypic density under periodically oscillating growth rate, in a concave framework closely related to [LMP11], when the effect of the mutations is much smaller than the frequency of the oscillations [MPS15]. In Section 4.3 we study a model where the fitness landscape varies with a linear trend but in an oscillatory manner, that is we consider a growth rate of the form R(e(t), z - ct) - I(t) [FIM18, FIM]. In this section similarly to Section 4.2, we assume that the effect of the mutations are much smaller than the frequency of the oscillations. However, on the contrary to the model in Section 4.2, we do not make any concavity assumption on R but we consider a linear dependence of the growth rate on the nonlocal competition term I. This model is motivated by the study of the adaptation of a population to a climate change. Note that many natural populations are subject both to directional change of phenotypic optimum and random fluctuations of the environment [CVT15]. Such analysis provides also new results in the case of oscillating environments, without linear change (c = 0) [FIM18]. While the work in Section 4.2 allows to study the dynamics of a population under fluctuating selection, the work in [FIM18], corresponding to the case c = 0 in Section 4.3, focuses on the long time outcome of such fluctuating selection. In particular, this analysis allows us to provide some possible explanation for the outcome of the experience in [KMZ⁺13] presented above. Finally in Section 4.4, we study a case where the environmental change occurs very slowly [CEM]. More precisely, we consider a piecewise constant environmental state with slow switch.

The adaptation of populations in time varying environments is a major topic in evolutionary biology (see for instance [BL95, LS96, CLM10, KM14]). In [LCDH15, LCC16, ABF⁺19] models closely related to (4.1) are studied. These works focus mostly, in the context of cancer therapy optimization, on the numerical study of the problem or on the study of particular forms of growth rate R where semi-explicit solutions or explicit solutions may be obtained. In the case of a directional environment change, that is by replacing z by z - ct in the growth rate R, closely related equations, but with a local reaction term and no fluctuation, have been widely studied (see for instance [BDNZ08, BR08, BR09, BDD14, BF18]). Such models are introduced to study dynamics of populations structured by a space variable neglecting evolution. It is shown in particular that there exists a critical speed of environment change c^* , such that the population survives if and only if the environment change occurs with a speed less than c^* . We also refer to [BL16] where an integro-difference model has been studied for the spatial dynamics of a population in the case of a randomly changing environment. Moreover, in [ABR17] both spatial and evolutionary dynamics of a population in an environment with linearly moving optimum has been studied. While here we don't include any spatial structure, in Section 4.3 we take into account an oscillatory change of the environment in addition to a change with linear trend.

The work in Section Section 4.2 is based on the result in [MPS15] in collaboration with B. Perthame and P. E. Souganidis. The work in Section 4.3 is based on the articles [FIM18, FIM] which are prepared in the framework of the PhD project of my student S. Figueroa Iglesias. Finally, Section 4.3 is based on the results in [CEM] which are obtained in the framework of the postdoctoral project of C. Etchegaray, who I mentored with M. Costa.

4.2 The evolutionary dynamics of a population in an oscillating environment

In this section, we extend the Hamilton-Jacobi approach to the case of oscillatory environments using homogenization tools. We study (4.1) considering an oscillating environment, with $e(\cdot)$ a 1-periodic function, and assuming small effect of the mutations $\sigma = \varepsilon^2$, with ε a small parameter. Being interested in the dynamics of the solution, we perform a change of variable in time $t \to \frac{t}{\varepsilon}$ and define

$$n_{\varepsilon}(t,z) = n(\frac{t}{\varepsilon},z), \qquad I_{\varepsilon}(t) = I(\frac{t}{\varepsilon}),$$

to accelerate the dynamics and observe the effect of small mutations, similarly to Chapter 1. Then, (4.1) becomes

$$\begin{cases} \varepsilon \frac{\partial}{\partial t} n_{\varepsilon}(t,z) - \varepsilon^2 \Delta n_{\varepsilon}(t,z) = n_{\varepsilon}(t,z) R(e(\frac{t}{\varepsilon}), z, I_{\varepsilon}(t)), & (t,z) \in \mathbb{R}^+ \times \mathbb{R}^d, \\ I_{\varepsilon}(t) = \int_{\mathbb{R}^d} n_{\varepsilon}(t,y) \psi(y) dy, & (4.2) \\ n_{\varepsilon}(0,z) = n_{\varepsilon,0}(z). \end{cases}$$

Our objective is to describe the limit of the solution n_{ε} as the effect of the mutations, measured by ε , vanishes. The major difficulty in the asymptotic analysis of this problem comes from the fast oscillatory function $e(\frac{t}{\varepsilon})$ combined with the nonlocal term $I_{\varepsilon}(t)$.

To provide an asymptotic analysis of this model we consider a concave framework closely related to [LMP11]. We prove that, as $\varepsilon \to 0$, the phenotypic density concentrates on an evolving point $\overline{z}(t)$ corresponding to a dominant trait. The dynamics of this dominant trait is given by a canonical equation, indicating that $\overline{z}(t)$ follows the gradient of an effective growth rate \mathcal{R} , obtained after a homogenization process from the growth rate R(e, z, I). The size of the population $\rho_{\varepsilon}(t) = \int n_{\varepsilon}(t, z)dz$ and the competition term $I_{\varepsilon}(t)$ may still have oscillations at the limit $\varepsilon \to 0$ and they converge only in a weak sense. For some particular forms of the growth rate R we also provide the long time behavior of the dominant trait $\overline{z}(t)$. Finally, we provide an example where our method allows to show that periodic fluctuations may increase the population size.

In subsection 4.2.1 we present our assumptions and in subsection 4.2.2 we provide our main results. In subsection 4.2.3 we provide an example where fluctuations may increase the population size.

4.2.1 Assumptions

We assume that

$$R: \mathbb{R} \times \mathbb{R}^d \times [0, \infty) \to \mathbb{R} \quad \text{is smooth and} \qquad e: \mathbb{R}^+ \to \mathbb{R} \text{ is 1-periodic.} \tag{4.3}$$

To ensure the survival and the boundedness of the population we assume that R takes positive values for "small enough populations" and negative values for "large enough populations", i.e., there exist a value $I_M > 0$ such that

$$\max_{0 \le s \le 1, \ z \in \mathbb{R}^d} R(e(s), z, I_M) = 0 \quad \text{and} \quad \mathcal{Z} := \{ z \in \mathbb{R}^d, \ \int_0^1 R(e(s), z, 0) ds > 0 \} \neq \emptyset.$$
(4.4)

In addition the growth rate R satisfies, for some positive constants K_i , i = 1, ..., 7, and all $(e, z, I) \in \mathbb{R} \times \mathbb{R}^d \times [0, I_M]$ and A > 0, the following decay and concavity assumptions:

$$-2K_1 \le D_z^2 R(e, z, I) \le -2K_2, \qquad -K_3 - K_1 |z|^2 \le R(e, z, I) \le K_4 - K_2 |z|^2, \tag{4.5}$$

$$-K_5 \le D_I R(e, z, I) \le -K_6,$$
(4.6)

$$D_z^3 R \in L^{\infty} \left(\mathbb{R} \times \mathbb{R}^d \times [0, A] \right) \quad \text{and} \quad |D_{z,I}^2 R| \le K_7.$$

$$(4.7)$$

The "uptake coefficient" $\psi : \mathbb{R}^d \to \mathbb{R}$ must be regular and bounded from above and below, i.e., there exist positive constants ψ_m , ψ_M and K_8 such that

$$0 < \psi_m \le \psi \le \psi_M$$
 and $\|\psi\|_{C^2} \le K_8.$ (4.8)

We also assume that the initial datum is "asymptotically monomorphic", i.e., it is close to a Dirac mass in the sense that there exist $z_0 \in \mathcal{Z}$, $\rho_0 > 0$ and a smooth $u_{\varepsilon,0} : \mathbb{R}^d \to \mathbb{R}$ such that

$$n_{\varepsilon,0} = e^{u_{\varepsilon,0}/\varepsilon}$$
 and, as $\varepsilon \to 0$, (4.9)

$$n_{\varepsilon}(0,\cdot) \xrightarrow[\varepsilon \to 0]{} \varrho_0 \delta(\cdot - z_0)$$
 weakly in the sense of measures. (4.10)

In addition there exist constants $L_i > 0, i = 1, ..., 4$, and a smooth $u_0 : \mathbb{R}^d \to \mathbb{R}$ such that, for all $z \in \mathbb{R}^d$,

$$-2L_1I \le D_z^2 u_{\varepsilon,0} \le -2L_2I, \quad -L_3 - L_1|z|^2 \le u_{\varepsilon,0}(z) \le L_4 - L_2|z|^2, \quad \max_{z \in \mathbb{R}^d} u_0(z) = 0 = u_0(z_0),$$
(4.11)

and, as $\varepsilon \to 0$,

$$u_{\varepsilon,0} \longrightarrow u_0$$
 locally uniformly in \mathbb{R}^d . (4.12)

4.2.2 Main results

To state our results we first need, as usual in homogenization, a cell problem given by the following Lemma:

Lemma 4.1. Assume (4.4), (4.5) and (4.8). For all $z \in \mathbb{Z}$, there exists a unique 1-periodic positive solution $\mathcal{I}(z,s) : [0,1] \to (0, I_M)$ to

$$\begin{cases} \frac{d}{ds}\mathcal{I}(z,s) = \mathcal{I}(z,s) \ R(e(s), z, \mathcal{I}(z,s)), \\ \mathcal{I}(z,0) = \mathcal{I}(z,1). \end{cases}$$
(4.13)

Moreover, as $\mathcal{Z} \ni z \to z_0 \in \partial \mathcal{Z}$,

$$\max_{0 \le s \le 1} \mathcal{I}(z, s) \to 0. \tag{4.14}$$

In view of (4.14), for $z \in \partial \mathcal{Z}$, we define, by continuity, $\mathcal{I}(z, s) = 0$. The function $\mathcal{I}(z, s)$ helps us to identify the weak limit of $I_{\varepsilon}(t)$. Moreover it helps us to derive, using a homogenization process, an effective growth rate \mathcal{R} which will replace R (see Theorem 4.2). This effective growth rate is defined, for all $y \in \mathbb{R}^d$ and $z \in \overline{\mathcal{Z}}$ (here $\overline{\mathcal{Z}}$ stands for the closure of \mathcal{Z}), as below

$$\mathcal{R}(y,z) := \int_0^1 R(e(s),y,\mathcal{I}(z,s)) ds$$

Notice also that, dividing (4.13) by $\mathcal{I}(z,s)$ and integrating in s and using the periodicity, we always have, for $z \in \overline{\mathcal{Z}}$,

$$\mathcal{R}(z,z)\equiv 0.$$

Example of effective growth rate: if the growth rate R has the following form

$$R(e, z, I) = a(e, z) - \kappa I,$$

then the effective growth rate \mathcal{R} is given by

$$\mathcal{R}(y,z) = \overline{a}(y) - \kappa \overline{\mathcal{I}}(z),$$

with

$$\overline{a}(y) = \int_0^1 a(e(s), y) ds, \qquad \overline{\mathcal{I}}(z) = \int_0^1 \mathcal{I}(z, s) ds.$$

Our first result is about the behavior of (n_{ε}) as $\varepsilon \to 0$. It asserts the existence of a fittest trait $\overline{z}(t)$ and a total population size $\overline{\rho}(t)$ at time t and provides a "canonical equation" for the evolution in time of \overline{z} in terms of the "effective fitness" $\mathcal{R}(y, z)$. In the sequel, $D_1\mathcal{R}$ denotes the derivative of \mathcal{R} with respect to the first argument.

Theorem 4.2 (Limit as $\varepsilon \to 0$ [MPS15]). Assume (4.3)–(4.12). There exist a fittest trait $\overline{z} \in C^1([0,\infty); \mathbb{Z})$ and a total population size $\overline{\rho} \in C^1([0,\infty); (0,\infty))$ such that, along subsequences as $\varepsilon \to 0$, $p_{\alpha}(t, z) \to \overline{z}(t)\delta(z, \overline{z}(t))$ such that along subsequences of measures

$$\begin{split} & \mathcal{H}_{\varepsilon}(\iota, \cdot) \longrightarrow \underline{\varrho}(\iota) o(\cdot - z(\iota)) \quad \text{weaking in the sense of measures,} \\ & I_{\varepsilon}(\cdot) \longrightarrow \overline{I}(\cdot) = \int_{0}^{1} \mathcal{I}(\overline{z}(\cdot), s) ds = \overline{\varrho}(\cdot) \psi(\overline{z}) \quad \text{ in } L^{\infty}(0, \infty) \text{ weak-}\star, \end{split}$$

and

$$R(e(\frac{t}{\varepsilon}), z, I_{\varepsilon}(t)) \longrightarrow \mathcal{R}(z, \overline{z}(t)) \quad weakly in the sense of measures in t and strongly in z.$$

Moreover, \overline{z} satisfies the canonical equation

$$\dot{\overline{z}}(t) = \left(-D_z^2 u(t, \overline{z}(t))\right)^{-1} \cdot D_1 \mathcal{R}(\overline{z}(t), \overline{z}(t)).$$
(4.15)

We note that, even if we are not in the framework of adaptive dynamics with rare mutations, we can still interpret the above terms in the language of adaptive dynamics: $\mathcal{R}(y, z)$ can be interpreted as the effective fitness of a mutant y in a resident population with a dominant trait z, while $D_1\mathcal{R}$ is usually called the selection gradient, since it represents the capability of invasion. The extra term $\left(-D_z^2 u(t, \overline{z}(t))\right)^{-1}$, as in Section 1.5, approximates the covariance matrix of the phenotypic distribution and is an indicator of the diversity around the dominant trait in the resident population.

The second issue is the identification of the long time limit of the fittest trait \overline{z} . We prove that, in the limit $t \to \infty$, the population converges to a, so called, Evolutionary Stable Strategy (ESS) corresponding to a distribution of population which is stable under introduction of small mutations (see [MSP73, Esh83, JR11] and Section 3.2.1 for a more detailed definition). See also for instance [DJMR08, Rao11] for studies of the local and global stability of stationary solutions of integro-differential population models in constant environments.

Theorem 4.3 (Limit as $t \to \infty$ [MPS15]). In addition to (4.3)–(4.12) assume that either d = 1 or, if d > 1, R is given, for some smooth $a, b, A, B : \mathbb{R}^d \to (0, \infty)$ by

$$R(e, z, I) = a(z)A(e, I) - b(z)B(e, I).$$
(4.16)

Then, as $t \to \infty$, the population reaches an Evolutionary Stable Distribution $\overline{\varrho}_{\infty}\delta(\cdot - \overline{z}_{\infty})$, i.e., $\overline{\varrho}(t) \longrightarrow \overline{\varrho}_{\infty}$ and $\overline{z}(t) \longrightarrow \overline{z}_{\infty}$, where $\overline{\varrho}_{\infty} > 0$ and \overline{z}_{∞} are characterized by (\mathcal{I} is defined in (4.13))

$$\mathcal{R}(\overline{z}_{\infty}, \overline{z}_{\infty}) = 0 = \max_{z \in \mathbb{R}^d} \mathcal{R}(z, \overline{z}_{\infty}) \quad and \quad \overline{\varrho}_{\infty} = \frac{1}{\psi(\overline{z}_{\infty})} \int_0^1 \mathcal{I}(\overline{z}_{\infty}, s) ds.$$
(4.17)

The proof of the above result for a growth rate R given by (4.16), is based on a Lyapunov functional defined by

$$L(t) := \frac{a(\overline{z}(t))}{b(\overline{z}(t))}.$$

Notice that we do not claim the uniqueness of the Evolutionary Stable Strategy. Indeed there may exist several $(\rho_{\infty}, \overline{z}_{\infty})$ satisfying (4.17). Here we only prove that there exists $(\rho_{\infty}, \overline{z}_{\infty})$

satisfying (4.17) such that, as $t \to \infty$, the phenotypic density converges to $\rho_{\infty} \delta(\cdot - \overline{z}_{\infty})$.

The difference between our conclusions and the results for time homogeneous environments in [LMP11] is that, in the canonical equation (4.15), the growth rate R is replaced by an effective growth rate \mathcal{R} which is derived after a homogenization process. Moreover, we are only able to prove that (I_{ε}) converges in L^{∞} weak-* and not a.e., which is the case for constant environments in [BMP09, LMP11] (see also Section 1.3). One can wonder whether the convergence of (I_{ε}) towards \overline{I} is indeed in the strong sense as in the case with constant environment. The numerical resolution of the problem suggests that I_{ε} oscillates with period of order ε , with non-vanishing amplitude and hence such convergence is indeed only in the weak sense (see Figure 4.2). This adds a difficulty in Theorem 4.3 and it is the reason why we are not able to describe, without additional assumptions, the long-time limit behavior of the fittest trait \overline{z} for general growth rate R when d > 1. This remains an open question.



Figure 4.2: Dynamics of the total resource consumption $I_{\varepsilon}(t)$ for $R(e, z, I) = (2 + e)\frac{2-z^2}{I+.5} - .5$, $e(s) = \sin(2\pi s), \psi(z) = 1$ and $\varepsilon = 0.01$. The function I_{ε} oscillates with period of order ε around a monotone curve \overline{I} .

In [MPS15], we give an example of \mathcal{R} not satisfying the assumptions of Theorem 4.3 for which $\overline{z}(t)$ exhibits a periodic behavior. This example fits the structure (4.18) below with general concavity properties on \mathcal{R} but it is not necessarily derived from a homogenization limit.

The proofs use in a fundamental way the classical Hopf-Cole transformation

$$u_{\varepsilon} = \varepsilon \ln n_{\varepsilon},$$

which yields the following Hamilton-Jacobi equation for u_{ε} :

$$\begin{cases} \partial_t u_{\varepsilon} = R\left(e(\frac{t}{\varepsilon}), z, I_{\varepsilon}(t)\right) + |\nabla u_{\varepsilon}|^2 + \varepsilon \Delta u_{\varepsilon}, & \text{in } (0, \infty) \times \mathbb{R}^d\\ u_{\varepsilon}(0, \cdot) = u_{\varepsilon, 0}(\cdot), & \text{in } \mathbb{R}^d. \end{cases}$$

The next theorem describes the behavior of (u_{ε}) , as $\varepsilon \to 0$.

Theorem 4.4. Assume (4.3)–(4.12). Along subsequences as $\varepsilon \to 0$, $u_{\varepsilon} \to u$ locally uniformly in $[0,\infty) \times \mathbb{R}^d$, where $u \in C([0,\infty) \times \mathbb{R}^d)$ is a viscosity solution of

$$\begin{cases} \partial_t u = \mathcal{R}(z, \overline{z}(t)) + |\nabla u|^2 & \text{in } \mathbb{R}^d \times (0, \infty), \\ \max_{z \in \mathbb{R}^d} u(t, z) = 0 = u(t, \overline{z}(t)) & \text{in } (0, \infty), \\ u(0, \cdot) = u_0 & \text{in } \mathbb{R}^d. \end{cases}$$
(4.18)

In the case of time homogeneous environments, one can derive such Hamilton-Jacobi equation without any concavity assumption on R and u_0 (see [PB08, BMP09] and Section 1.3). In order to extend our result beyond the concave framework, the major difficulty is to determine the effective growth rate \mathcal{R} . We need indeed assumptions (4.5) and (4.11) to ensure a priori that n_{ε} goes to a single Dirac mass at the point $\overline{z}(t)$. This information is needed in order to write (4.18). We point out that it is also known for such models that, while $z \in \mathbb{R}$ and R is monotonic with respect to z and making no concavity assumption, n_{ε} still converges to a single Dirac mass. This suggests that we could relax the concavity assumptions in this latter case or in similar situations where such information is known a priori.

It is also possible to obtain such type of results, in the multi-d case and beyond the concave framework, for particular forms of R without using the concentration of n_{ε} as a Dirac mass. In [MPS15] we give indeed an example of a non-concave R, that is

$$R(e, z, I) = a(z)A(e, I) - B(e, I),$$

for which the effective growth rate \mathcal{R} can be computed independently of $\overline{z}(t)$ and equivalent results can be obtained. This example is all the more interesting since the corresponding effective growth rate \mathcal{R} has a natural form that can be computed explicitly, while in general not much is known about the structure of \mathcal{R} .

4.2.3 A qualitative effect: fluctuations may increase the population size

We conclude with an example that shows that the time-oscillations may lead to a strict increase of the population size at the evolutionary stable state, a conclusion which also holds in the context of physiologically structured populations [CGP07]. Note that the positive or negative impact of time-oscillations on the population size depends on the choice of the growth rate R. In Section 4.3 we will provide other examples of growth rate where on the contrary the fluctuations have a negative impact on the population size.

We consider, the rate function

$$R(e, z, I) = a(z) - B_1(e)B_2(I),$$

with $e(\cdot)$ a 1-periodic function and a, B_1 and B_2 smooth functions such that

$$0 < B_i, \qquad 0 < a_m \le a(z) \le a_M, \qquad 0 < \frac{\partial}{\partial I} B_2(I), \qquad \frac{\partial^2}{\partial I^2} B_2(I) < 0, \qquad \text{for } i = 1, 2,$$

and

$$\lim_{I \to 0} \min_{0 \le s \le 1} \left(a_m - B_1(e(s)) B_2(I) \right) > 0, \qquad \lim_{I \to +\infty} \max_{0 \le s \le 1} \left(a_M - B_1(e(s)) B_2(I) \right) < 0.$$

We also take $\psi \equiv 1$. Note that in [MPS15] we have shown that the analysis in Section 4.2.2 can be applied to such type of growth rate without any concavity assumption on a. Such analysis is obtained via an effective growth rate which is defined independently of the dominant trait \overline{z} . Here, we don't go through the details of such analysis. We only focus on the longtime outcome that can be determined using the same effective growth rate \mathcal{R} as above and under the assumption that there exists a unique $z_m \in \mathbb{R}^d$ such that

$$\max_{z \in \mathbb{R}^d} a(z) = a(z_m).$$

The goal is to compare the long time limit of the total population size for the model with fluctuations $(\bar{\rho}_{\infty})$ given by (4.17) to the one (ρ_{∞}^c) obtained from the model with the "averaged rate":

$$R_{\rm av}(z,I) = a(z) - B_{1,{\rm av}}B_2(I)$$
 with $B_{1,{\rm av}} = \int_0^1 B_1(e(s))ds$.

Later, we write f_{av} for the average of the 1-periodic map $f : \mathbb{R} \to \mathbb{R}$, i.e., $f_{av} = \int_0^1 f(s) ds$. In the constant case, the outcome can be obtained from the previous results [PB08, LMP11]: as $t \to \infty$ the dominant trait is given by $z_{\infty}^c = z_m$ and the total population size is given by

$$\rho_{\infty}^{c} = I_{\infty}^{c}, \qquad \text{such that} \qquad a(z_{m}) = B_{1,\text{av}}B_{2}(\rho_{\infty}^{c}). \tag{4.19}$$

In the periodic case, the outcome is given by Theorem 4.3: as $t \to \infty$ the dominant trait is given by $z_{\infty}^{c} = z_{m}$ and the total population size is given by

$$\rho_{\infty}^{p} = I_{\infty}^{p}, \quad \text{such that} \quad \rho_{\infty}^{p} = \int_{0}^{1} \mathcal{I}(z_{m}, s) ds,$$

where \mathcal{I} solves (4.13). Multiplying (4.13) by suitable functions of \mathcal{I} and integrating with respect to s, we obtain that

$$a(z_m) = \int_0^1 B_1(e(s)) B_2(\mathcal{I}(z_m, s)) ds \text{ and } \int_0^1 B_2(\mathcal{I}(z_m, s)) ds \ a(z_m) = \int_0^1 B_1(e(s)) B_2^2(\mathcal{I}(z_m, s)) ds.$$
(4.20)

A straightforward application of the Cauchy-Schwarz inequality in (4.20), since $B_2(\mathcal{I}(z_m, s))$ is not constant, yields

$$a(z_m)^2 < B_{1,av} a(z_m) \int_0^1 B_2(\mathcal{I}(z_m, s)) ds,$$

and thus

$$a(z_m) < B_{1,av} \int_0^1 B_2(\mathcal{I}(z_m, s)) ds.$$
 (4.21)

Combining (4.19) and (4.21) we obtain that

$$B_2(\rho_{\infty}^c) < \int_0^1 B_2(\mathcal{I}(z_m, s)) ds.$$

Using the concavity and the monotonicity of B_2 we obtain that

$$\rho_{\infty}^{c} < \int_{0}^{1} \mathcal{I}(z_{m}, s) ds = \overline{\rho}_{\infty}$$

which substantiates our claim about the possible effect of the time oscillations.
4.3 Adaptation to a shifting and oscillating environment

In this section, we study the evolutionary dynamics of a phenotypically structured population in an environment where the fitness landscape varies with a linear trend but in an oscillatory manner. We study the following non-local parabolic equation

$$\begin{cases} \partial_t \widetilde{n} - \sigma \partial_{zz} \widetilde{n} = \widetilde{n}[a(e(t), z - \widetilde{c}t) - \kappa \widetilde{\rho}(t)], & (t, z) \in [0, +\infty) \times \mathbb{R}, \\ \widetilde{\rho}(t) = \int_{\mathbb{R}} \widetilde{n}(t, z) dz, & (4.22) \\ \widetilde{n}(t = 0, z) = \widetilde{n}_0(z). \end{cases}$$

This equation models the dynamics of a population which is structured by a phenotypic trait $z \in \mathbb{R}$. Here, $\tilde{n}(t, z)$ corresponds to the density of individuals with trait z at time t. We denote by $a(e(t), z - \tilde{c}t)$ the intrinsic growth rate of an individual with trait z at time t. The term $-\tilde{c}t$ has been introduced to consider a variation of the fitness landscape with a linear trend. The term e is an environment state which varies periodically to consider fluctuations of the environment. The term $\tilde{\rho}$ which corresponds to the total size of the population represents a competition term. Here, we assume indeed a uniform competition between all the individuals with intensity κ . The diffusion term models the mutations, with σ a measure of the effect of the mutations.

A natural motivation to study such type of problem is the fact that many natural populations are subject both to directional change of phenotypic optimum and random fluctuations of the environment or seasonal changes ([BL95, CVT15]). Here, we consider a deterministic growth rate that varies with a linear trend but in an oscillatory manner as a first step to study such situations. Will the population be able to adapt to the environmental change? Is there a maximum speed above which the population will get extinct? How is such maximal speed modified due to the fluctuations? Such questions have been studied in the biological literature using related quantitative genetics models under simplifying assumptions and in particular considering Gaussian distributions (see for instance [BL95, LS96, GH09]).

In this work we prove indeed that there exists a critical speed of linear change above which the population will get extinct. When the speed of linear change is less than this critical speed we prove that the population persists and we characterize the long time behavior of the solution to (4.22) assuming small effect of mutations (σ small). We also provide an asymptotic expansion for the critical speed in terms of σ . These results allow us to provide analytic approximations for the moments of the phenotypic density. We provide such analytic approximations for several case studies leading to different biological behaviors. In particular, we investigate the role of the fluctuations in the ability of the population to follow the environment linear change.

The analysis in this section provides also new results in the case of pure environmental oscillations without change with linear trend (the case with c = 0) [FIM18]. Note that the work in Section 4.2 allows to study the dynamics of a population under fluctuating selection, while the work in [FIM18], corresponding to the case c = 0 in this section, focuses on the long time outcome of such fluctuating selection. In particular, this analysis allows us to provide some possible explanation for the outcome of the experience in [KMZ⁺13] presented above (see subsection 4.3.4.2). Note also that the results in Section 4.2 were obtained under concavity assumptions on the growth rate R. Here, we don't make any concavity assumption but we consider a linear dependence of the growth rate on the competition term I.

The work in [FIM18], corresponding to the case c = 0, is performed in the multi-d case, that is

 $z \in \mathbb{R}^d$. In [FIM] we assume $z \in \mathbb{R}$. However, we believe that the results may be extended to the multi-d case with no significant difficulty. Here, we present our results in the framework of [FIM] considering $z \in \mathbb{R}$. In subsection 4.3.1 we provide our assumptions and our preliminary results. In subsection 4.3.2 we present our main results. In subsection 4.3.3 we provide some heuristic computations which show on the one hand where the corresponding Hamilton-Jacobi equation comes from. On the other hand these computations lead to a system of equations for the next order term. The computation of the next order term allows us, as in subsection 3.4.2, to provide analytic approximations of the moments of the phenotypic distribution. Such analytic approximations are provided in 4.3.4 considering several case studies and show that depending on how the fluctuations modify the growth rate R different biological behaviors may be observed.

4.3.1 Assumptions and preliminary results

To introduce our assumptions, we first define

$$\overline{a}(y) = \frac{1}{T} \int_0^T a(e(t), y) dt.$$

We then assume that $e : \mathbb{R}^+ \to \Omega \subset \mathbb{R}$ is a periodic function and that, for all $t \in \mathbb{R}^+$ and $e_0 \in \Omega$,

$$e(t) = e(t+T), \qquad \|a\|_{L^{\infty}(\Omega \times \mathbb{R})} \le C_0, \qquad a(e_0, \cdot) \in C^3(\mathbb{R}).$$

$$(4.23)$$

In addition, the averaged function \overline{a} attains its maximum and

$$\max_{z \in \mathbb{R}} \ \overline{a}(z) > 0, \tag{4.24}$$

which means that there exist at least some traits with strictly positive average growth rate. Moreover, for some of our results we assume that there exists a unique optimal trait z_m in the average environment, that is

$$\exists ! \ z_m \in \mathbb{R} : \max_{z \in \mathbb{R}} \overline{a}(z) = \overline{a}(z_m), \tag{4.25}$$

and also

$$\exists ! \ \overline{z} \le z_m; \ \overline{a}(\overline{z}) + \frac{\widetilde{c}^2}{4\sigma} = \overline{a}(z_m).$$
(4.26)

Furthermore, we make the following assumption on the initial condition which means that the density of individuals with large trait in the initial population is exponentially small:

$$0 \le \widetilde{n}_0(z) \le e^{C_1 - C_2|z|}, \quad \forall z \in \mathbb{R},$$

$$(4.27)$$

for some positive constants C_1, C_2 .

To avoid the shift in the growth rate a, we transform our problem with a change of variable. We introduce indeed $n(t, z) = \tilde{n}(t, z + \tilde{c}t)$ which satisfies:

$$\begin{cases} \partial_t n - \widetilde{c} \partial_z n - \sigma \partial_{zz} n = n[a(e(t), z) - \rho(t)], & (t, z) \in [0, +\infty) \times \mathbb{R}, \\ \rho(t) = \int_{\mathbb{R}} n(t, y) dy, & (4.28) \\ n(t = 0, z) = n_0(z) = \widetilde{n}_0(z). \end{cases}$$

Next, we introduce the linearized problem associated to (4.28). Let $m(t,z) = n(t,z)e^{\int_0^t \rho(s)ds}$, for *n* the solution of (4.28), then *m* satisfies

$$\begin{cases} \partial_t m - \tilde{c} \partial_z m - \sigma \partial_{zz} m = a(e(t), z)m, & (t, z) \in [0, +\infty) \times \mathbb{R} \\ m(t = 0, z) = n_0(z), & z \in \mathbb{R} \end{cases}$$

We also introduce the corresponding parabolic eigenvalue problem as follows

$$\begin{cases} \partial_t p_{\tilde{c}} - \tilde{c} \partial_z p_{\tilde{c}} - \sigma \partial_{zz} p_{\tilde{c}} - a(e(t), z) p_{\tilde{c}} = \lambda_{\tilde{c}, \sigma} p_{\tilde{c}}, & (t, z) \in [0, +\infty) \times \mathbb{R}, \\ 0 < p_{\tilde{c}}(t, z); \ p_{\tilde{c}}(t, z) = p_{\tilde{c}}(t + T, z), & (t, z) \in [0, +\infty) \times \mathbb{R}. \end{cases}$$

$$(4.29)$$

We also define the eigenvalue problem in the bounded domain [-R, R], for some R > 0,

$$\begin{cases} \partial_t p_R - \tilde{c} \partial_z p_R - \sigma \partial_{xx} p_R - a(e(t), z) p_R = \lambda_R p_R, & (t, z) \in [0, +\infty) \times [-R, R], \\ p_R(t, z) = 0, & (t, z) \in [0, +\infty) \times \{-R, R\}, \\ 0 < p_R(t, z); \ p_R(t, z) = p_R(t + T, z), & (t, z) \in [0, +\infty) \times [-R, R]. \end{cases}$$
(4.30)

It is known that problem (4.30) has a unique eigenpair (λ_R, p_R) with p_R a strictly positive eigenfunction such that $\|p_R(0, \cdot)\|_{L^{\infty}([-R,R])} = 1$ (see [Hes91]). Another fundamental result (see for instance [HP08]), for our purpose is that the function $R \mapsto \lambda_R$ is decreasing and $\lambda_R \to \lambda_{\tilde{c},\sigma}$ as $R \to +\infty$.

We next assume that there exist positive constants δ and R_0 such that

$$a(e(t), z) + \lambda_{\tilde{c}, \sigma} \le -\delta, \quad \text{for all } 0 \le t, \text{ and } R_0 \le |z|,$$

$$(4.31)$$

which means that large traits have negative enough growth rates. Under the above additional assumption, there exists a unique principal eigenpair ($\lambda_{\tilde{c},\sigma}, p_{\tilde{c}}$) for (4.29) [HP08]:

Proposition 4.5. Assume (4.23), (4.27) and (4.31). Then for problem (4.29) there exists a unique principal eigenfunction $p_{\tilde{c}}$ associated to $\lambda_{\tilde{c},\sigma}$, with $\|p_{\tilde{c}}(0,\cdot)\|_{L^{\infty}(\mathbb{R})} = 1$. Moreover, we have $p_{\tilde{c}} = \lim_{R \to \infty} p_R$ and

$$p_{\tilde{c}}(t,z) \le \|p_{\tilde{c}}\|_{L^{\infty}} e^{-\nu(|z|-R_0)}, \qquad \forall (t,z) \in [0,+\infty) \times \mathbb{R},$$

for $\nu = -\frac{\tilde{c}}{2\sigma} + \sqrt{\frac{\delta}{\sigma} + \frac{1}{2} \left(\frac{\tilde{c}}{\sigma}\right)^2}$. Finally, the eigenfunction $p_{\tilde{c}}($

Finally, the eigenfunction $p_{\tilde{c}}(t,z)$ is exponentially stable, in the following sense; there exists $\alpha > 0$ such that:

$$\|m(t,z)e^{t\lambda_{\tilde{c},\sigma}} - \alpha p_{\tilde{c}}(t,z)\|_{L^{\infty}(\mathbb{R})} \to 0 \quad exponentially \ fast \ as \ t \to \infty.$$

We next define the *T*-periodic functions $Q_{\tilde{c}}(t)$ and $P_{\tilde{c}}(t,z)$ as follows:

$$Q_{\tilde{c}}(t) = \frac{\int_{\mathbb{R}} a(e(t), z) p_{\tilde{c}}(t, z) dz}{\int_{\mathbb{R}} p_{\tilde{c}}(t, z) dz}, \quad P_{\tilde{c}}(t, z) = \frac{p_{\tilde{c}}(t, z)}{\int_{\mathbb{R}} p_{\tilde{c}}(t, z) dz},$$

Finally we state a Lemma which will be useful for the next section.

Lemma 4.6. There exists a positive solution $\hat{\rho}(t)$ to the problem

$$\begin{cases} \frac{d\widehat{\rho}}{dt} = \widehat{\rho} \left[Q_{\widehat{c}}(t) - \widehat{\rho} \right], \quad t \in (0, T), \\ \widehat{\rho}(0) = \widehat{\rho}(T), \end{cases}$$
(4.32)

if and only if $\int_0^T Q_{\tilde{c}}(t)dt > 0$. Moreover, this solution is unique.

4.3.2 Main results

First, we determine conditions on the environment shift speed \tilde{c} which lead to extinction or survival of the population. Next, in the case of the population survival, we characterize asymptotically the population density considering small effect of the mutations.

To present our result on the survival criterion, we define the "critical speed".

Definition 4.7. We define the critical speed \tilde{c}^*_{σ} as follows

$$\widetilde{c}_{\sigma}^{*} = \begin{cases} 2\sqrt{-\sigma\lambda_{0,\sigma}}, & \text{if } \lambda_{0,\sigma} < 0, \\ 0, & \text{otherwise,} \end{cases}$$

where $\lambda_{0,\sigma}$ corresponds to the principal eigenvalue introduced by Proposition 4.5, in the case c = 0.

The next result shows that \tilde{c}_{σ}^* is indeed a critical speed of climate change above which the population goes extinct.

Proposition 4.8 (Survival criterion [FIM]). Let n(t, z) be the solution of (4.28). Assume (4.23), (4.24), (4.27) and (4.31). Then the following statements hold:

- (i) if $\tilde{c} \geq \tilde{c}_{\sigma}^*$, then the population will go extinct, i.e. $\rho(t) \to 0$, as $t \to \infty$,
- (ii) if $\tilde{c} < \tilde{c}_{\sigma}^*$, then $|\rho(t) \hat{\rho}(t)| \to 0$, as $t \to \infty$, with $\hat{\rho}$ the unique solution to (4.32).
- (iii) Moreover, $\left\|\frac{n(t,z)}{\rho(t)} P_{\tilde{c}}(t,z)\right\|_{L^{\infty}} \longrightarrow 0$, as $t \to \infty$. Consequently we have, as $t \to \infty$:

$$\|n(t,\cdot) - \widehat{\rho}(t)P_{\widetilde{c}}(t,\cdot)\|_{L^{\infty}} \to 0$$
, if $\widetilde{c} < \widetilde{c}_{\sigma}^*$ and $\|n\|_{L^{\infty}} \to 0$, if $\widetilde{c} \ge \widetilde{c}_{\sigma}^*$.

Remark 4.9. Note that if $\lambda_{0,\sigma} \ge 0$, then $\tilde{c}^*_{\sigma} = 0$ which means that the population goes extinct even without climate change, that is with $\tilde{c} = 0$.

Proposition 4.8 shows that if the environment change occurs "too fast" the population will not be able to follow the environment change and will get extinct. However, if the change speed is "moderate" the phenotypic density n converges to the periodic function $n_{\tilde{c}}(t,z) = \hat{\rho}(t)P_{\tilde{c}}(t,z)$, which is in fact the unique periodic solution of (4.28).

Next, we are interested in describing this periodic solution $n_{\tilde{c}}$, asymptotically as the effect of the mutations is small. To this end, we take $\sigma = \varepsilon^2$ and $\tilde{c} = \varepsilon c$, with ε small, and we study asymptotically the solution $(n_{\varepsilon c}, \hat{\rho}_{\varepsilon c})$ as ε vanishes. For better legibility, we also define $c_{\varepsilon}^* := \frac{\tilde{c}_{\varepsilon^2}^*}{\varepsilon}$ where $\tilde{c}_{\varepsilon^2}^*$ stands for the critical speed \tilde{c}_{σ}^* with $\sigma = \varepsilon^2$. Note that, in view of Proposition 4.8, to provide an asymptotic analysis considering $\sigma = \varepsilon^2$ small, a rescaling of the climate shift speed as $\tilde{c} = \varepsilon c$ is necessary. The population can tolerate only a climate change with small speed if the mutations have small effect. Otherwise the phenotypic variability is not high enough for the population to adapt to the climate change.

In order to keep the notations simpler we denote $(n_{\varepsilon c}, \hat{\rho}_{c\varepsilon}) = (n_{\varepsilon}, \rho_{\varepsilon})$, which is the unique periodic solution of the problem:

$$\begin{cases} \partial_t n_{\varepsilon} - \varepsilon c \partial_z n_{\varepsilon} - \varepsilon^2 \partial_{zz} n_{\varepsilon} = n_{\varepsilon} [a(e(t), z) - \rho_{\varepsilon}(t)], \quad (t, z) \in [0, +\infty) \times \mathbb{R}, \\ \rho_{\varepsilon}(t) = \int_{\mathbb{R}} n_{\varepsilon}(t, z) dz, \\ n_{\varepsilon}(0, z) = n_{\varepsilon}(T, z). \end{cases}$$

$$(4.33)$$

To study asymptotically this problem we perform a Hopf-Cole transformation

$$n_{\varepsilon} = \frac{1}{\sqrt{2\pi\varepsilon}} \exp\left(\frac{u_{\varepsilon}}{\varepsilon}\right). \tag{4.34}$$

Here is our first main result:

Theorem 4.10 (Limit as $\varepsilon \to 0$ [FIM]). Assume (4.23), (4.24) and (4.31) and also that $c < \liminf_{\varepsilon \in C} c_{\varepsilon}^*$. Then the following statements hold:

- (i) As $\varepsilon \to 0$, we have $\|\rho_{\varepsilon}(t) \widetilde{\varrho}(t)\|_{L^{\infty}} \to 0$, with $\widetilde{\varrho}(t)$ a *T*-periodic function.
- (ii) Moreover, as $\varepsilon \to 0$, $u_{\varepsilon}(t, z)$ converges locally uniformly to a function $u(z) \in C(\mathbb{R})$, a viscosity solution to the following equation:

$$\begin{cases}
-\left|\partial_z u + \frac{c}{2}\right|^2 = \overline{a}(z) - \overline{\rho} - \frac{c^2}{4}, \quad z \in \mathbb{R}, \\
\max_{z \in \mathbb{R}} u(z) = 0, \\
-A_1|z|^2 - \frac{c}{2}z - A_2 \leq u \leq c_1 - c_2|z|,
\end{cases}$$
(4.35)

with

$$\bar{\rho} = \int_0^T \widetilde{\varrho}(t) dt,$$

and some positive constants A_1, A_2, c_1 and $c_2 = -\frac{c}{2} + \sqrt{\delta + \frac{c^2}{2}}$.

The prove of the theorem above is based on appropriate regularity estimates on u_{ε} . A main difficulty to obtain such estimates, comparing to the previous related works (see for instance [BMP09]), comes from the time periodicity condition in this problem.

Next, under the additional assumptions (4.26)–(4.31) we provide the uniqueness and the identification of the limit:

Theorem 4.11 (Uniqueness and identification of the limit [FIM]). Asssume (4.23)–(4.26), (4.31) and that $c < \liminf_{\epsilon \to 0} c_{\epsilon}^*$. Then the following statements hold:

(i) The viscosity solution of (4.35) is unique and it is indeed a classical solution given by

$$u(z) = \frac{c}{2}(\overline{z} - z) + \int_{\overline{z}}^{z_m} \sqrt{\overline{a}(z_m) - \overline{a}(y)} dy - \left| \int_{z_m}^z \sqrt{\overline{a}(z_m) - \overline{a}(y)} dy \right|,$$

where $\overline{z} < z_m$ is given in (4.26).

(ii) Furthermore, let n_{ε} solve (4.33), then

$$n_{\varepsilon}(t,z) - \widetilde{\varrho}(t)\delta(z-\overline{z}) \rightharpoonup 0, \quad as \ \varepsilon \to 0,$$

pointwise in time, weakly in z in the sense of measures, with $\tilde{\varrho}$ the unique periodic solution of the following equation

$$\left\{ \begin{array}{ll} \displaystyle \frac{d\widetilde{\varrho}}{dt} = \widetilde{\varrho} \left[a(e(t), \bar{z}) - \widetilde{\varrho} \right], \quad t \in (0, T), \\ \displaystyle \widetilde{\varrho}(0) = \widetilde{\varrho}(T). \end{array} \right.$$

Remark 4.12. The statement (ii) in Theorem 4.11 implies for the solution \tilde{n}_{ε} to the original problem (4.22) with $\sigma = \varepsilon^2$ and $\tilde{c} = c\varepsilon$ that

$$\widetilde{n}_{\varepsilon}(t,z) - \widetilde{\varrho}(t)\delta(x - \overline{z} - ct) \rightharpoonup 0, \quad as \ \varepsilon \to 0,$$

pointwise in time, weakly in z in the sense of measures. This implies that the phenotypic density of the population concentrates on a dominant trait which follows the optimal trait with the same speed but with a constant lag $z_m - \overline{z}$.

Note that the proof of the uniqueness of the viscosity solution to (4.35) is not straight forward. To prove such result, we first introduce a new function $\tilde{u}(x) = u(x) + \frac{c}{2}x$ which solves

$$\begin{cases} -|\partial_z \widetilde{u}|^2 = \overline{a}(z) - \overline{\rho} - \frac{c^2}{4}, \quad z \in \mathbb{R}, \\ -A_1|z|^2 - A_2 \le \widetilde{u}(z) \le c_1 - c_2|z| + \frac{c}{2}z, \end{cases}$$
(4.36)

where the constants A_1 , A_2 , c_1 , c_2 are the same as in (4.35). Then a main ingredient comes from the fact that any viscosity solution to a Hamilton-Jacobi equation of type (4.36) but in a bounded domain Γ can be uniquely determined by its values on the boundary points of Γ and by its values at the maximum points of the r.h.s of the Hamilton-Jacobi equation [Lio82b]. However, here we have an unbounded domain. To control the trajectories coming from infinity we use indeed the bounds above on $\tilde{u}(z)$ and prove that the solution to (4.36) is indeed determined by its value at the maximum point of \bar{a} that is z_m . Finally the constraint max_z u(z) = 0 allows us to identify u.

Finally, to present our last result, let us consider the eigenproblem (4.29) for $\sigma = \varepsilon^2$ and $\tilde{c} = c\varepsilon$, that is:

$$\begin{cases} \partial_t p_{c\varepsilon} - \varepsilon c \partial_z p_{c\varepsilon} - \varepsilon^2 \partial_{zz} p_{c\varepsilon} - a(e(t), z) p_{c\varepsilon} = p_{c\varepsilon} \lambda_{c,\varepsilon}, & (t, z) \in [0, +\infty) \times \mathbb{R}, \\ 0 < p_{c\varepsilon}; \ p_{c\varepsilon}(t, z) = p_{c\varepsilon}(t + T, z), & (t, z) \in [0, +\infty) \times \mathbb{R}. \end{cases}$$
(4.37)

Here $\lambda_{c,\varepsilon}$ denotes the eigenvalue $\lambda_{\tilde{c},\sigma}$ with $\sigma = \varepsilon^2$ and $\tilde{c} = c\varepsilon$ for better legibility. We next provide an asymptotic expansion for λ_{ε} and consequently for $\overline{\rho}_{\varepsilon}$ and c_{ε}^* :

Theorem 4.13. (Asymptotic expansions [FIM]) Let $\lambda_{c,\varepsilon}$ be the principal eigenvalue of problem (4.37) and assume (4.23), (4.25) and (4.31). Assume in addition that $c < \liminf_{\varepsilon \to 0} c_{\varepsilon}^*$, then the following asymptotic expansions hold

$$\overline{\rho}_{\varepsilon} = -\lambda_{c,\varepsilon} = \overline{a}(z_m) - \frac{c^2}{4} - \varepsilon \sqrt{-\overline{a}_{zz}(z_m)/2} + o(\varepsilon),$$
$$c_{\varepsilon}^* = 2\sqrt{\overline{a}(z_m)} - \varepsilon \sqrt{-\frac{\overline{a}_{zz}(z_m)}{2\,\overline{a}(z_m)}} + o(\varepsilon).$$

Such expansion is indeed related to the harmonic approximation of the energy of the ground state of the Shrödinger operator ([Hel04]). However, here we have a parabolic, non self-adjoint operator.

4.3.3 Heuristics and the next order term

We next provide some heuristic computations which allow us to derive the Hamilton-Jacobi equation (4.35) but also to identify the corrector term v, with the ansatz

$$u_{\varepsilon}(t,z) = u(t,z) + \varepsilon v(t,z) + \varepsilon^2 w(t,z) + o(\varepsilon^2), \quad \rho_{\varepsilon}(t) = \widetilde{\rho}(t) + \varepsilon \kappa(t) + o(\varepsilon), \quad (4.38)$$

where the coefficients of the developments are time periodic. We also define

$$\overline{\rho} = \frac{1}{T} \int_0^T \widetilde{\rho}(t) dt, \qquad \overline{\kappa} = \frac{1}{T} \int_0^T \kappa(t) dt.$$

Replacing (4.34) in (4.33), we first notice that u_{ε} solves

$$\begin{cases} \frac{1}{\varepsilon} \partial_t u_{\varepsilon} - c \partial_z u_{\varepsilon} - \varepsilon \partial_{zz} u_{\varepsilon} &= |\nabla u_{\varepsilon}|^2 + a(e(t), z) - \rho_{\varepsilon}(t), \quad (t, z) \in [0, +\infty) \times \mathbb{R}^d, \\ u_{\varepsilon}(t = 0, z) &= u_{\varepsilon}^0(z) = \varepsilon \ln n_{\varepsilon}^0(z). \end{cases}$$
(4.39)

We then substitute (4.38) into (4.39) and obtain:

$$\frac{1}{\varepsilon}\partial_t u + \partial_t v - c\partial_z u - c\varepsilon\partial_z v - \varepsilon\partial_{zz} u - \varepsilon^2\partial_{zz} v = |\partial_z u + \varepsilon\partial_z v|^2 + a(e(t), z) - \widetilde{\rho} - \varepsilon\kappa + o(\varepsilon).$$

We keep respectively the ε^{-1} and ε^{0} order terms to obtain

$$\partial_t u(t,z) = 0 \Leftrightarrow u(t,z) = u(z),$$

and

$$\partial_t v - c \partial_z u = |\nabla u|^2 + a(e(t), z) - \widetilde{\rho}.$$

Integrating this latter equation in $t \in [0, T]$ and using the *T*-periodicity of *v*, leads to the first equation in (4.35).

Finally, the corrector terms v and $\overline{\kappa}$ can be determined by the equations obtained by keeping respectively the ε^0 and ε order terms:

$$\begin{cases} \partial_t v = \left| \partial_z u + \frac{c}{2} \right|^2 + a(e(t), z) - \frac{c^2}{4} - \widetilde{\rho}, \\ -\partial_{zz} u = [2\partial_z u + c] \frac{1}{T} \int_0^T \partial_z v(t, z) dt - \overline{\kappa}. \end{cases}$$
(4.40)

Once u and v are known, one can provide analytic approximations for the moments of the phenotypic distribution following similar arguments as in Section 3.4.2.

4.3.4 Biological applications

In this section we discuss the effect of the periodic fluctuations on the critical speed of survival and the dynamics of the phenotypic density considering the following particular form of growth rate, taking into account a quadratic stabilizing selection ([B00]– pages 117-121 and chapter VI),

$$a(e, x) = r - g(e)(z - \theta(e))^2.$$
(4.41)

Here, r is a positive constant corresponding to the maximal intrinsic growth rate. The function θ represents the optimal trait and the positive function g represents the pressure of the selection, both being functions of the environment state e.

In subsection 4.3.4.1 we will provide some preliminary computations considering above growth rate which will be helpful in the following subsections.

In subsection 4.3.4.2 we will consider the above growth rate with c = 0, that is the case where we have pure oscillations. Being motivated by the experiment of [KMZ⁺13] we investigate in which situations a population that has evolved in a periodic environment will outperform a population that has evolved in an average constant environment when both populations are placed at the same constant environment. We show indeed that when the fluctuations act on the pressure of the selection, that is when g is a function of the periodic environment state e but when θ is constant, then such property may be observed.

In subsection 4.3.4.3 we will consider the same growth rate but with c > 0. We show in this case that, when the population persists, the phenotypic mean follows the dominant trait with the same speed but with a constant lag. We also show that when the fluctuations act on the pressure of the selection, the periodic fluctuations may help the population to follow the linear environment change. More precisely, we show that in presence of periodic fluctuations, the critical speed of survival may be greater than the case without fluctuations of the environment.

4.3.4.1 Preliminary computations

In this subsection, using our method we provide some preliminary computations that we will use in the following subsections. We compute the mean of a(e, z)

$$\overline{a}(z) = \int_0^1 a(e(t), z) dt = r - z^2 \overline{g} + 2zg_1 - g_2,$$

where

$$\bar{g} = \int_0^1 g(e(t))dt, \quad g_1 = \int_0^1 g(e(t))\theta(e(t))dt, \quad g_2 = \int_0^1 g(e(t))\theta^2(e(t))dt,$$

and we observe that the maximum of $\overline{a}(z)$ is attained at $z_m = \frac{g_1}{\overline{a}}$, with

$$\bar{a}(z_m) = r + \frac{g_1^2}{\bar{g}} - g_2.$$

From Theorem 4.11-(ii) we obtain that u(z) is given by

$$u(z) = -\frac{\sqrt{\bar{g}}}{2} (z - \bar{z})^2, \qquad \bar{z} = z_m - \frac{c}{2\sqrt{\bar{g}}} = \frac{g_1}{\bar{g}} - \frac{c}{2\sqrt{\bar{g}}}.$$

To present our results we introduce the following notations:

$$\begin{cases} \overline{\rho}_{\varepsilon,p} = \frac{1}{T} \int_{0}^{T} \rho_{\varepsilon}(t) dt, \\ \mu_{\varepsilon,p}(t) = \frac{1}{\rho_{\varepsilon,p}(t)} \int_{\mathbb{R}^{d}} z \ n_{\varepsilon}(t,z) dz, \\ \sigma_{\varepsilon,p}^{2}(t) = \frac{1}{\rho_{\varepsilon}(t)} \int_{\mathbb{R}^{d}} (z - \mu_{\varepsilon}(t))^{2} n_{\varepsilon}(t,z) dz, \\ F_{\varepsilon,p}(e_{0}) = \int_{\mathbb{R}^{d}} a(e_{0},z) \frac{1}{T} \int_{0}^{T} \frac{n_{\varepsilon}(t,z)}{\rho_{\varepsilon}(t)} dt dz. \end{cases}$$

Here, $\overline{\rho}_{\varepsilon,p}$ corresponds to the average population size of a population evolved for a long time in the periodic environment. The terms $\mu_{\varepsilon,p}(t)$ and $\sigma_{\varepsilon,p}^2(t)$ correspond respectively to the mean phenotypic trait and the phenotypic variance of such population. The last term $F_{\varepsilon,p}(e_0)$ corresponds to the average growth rate of such population when it is placed at environment state e_0 . This quantity corresponds to what was measured in the experiment of [KMZ⁺13].

The asymptotic expansions in Theorem 4.13 imply that

$$\overline{\rho}_{\varepsilon,p} = r + \frac{g_1^2}{\overline{g}} - g_2 - \frac{c^2}{4} - \varepsilon \sqrt{\overline{g}} + o(\varepsilon), \quad c_{\varepsilon,p}^* = 2\sqrt{r + \frac{g_1^2}{\overline{g}}} - g_2 - \varepsilon \sqrt{\frac{\overline{g}}{r + \frac{g_1^2}{\overline{g}} - g_2}} + o(\varepsilon).$$

Furthermore, following the arguments in Section 3.4.2, we can also obtain formally an approximation of order ε for the phenotypic mean $\mu_{\varepsilon,p}$ and the variance $\sigma_{\varepsilon,p}^2$ of the phenotypic distribution, that is:

$$\mu_{\varepsilon,p}(t) = \frac{g_1}{\bar{g}} - \frac{c}{2\sqrt{\bar{g}}} + \varepsilon D(t) + o(\varepsilon), \qquad \sigma_{\varepsilon,p}^2(t) = \frac{\varepsilon}{\sqrt{\bar{g}}} + o(\varepsilon)$$

where $D(t) = \partial_z v(\bar{z}, t)$ for v the solution of the system (4.40). One can verify that for this growth rate, and choosing the period T = 1, we have

$$D(t) = -c\sqrt{\bar{g}}\left(t - \frac{1}{2}\right) + 2\int_0^1 \int_0^\tau g(s)(\bar{z} - \theta(s))dsd\tau - 2\int_0^t g(s)(\bar{z} - \theta(s))ds.$$

4.3.4.2 No environment change with linear trend (c = 0); A comparison with the experiment of [KMZ⁺13].

In this subsection, we apply our analysis to two biological case studies considering c = 0. This analysis is motivated by the experiment in [KMZ⁺13] (see the introduction of Chapter 4 for the presentation of this biological experiment). We are interested in the comparison of the phenotypic distributions of populations evolved for a long time whether in a periodic environment or in a constant environment at an average state. We seek in particular to determine conditions under which the population evolved in a time periodic environment, with a *T*-periodic state e(t)(*e* corresponding for instance to the temperature), will outperform a population evolved in a constant environment (with state $\bar{e} = \frac{1}{T} \int_0^T e(s) ds$) when both strains are placed at the constant environment \bar{e} .

We will consider the quadratic stabilizing selection (4.41) presented above, in two particular cases. First, we will assume that the variations of the environment modify the optimal trait but have no influence on the pressure of the selection (g(e) = g > 0 is constant). Next, we will assume that the environment variations modify the pressure of the selection but have no

influence on the optimal trait $(\theta(e) = 0)$.

Oscillations on the optimal trait. For this case study, we take

$$g(e) = g > 0, \qquad \theta(e) = e, \qquad e(t) = l \sin\left(\frac{2\pi}{b}t\right).$$
 (4.42)

Then, considering a population evolved in such a periodic environment, the average size of the population, the mean phenotypic trait, the phenotypic variance and the mean fitness of the population in an environment with state $\bar{e} = 0$, are given by

$$\overline{\rho}_{\varepsilon,p} = r - \frac{gl^2}{2} - \varepsilon\sqrt{g} + o(\varepsilon), \quad \mu_{\varepsilon,p}(t) = \frac{\varepsilon lb\sqrt{g}}{\pi} \sin\left(\frac{2\pi}{b}(t - \frac{b}{4})\right) + o(\varepsilon)$$
$$\sigma_{\varepsilon,p}^2(t) = \frac{\varepsilon}{\sqrt{g}} + o(\varepsilon), \quad F_{\varepsilon,p}(e_0 = 0) = r - \varepsilon\sqrt{g} + o(\varepsilon).$$

We observe indeed that the mean trait $\mu_{\varepsilon,p}(t)$ oscillates with the same period as the optimal trait but with a delay $\frac{b}{4}$ and a small amplitude (such property was obtained for a related model in [LS96]).

We next compare these approximations with the ones obtained for a population evolved in a constant environment with $\theta(e) = \theta(\overline{e} = 0) = 0$ and g(e) = g:

$$\rho_{\varepsilon,0} = r - \varepsilon \sqrt{g}, \quad \mu_{\varepsilon,0} = 0, \quad \sigma_{\varepsilon,0}^2 = \frac{\varepsilon}{\sqrt{g}}, \quad F_{\varepsilon,0}(e_0 = 0) = r - \varepsilon \sqrt{g}.$$
(4.43)

Note that these quantities correspond to the approximations obtained in subsection 4.3.4.1 with $\theta(e) = 0$ and g(e) = g. However, here these are exact values and not only approximations for ε small, since the problem with constant g and θ can indeed be solved explicitly.

We deduce that both populations (the one evolved in the constant environment $e_0 = 0$ and the other evolved in the fluctuating environment) have the same mean fitness, up to order ε , when they are placed at the same constant environment $e_0 = 0$.

Oscillations on the pressure of the selection. For this case study, we take

 $\theta(e) = 0$, g(e) a positive function of e and e(t) a 1-periodic function of time. (4.44)

We also define

$$\overline{g} = \int_0^1 g(e(s)) ds.$$

Then, considering a population evolved in such a periodic environment, the average size of the population, the mean phenotypic trait, the phenotypic variance and the mean fitness of the population in an environment with state \bar{e} , are given by

$$\overline{\rho}_{\varepsilon} = r - \varepsilon \sqrt{\overline{g}} + o(\varepsilon), \quad \mu_{\varepsilon,p} = o(\varepsilon), \quad \sigma_{\varepsilon,p}^2 = \frac{\varepsilon}{\sqrt{\overline{g}}} + o(\varepsilon), \qquad F_{\varepsilon,p}(e_0 = \overline{e}) = r - \varepsilon \frac{g(\overline{e})}{\sqrt{\overline{g}}} + o(\varepsilon).$$

We next compare these quantities with the ones corresponding to a population evolved in a constant environment with average state \overline{e} , given by (4.43). We remark that if we choose g(e) such that

$$g(\overline{e}) < \overline{g},\tag{4.45}$$

then we obtain that, for ε small,

$$\overline{\rho}_{\varepsilon,p} < \rho_{\varepsilon,0}, \quad \mu_{\varepsilon,p} \approx \mu_{\varepsilon,0} = 0, \quad \sigma_{\varepsilon,p}^2 < \sigma_{\varepsilon,0}^2 \quad \text{and} \quad F_{\varepsilon,0}(\overline{e}) < F_{\varepsilon,p}(\overline{e}).$$

We hence obtain that, depending on the choice of g, the population which evolved in a fluctuating environment may have a larger fitness than the one evolved in a constant environment \overline{e} , when they are both placed at the same constant environment \overline{e} . This property corresponds to what was observed in the biological experiment in [KMZ⁺13]. Note that both of these environments select for populations with the same mean phenotypic trait z = 0. However, the population evolved in a periodic environment has a smaller variance comparing to the one evolved in a constant environment. This makes the population evolved in the periodic environment more performant.

Discussion. The above approximations show that, when the mutational effect ε is small, and when the environment variations modify the pressure of the selection, the population evolved in a time periodic environment may outperform a population evolved in a constant environment (with average state \overline{e}) when both strains are placed at the constant environment \overline{e} . By contrast, when the environment variations modify the optimal trait, the two environments select for populations with approximately similar performance. S. Figuera, in her PhD thesis [FI19], extends these results to the case where the mutations have larger effects using numerical simulations and also by obtaining semi-explicit solutions for the case studies above, following the method of [ABF⁺19]. She shows on the one hand that, when the environment variations modify the optimal trait (case (4.42)) and when the mutations have large effects, the mean fitness of the population evolved in a periodic environment is smaller than the mean fitness of the population evolved in a constant environment. Larger rate of mutations increases indeed the cost of fluctuations in this case. On the other hand, when the environment variations modify the pressure of the selection (case (4.44)), then the fluctuating environment may still select for a more performant population.

4.3.4.3 Environment change with linear trend (c > 0); the population follows the optimum with a constant lag

In this subsection, we provide our results for the case c > 0, in the two case studies where the oscillations act whether on the optimal trait or on the pressure of the selection. We show on the one hand, in both cases, that the average phenotypic mean follows the average optimal trait with a constant lag. On the other hand, we show that when the environment variations modify the pressure of the selection, the periodic fluctuations may help the population to follow the environment change.

Oscillations on the optimal trait. We take g(e) = g constant and $\theta(e)$ a function of the environment state e(t), which is itself 1-periodic. We also define

$$\overline{\theta} = \int_0^1 \theta(e(t)) dt.$$

We next compute the average population size, the average mean phenotypic trait and the critical speed $c_{\varepsilon,n}^*$ following the method presented above:

$$\begin{split} \overline{\rho}_{\varepsilon,p} &= r + g \left[\overline{\theta}^2 - \int_0^1 \theta^2(e(t)) dt \right] - \frac{c^2}{4} - \varepsilon \sqrt{g} + o(\varepsilon), \qquad \overline{\mu}_{\varepsilon,p} = \int_0^1 \mu_{\varepsilon,p}(t) dt = \overline{\theta} - \frac{c}{2\sqrt{g}} + o(\varepsilon), \\ c_{\varepsilon,p}^* &= 2 \sqrt{r + g \left[\overline{\theta}^2 - \int_0^1 \theta^2(e(t)) dt \right]} - \varepsilon \sqrt{\frac{g}{r + g \left[\overline{\theta}^2 - \int_0^1 \theta^2(e(t)) dt \right]}} + o(\varepsilon). \end{split}$$

Note in view of Remark 4.12 that, the above approximation of the average phenotypic mean indicates that the population follows the average optimal trait with a constant lag which is approximated by $\frac{c}{2\sqrt{g}}$.

In order to study the effect of the fluctuations on the adaptation of the population to the environment change we consider two sub-cases where e is constant or periodic.

a) If $e(t) \equiv e_0$ is constant we obtain that

$$\overline{\rho}_{\varepsilon,0} = r - \frac{c^2}{4} - \varepsilon \sqrt{g}, \quad \overline{\mu}_{\varepsilon,0} = \theta(e_0) - \frac{c}{2\sqrt{g}}, \quad c^*_{\varepsilon,0} = 2\sqrt{r} - \varepsilon \sqrt{\frac{g}{r}} + o(\varepsilon).$$

b) If e(t) is a 1-periodic function then, $\bar{\theta}^2 < \int_0^1 \theta^2(e(t))dt$ and we obtain that, for ε small,

$$\overline{\rho}_{\varepsilon,p} < r - \frac{c^2}{4} - \varepsilon \sqrt{g}, \quad \left\langle \mu_{\varepsilon,p}(t) \right\rangle \approx \overline{\theta} - \frac{c}{2\sqrt{g}}, \quad c^*_{\varepsilon,p} < 2\sqrt{r} - \varepsilon \sqrt{\frac{g}{r}}.$$

Thus, by keeping the pressure of selection constant, we deduce that, for ε small,

$$\overline{\rho}_{\varepsilon,p} < \overline{\rho}_{\varepsilon,0}$$
 and $c^*_{\varepsilon,p} < c^*_{\varepsilon,0}$.

This means that having an oscillating optimal trait is not beneficial for the population, in the sense that the mean total size of the population decreases with respect to the case with a constant optimal trait and the critical speed which leads the population to extinction is smaller in the periodic case.

Oscillations on the pressure of the selection. We take $\theta(e) = \theta$ constant and g(e) a function of the environment state e(t), which is itself 1-periodic. Note that, in such a case $g_1 = \bar{g}\theta$ and $g_2 = \bar{g}\theta^2$. We compute

$$\overline{\rho}_{\varepsilon,p} = r - \frac{c^2}{4} - \varepsilon \sqrt{\overline{g}} + o(\varepsilon), \qquad \overline{\mu}_{\varepsilon,p}(t) = \theta - \frac{c}{2\sqrt{\overline{g}}} + o(\varepsilon), \qquad c^*_{\varepsilon,p} = 2\sqrt{r} - \varepsilon \sqrt{\frac{\overline{g}}{r}} + o(\varepsilon).$$

We compare then, the sub-cases where e is constant or periodic.

a) If $e(t) \equiv e_0$ is constant, we obtain that

$$\overline{\rho}_{\varepsilon,0} = r - \frac{c^2}{4} - \varepsilon \sqrt{g(e_0)}, \quad \overline{\mu}_{\varepsilon,0}(t) = \theta - \frac{c}{2\sqrt{g(e_0)}}, \quad c^*_{\varepsilon,0} \approx 2\sqrt{r} - \varepsilon \sqrt{\frac{g(e_0)}{r}}.$$

b) If e(t) is a 1-periodic function then we obtain

$$\overline{\rho}_{\varepsilon,p} = r - \frac{c^2}{4} - \varepsilon \sqrt{\overline{g}} + o(\varepsilon), \quad \overline{\mu}_{\varepsilon,p}(t) = \theta - \frac{c}{2\sqrt{\overline{g}}} + o(\varepsilon), \quad c^*_{\varepsilon,p} = 2\sqrt{r} - \varepsilon \sqrt{\frac{\overline{g}}{r}} + o(\varepsilon).$$

In this case we deduce that if we choose the functions g and e such that

$$\bar{g} < g(e_0), \tag{4.46}$$

then we obtain that, for ε small,

$$\overline{\rho}_{arepsilon,0} < \overline{
ho}_{arepsilon,p} \quad ext{ and } \quad c^*_{arepsilon,0} < c^*_{arepsilon,p},$$

This means that the mean total size of the population increases with respect to the case with a constant selection pressure. Moreover, the critical speed above which the population goes extinct is larger in the periodic case. This means that the periodic fluctuations can help the population to follow the environment change.

Note that the condition (4.46) imposed to g is the opposite to the one imposed by (4.45) leading to more performant populations. There, we saw that in presence of the mutations and while the fluctuations act on the pressure of the selection (that is with a similar growth rate, however with c = 0 and under the condition $\bar{g} > g(e_0)$), a fluctuating environment can select for a population with smaller variance and in this way lead to more performant populations. What is beneficial in a (in average) constant environment may indeed be disadvantageous in a changing environment.

4.4 Evolutionary dynamics of a population in a piecewise constant environment with slow switch

In this section we study another type of time varying environment where the environment state is a piecewise constant function. The model is the same as (4.1), that we rewrite as below, taking $\sigma = \varepsilon^2$,

$$\begin{cases} \frac{\partial}{\partial t}\widetilde{n}_{\varepsilon} - \varepsilon^{2}\Delta\widetilde{n}_{\varepsilon} = \widetilde{n}_{\varepsilon}R(\widetilde{e}(t), z, \widetilde{\rho}_{\varepsilon}), \quad z \in \mathbb{R}^{d}, \\ \widetilde{\rho}_{\varepsilon}(t) = \int_{\mathbb{R}^{d}}\widetilde{n}_{\varepsilon}(t, y)\psi(y) \, dy, \\ \widetilde{n}_{\varepsilon}(0, z) = \widetilde{n}_{\varepsilon, 0}(z). \end{cases}$$

The meaning of the terms are the same as in (4.1). The environment state $\tilde{e}(t)$ is a piecewise constant environment, that is there exists an increasing sequence of times $(t_k)_{k\geq 0}$, with $t_k \to \infty$ as $k \to \infty$, such that

$$\widetilde{e}(t) = e_k, \text{ for } t \in [\widetilde{t}_{k-1}, \widetilde{t}_k).$$

With such choice of \tilde{e} , on the contrary to the previous models the environment is not subject to a gradual change but it experiences sudden changes. In the models in the previous sections, the frequency of the environment oscillations was much larger than the effect of the mutations (measured by ε). In those models the population adapted to an average environment since the population did not have the time to adapt to the environment changes ponctually. Here we consider a smaller frequency for the environment change which is comparable with the effect of the mutations. That is we choose

$$\widetilde{t}_k = \frac{t_k}{\varepsilon}$$

In this way the switch between the environment states occurs slowly such that the population has the time to follow the punctual environment state before the switch. This work is motivated by the study of the prey and predator interactions, where the prey has a much slower dynamics than the predator (for instance trees and insects) [Cos16]. However, the model studied in this section is a very simplified version of such problem.

In order to take into account the slow switch between the environment states and to study the effect of small mutations and slow change of the environment we make a change of variable in time, and define

$$n_{\varepsilon}(t,z) = \widetilde{n}_{\varepsilon}(\frac{t}{\varepsilon},z), \qquad \rho_{\varepsilon}(t) = \widetilde{\rho}_{\varepsilon}(\frac{t}{\varepsilon}), \qquad e(t) = \widetilde{e}(\varepsilon t).$$

The above rescaling leads to

$$\begin{cases} \varepsilon \frac{\partial}{\partial t} n_{\varepsilon} - \varepsilon^2 \Delta n_{\varepsilon} = n_{\varepsilon} R(e(t), z, \rho_{\varepsilon}), \\ \rho_{\varepsilon}(t) = \int_{\mathbb{R}^d} n_{\varepsilon}(t, y) \, dy, \\ n_{\varepsilon}(0, z) = n_{\varepsilon, 0}(z). \end{cases}$$
(4.47)

In the case of constant environment state e the asymptotic analysis of such equation is wellknown (see [PB08, BMP09, LMP11] and Chapter 1). However, all the known results suppose some initial conditions which guarantee that the initial population is well-adapted to the environment such that the population will not get extinct asymptotically at the initial time. The study of the case with piecewise constant environment requires however to understand the dynamics of the solution for general initial condition. In particular, we look for a criterion on the initial population which leads to extinction or survival of the population.

As explained above the asymptotic study of the problem reduces to the understanding of the asymptotic dynamics of the solutions considering a constant environment but general initial condition. In subsection 4.4.1 we study such problem and provide criteria on the initial condition leading to persistence or extinction of the population. In subsection 4.4.2 we deduce the asymptotic dynamics of the phenotypic density thanks to the results of subsection 4.4.1 and in the concavity framework of Section 1.4.

4.4.1 Persistence and extinction criterion on the initial condition

In this subsection, we provide persistence and extinction criteria for the selection-mutation model (4.47) in the case of constant environments but under general conditions. Once the survival of the population is guaranteed, then similar results as in [PB08, BMP09] on the asymptotic analysis of the equation apply. Note that in the case of constant environment, (4.47) is nothing but equation (1.3) in Chapter 1, that we recall below:

$$\begin{cases} \varepsilon \frac{\partial}{\partial t} n_{\varepsilon} - \varepsilon^2 \Delta n_{\varepsilon} = n_{\varepsilon} R(z, \rho_{\varepsilon}), \\ \rho_{\varepsilon}(t) = \int_{\mathbb{R}^d} n_{\varepsilon}(t, y) \, dy, \\ n_{\varepsilon}(0, z) = n_{\varepsilon, 0}(z). \end{cases}$$

Here, we use similar assumptions as in Section 1.2, however we relax Assumption (1.11) which is a sufficient condition on the initial population guaranteeing that it is well adapted and that the population does not get extinct.

Before providing our main result, we assume that the WKB transformation of the initial condition $u_{\varepsilon,0}$, given by

$$n_{\varepsilon,0} = \exp\left(\frac{u_{\varepsilon,0}}{\varepsilon}\right),$$

converges, as $\varepsilon \to 0$, to u_0 and we define

$$\Gamma_0 := \{ z \in \mathbb{R}^d, \, u_0(z) = 0 \}.$$
(4.48)

Theorem 4.14. [CEM] Assume (1.4)–(1.10).

i) If

Supp
$$n(0, \cdot) \cap \{z \in \mathbb{R}^d, R(z, 0) > 0\} \neq \emptyset$$

then the population persists, that is for every T > 0, there exists $\underline{I}(T) > 0$ and $\varepsilon(T) > 0$ such that

$$I_{\varepsilon}(t) \ge \underline{I}(T), \quad \forall t \in [0,T), \forall \varepsilon \le \varepsilon(T)$$

and $(I_{\varepsilon})_{\varepsilon}$ converges up to a subsequence a.e in $t \in \mathbb{R}^+$ towards a BV function I. In this case the results of Theorems 1.1 and 1.2 are valid.

ii) If

$$\exists C > 0 \text{ such that } \Gamma_0 \subseteq \{z \in \mathbb{R}^d, R(z, 0) \leq -C\}$$

then the population goes extinct asymptotically, that is there exists a finite and positive constant $0 < T_0 < +\infty$, such that $\lim_{\varepsilon \to 0} I_{\varepsilon}(t)|_{(0,T_0)} = 0$.

iii) If

$$\Gamma_0 \subseteq \left\{ z \in \mathbb{R}^d, \, R(z,0) \le 0 \right\},\,$$

then the population goes extinct asymptotically and punctually, that is $\forall \delta > 0, \exists \varepsilon_{\delta} > 0, \forall \varepsilon < \varepsilon_{\delta}, \exists t_{\varepsilon} \in [0,T]$, such that

$$I_{\varepsilon}(t_{\varepsilon}) < \delta.$$

Note that these conditions are based on the viability of the traits initially present in the population, and not on the growth rate of the individuals at initial time, since we look at $R(\cdot, 0)$ rather than at $R(\cdot, I_{\varepsilon}(0))$. A main difficulty in the proof of the persistence, is indeed to show that when the competition term I_{ε} drops below a threshold such that some viable traits have positive growth rates, such viable traits are still present in the population so that they can rescue the population from extinction.

Remark 4.15. Note that in the condition for asymptotic persistence in i) the set Supp $n(0, \cdot)$ is involved, which is a smaller set than Γ_0 , the set involved in the condition for asymptotic extinction in ii), iii). Therefore, one situation is not described, namely when

Supp $n(0, \cdot) \subseteq \{z \in \mathbb{R}^d, R(z, 0) \leq 0\}$ and $\Gamma_0 \cap \{z \in \mathbb{R}^d, R(z, 0) > 0\} \neq \emptyset$.

We illustrate this situation numerically in [CEM]. The numerical resolution of the problem indicates indeed that in this case the mass drops immediately as the population carrying traits with negative growth rates disappears, but it quickly increases again thanks to the viable traits in the set Γ_0 .

4.4.2 The dynamics of the population in a piecewise constant environment in the concave framework

In this section we go back to the case of piecewise constant environments. Thanks to Theorem 4.14 and similarly to Section 1.4, additional concavity assumptions on R and u_0 allow us to identify the phenotypic density as an evolving Dirac mass, as long as the population persists.

To identify the asymptotic dynamics of the phenotypic density, we recall that, thanks to Theorem 4.14-(i) and as long as the population persists, a Hamilton-Jacobi equation with constraint is derived from the Hopf-Cole transformation of the problem, that is

$$\begin{cases} \partial_t u = |\nabla u|^2 + R(e, z, I), & (t, z) \in \mathbb{R}^+ \times \mathbb{R}^d, \\ \max_{z \in \mathbb{R}^d} u(t, z) = 0, & t \in \mathbb{R}^+, \\ u(0, z) = u_0(z). \end{cases}$$

The dynamics of the phenotypic density is then identified by the following proposition.

Proposition 4.16. [CEM] Assume (1.4)-(1.10) and (1.14)-(1.15). (i) As long as the population persists, as $\varepsilon \to 0$, the population concentrates on a dominant trait which follows the gradient of the environment:

$$n_{\varepsilon}(t,z) \longrightarrow \rho(t)\,\delta(z-\overline{z}(t)), \quad \dot{\overline{z}} = (-D^2u)^{-1}(t,\overline{z}(t))\nabla R(e,\overline{z},I).$$

(ii) If the environment switches from e_k to e_{k+1} at time t_k . Then, the population goes extinct, asymptotically as $\varepsilon \to 0$, if

$$R(e_{k+1}, \overline{z}(t_k), 0) \le 0.$$

Otherwise, the population persists until the next switch.

Perspectives

I will formulate here some questions related to the works presented in this manuscript.

Hamilton-Jacobi equations with constraint and concentration of the phenotypic density

In chapter 1, we studied the following selection-mutation model:

$$\begin{cases} \varepsilon \partial_t n_{\varepsilon} - \varepsilon^2 \Delta n_{\varepsilon} = n_{\varepsilon} R(z, I_{\varepsilon}), & (t, z) \in \mathbb{R}^+ \times \mathbb{R}^d, \\ I_{\varepsilon}(t) = \int_{\mathbb{R}^d} n_{\varepsilon}(t, z) \psi(z) dz, & \\ n_{\varepsilon}(0, z) = n_{\varepsilon, 0}(z). \end{cases}$$
(5.1)

We showed that a Hopf-Cole transformation of the problem, that is,

$$n_{\varepsilon}(t,z) = \frac{1}{(2\pi\varepsilon)^d} \exp\left(\frac{u_{\varepsilon}(t,z)}{\varepsilon}\right),\tag{5.2}$$

leads to the following Hamilton-Jacobi equation with constraint [PB08, BMP09]:

$$\begin{cases} \partial_t u = |\nabla u|^2 + R(z, I), & (t, z) \in \mathbb{R}^+ \times \mathbb{R}^d, \\ \max_{z \in \mathbb{R}^d} u(t, z) = 0, & t \in \mathbb{R}^+, \\ u(0, z) = u_0(z), \end{cases}$$

with $I : \mathbb{R}^+ \to \mathbb{R}^+$ and $u \in C(\mathbb{R}^+ \times \mathbb{R}^d)$ a solution in the viscosity sense. Here, I may be thought of as a sort of regulator, or a sort of Lagrange multiplier, to maintain the maximum of u equal to 0. Such result would then allow to determine the limit of n_{ε} as $\varepsilon \to 0$. As $\varepsilon \to 0$, n_{ε} converges indeed in L^{∞} (w*(0, ∞); $\mathcal{M}^1(\mathbb{R}^d)$) to a measure n which has the following property

supp
$$n(t, z) \subset \{z | u(t, z) = 0\} \subset \{z | R(z, I(t)) = 0\},$$
 a.e. t.

We expect that the solution of (5.1), in the limit as $\varepsilon \to 0$, concentrates as a single Dirac mass corresponding to a dominant trait. In Chapter 1, we derived such concentration phenomenon under concavity assumptions on u_0 and R [LMP11]. Within this framework, we also provided uniqueness and regularity of the solution (u, I) to the Hamilton-Jacobi equation with constraint [MR16]. In a work in preparation with J.-M. Roquejoffre, we prove that this result leads to error estimates and asymptotic expansions for u_{ε} and I_{ε} in terms of ε [MR15b]:

$$u_{\varepsilon} = u + \varepsilon v + O(\varepsilon^2), \qquad I_{\varepsilon} = I + \varepsilon J + O(\varepsilon^2).$$

Such asymptotic expansion would then allow to obtain an approximation of the phenotypic density n_{ε} for ε small but nonzero. Such approximation is particularly interesting for the biological applications, since in many biological situations it is more relevant to consider small but nonvanishing effect of mutations. These approximations lead indeed to more quantitative results which allow to bring insight into current biological questions. In chapters 3 and 4 we provided several examples of applications of such approximations.

In a second current project, with J.-M. Roquejoffre and C. Quininao, we try to extend our result in [MR16] to the case where R and u_0 are only locally concave. This would allow us to obtain concentration of the solution on a single point in a more general range of situations. Note indeed that while the derivation of the Hamilton-Jacobi equation from the selection-mutation model above is well-known under general assumptions [PB08, BMP09], the identification of the phenotypic density n as a Dirac mass, which is equivalent with the identification of the maximum points of the viscosity solution u to the corresponding Hamilton-Jacobi equation, is only obtained in particular cases. Such concentration is indeed proved whether in 1-d and under monotony assumption on R with respect to z [PB08], or under concavity assumptions on R and u_0 [LMP11, MR16]. However, we expect that such concentration property would hold under general assumptions. This is indeed related to the principle of competitive exclusion [Lev70, Sch74] in evolutionary biology which states that when there are k limiting factors for the population, no more than k distinct traits may coexist. In this model there is only one limiting factor which is the resource taken into account via the competition term I.

We expect indeed that the current project with J.-M. Roquejoffre and C. Quininao would allow us to consider the cases where u has several maxima, which means that n is a sum of Dirac masses. Such polymorphic situations may arise in more complex models where for instance several integral terms are considered corresponding to several nutrients [DJMP05]. While the study of polymorphic populations and in particular the emergence of evolutionary branching (the creation of two Dirac masses from one) is an important question in evolutionary biology, there are not many results describing the dynamics of such polymorphic populations (see however [CJ11] where a Hamilton-Jacobi equation with constraint is derived in the case of several competition terms I_k and considering a particular form of the growth rate R). Note however that the study of polymorphic populations at equilibrium is an easier question (see for instance [Mir17] and chapter 3 where dimorphic populations are described in the case of spatially heterogeneous environments).

Fractional reaction-diffusion equations

In Chapter 2, we studied the following selection-mutation model, where the mutations are modeled with an integral kernel with thick tails:

$$\begin{cases} \varepsilon \partial_t n_{\varepsilon}(t,z) = \int_0^\infty \int_{\nu \in S^{d-1}} \left(n_{\varepsilon}(t,z+(e^{\varepsilon k}-1)\nu) - n_{\varepsilon}(t,z) \right) \frac{e^k}{|e^k-1|^{1+2\alpha}} dS dk + n_{\varepsilon}(t,z) R(z,\rho_{\varepsilon}(t)), \\ \rho_{\varepsilon}(t) = \int_{\mathbb{R}} n_{\varepsilon}(t,z) dz, \\ n_e(x,0) = n_{\varepsilon}^0(x). \end{cases}$$

We showed that a Hopf-Cole transformation, similar to (5.2), leads to the following Hamilton-Jacobi equation with constraint

$$\begin{cases} \partial_t u - \int_0^\infty \int_{\nu \in S^{d-1}} \left(e^{kD_z u \cdot \nu} - 1 \right) \frac{e^k dS dk}{|e^k - 1|^{1+2\alpha}} = R(z, I), \\ u(z, 0) = u^0(z). \end{cases}$$
(5.3)

We prove indeed that the function u_{ε} converges to a viscosity supersolution of (5.3) which is minimal in a certain class of supersolutions. Similarly to the case of classical laplacian, we also obtain that

$$\max_{z} u(t, z) = 0, \qquad \text{supp } n(t, z) \subset \{ z | u(t, z) = 0 \} \subset \{ z | R(z, I(t)) = 0 \}, \text{ a.e. } t.$$

Such property allows then to conclude, in the case where $z \in \mathbb{R}$ and R is monotonous with respect to z, that n is indeed a single Dirac mass for a.e. t. A question that I am interested in, is whether in the general multi-d case there exists a framework similar to the concave framework in Chapter 1 (see also [LMP11, MR16]) where we can obtain such concentration property for nand provide more precise description of the limit n. To this end, a natural first step is to study the properties of the Hamilton-Jacobi equation (5.3), where the Hamiltonian can take infinite values, and the corresponding minimal viscosity supersolution. Note however that since such minimal viscosity supersolution u has logarithmic decay according to Theorem 2.3, it cannot be strictly concave. Therefore, we do not expect that the theory in Chapter 1 would apply directly to such problem.

Another direction of study is to investigate whether the asymptotic analysis performed in Chapter 2 for the fractional Fisher-KPP equation and for the fractional selection-mutation model is related to a large deviation principle. Note that in the case of classical KPP equation, such asymptotic analysis, dealing with Hamilton-Jacobi equations, was first performed using stochastic tools and large deviation type estimates in the works of Friedlin [Fre85a, Fre85b, Fre86]. Similarly, an asymptotic analysis of the classical selection-mutation model using probabilistic tools has been provided recently [CH19] (see also [Fre87]). In the case of fractional reaction-diffusion equations, up to my knowledge, such stochastic analysis is not provided. I am interested in the understanding of the stochastic interpretation of our results and in investigating whether this is related to a large deviation principle for Lévy type operators.

In Section 2.4 in the framework of the PhD thesis of A. Léculier, a student that I supervise with J.-M. Roquejoffre, we presented a work on the following fractional Fisher-KPP type problem with obstacles

$$\begin{cases} \partial_t n + (-\Delta)^{\alpha} n = R(n), & (t, x) \in (0, \infty) \times \Omega, \\ n(t, x) = 0, & (t, x) \in [0, \infty) \times \Omega^c, \\ n(0, x) = n_0(x), & x \in \Omega, \end{cases}$$
(5.4)

with $\Omega \subset \mathbb{R}$ an open set. The function R(n) is a reaction term of KPP type : one can consider for instance R(n) = n(1-n). Here, x is the position in space and n corresponds to the population density as a function of time and position in space. Such study allows to investigate the impact of obstacles, as urban zones or rivers and seas, in the propagation of populations with nonlocal dispersion.

In a case where Ω is a periodic set of intervals of equal sizes, we prove that when such intervals are large enough, there exists a unique nonzero steady solution to (5.4) and that such steady state invades the unstable steady state 0 in the whole domain Ω with an exponential speed. Note that the uniqueness of the steady state is due to the nonlocal diffusion operator and it does not hold for the case with the classical diffusion $\alpha = 1$. Additionally, A. Léculier attempts to obtain a survival criterion, as a function of the domain or α , considering more general domains where the size of the obstacles are not fixed or considering for instance only two connected intervals in Ω (see [BCV16] for a related work). As a second step, we plan to study a model with trait dependence as follows

$$\begin{cases} \partial_t n + \sigma_1(-\Delta)_x^{\alpha} n - \sigma_2 \Delta_z n = n(R(z,x) - \rho(t,x)), & (t,x,z) \in (0,\infty) \times \Omega \times \mathbb{R}, \\ \rho(t,x) = \int_{\mathbb{R}} n(t,x,z) dz, & \\ n(t,x,z) = 0, & (t,x,z) \in [0,\infty) \times \Omega^c \times \mathbb{R}, \\ n(0,x,z) = n_0(x,z), & (x,z) \in \Omega \times \mathbb{R}. \end{cases}$$
(5.5)

Such analysis would allow to investigate the impact of nonlocal dispersion on the genetic adaptation of a population to its environment in presence of obstacles. The role of long distance dispersal is for instance important in the study of adaptation of forest trees to climate change [KRRA⁺12]. Note that models structured with trait and space with a local dispersion term are studied widely recently (see for instance [ACR13, BM15, Tur15, BJS16, BHR17, BMR] and [CM07] where such models were derived from stochastic individual based models and [ADP12] where the mathematical properties of such equations were first studied). In such models the reaction term has generally an integral term with respect to only one variable as above. Such problems do not admit comparison principle. Moreover, it is usually hard to obtain regularity estimates for such problems due to the nonlocal term only with respect to one variable. Note also that the non-local term may lead to different behaviors compared to an analogous local model (see for instance [CHM⁺] where we prove that the nonlocal competition may slow down the propagating front compared to an analogous model with local competition).

As a first step, we are interested in the asymptotic study of the equilibrium of (5.5) considering small effect of mutations ($\sigma_2 = \varepsilon^2$). Such analysis would be closely related to the work in Chapter 3, where several habitats are considered. However, here we consider a continuous space structure in each habitat. For instance in the case where Ω is a set of two distinct intervals, by an analogy to model (3.1) we expect that the equilibrium population would be monomorphic or dimorphic. A first step is to investigate the validity of such claim and to identify the conditions leading to polymorphic populations. Note that the analysis of models with continuous space structure in presence of obstacle, compared to the case of discrete space structured models, allows to consider more realistic models for instance to study the role of adaptation in response to climate change. The scaling $\sigma_2 = \varepsilon^2$ is also similar to the scaling used in articles [PS16, LL17]. However those articles consider a connected domain and local dispersion where the trait z models the dispersal rate and has no influence on the growth rate.

More complex reproduction mechanisms

A major problem in evolutionary biology is to take into account the sexual mechanisms of reproduction. Most of my previous works coming from selection-mutation models focused on the study of asexual reproduction, leading to nonlocal reaction-diffusion equations or systems. More recently, I have become interested in the study of sexual mechanisms which lead to some nonlinear integral terms adding significant difficulty to the previous analysis.

A first problem that I am interested in is the extension of the work presented in chapter 3 to the case of sexual reproduction:

$$\begin{cases} (-R_1(z) + \rho_{\varepsilon,1}) n_{\varepsilon,1}(z) = r_1 \mathcal{B}(n_{\varepsilon,1})(z) + m_2 n_{\varepsilon,2}(z) - m_1 n_{\varepsilon,1}(z), \\ (-R_2(z) + \rho_{\varepsilon,2}) n_{\varepsilon,2}(z) = r_2 \mathcal{B}(n_{\varepsilon,2})(z) + m_1 n_{\varepsilon,1}(z) - m_2 n_{\varepsilon,2}(z), \\ \rho_{\varepsilon,i} = \int n_{\varepsilon,i}(z) dz, \quad i = 1, 2. \end{cases}$$
(5.6)

Here, z corresponds to the phenotypic trait, and $n_{\varepsilon,i}$ is the phenotypic density in habitat *i*, for i = 1, 2. The term $R_i(z)$ is an intrinsic mortality rate that depends on the trait and on the habitat. We take into account a mortality term due to competition, through the total size of the population in each habitat $\rho_{\varepsilon,i}$. We also assume migration between patches, with migration rates m_i . The difference with the model in Chapter 3, that is (3.1), is in the term $r_i \mathcal{B}$ which corresponds to sexual reproduction. The rates of reproduction are denoted by r_i . New offsprings appear following Fisher's infinitesimal model [Fis19]. The trait of each offspring is the average of parental traits plus a random variable normally distributed with variance $\epsilon^2/2$. This translates into the following expression:

$$\mathcal{B}(n)(z) = \iint_{\mathbb{R}^2} G_{\epsilon} \left(z - \frac{z_1 + z_2}{2} \right) n(z_1) \left(\frac{n(z_2)}{\int_{\mathbb{R}} n(z_2') \, dz_2'} \right) \, dz_1 dz_2.$$

Note that Fisher's infinitesimal model is derived considering infinitely many genes each of which having a small contribution to the phenotype z [Fis19, BEV17]. The study of models with such sexual reproduction term has gained recent attention from the mathematical community. In [BCGL] the existence of equilibria to such equations without space heterogeneity was studied. In [CGP] an asymptotic analysis of such problem was provided using a similar scaling as in (5.6). In [Rao], another model with continuous space heterogeneity but with the same sexual reproduction term \mathcal{B} was studied with a different approach using the Wasserstein distance (see also [MR13]).

In the framework of the master's project and PhD thesis of L. Dekens, a student that I supervise with V. Calvez, we are interested in an asymptotic study of the selection-mutation-migration model (5.6) as $\varepsilon \to 0$. Note that in the case of the model with sexual reproduction (5.6) we expect to observe very different behaviors comparing to the results obtained for the asexual reproduction [Mir17]. In particular, it is known thanks to numerical simulations that in the case of sexual reproduction, non-symmetric steady solutions may exist even if we consider symmetric habitats [RK01].

The sexual reproduction term \mathcal{B} adds significant difficulty to the problem and even the case of one habitat is not entirely understood. In particular, even in the one habitat model, there may exist multiple steady solutions [CGP]. This is in contrast with the case of a sexual reproduction, where generally the Krein-Rutman Theorem guarantees the uniqueness of the solution. In the case of one habitat, it is shown heuristically in [BBC⁺] that a different rescaling, replacing ε by ε^2 in (5.2), allows to obtain a limiting problem indicating that the dominant term of the solution is a Gaussian. Then, again in the case of one habitat, in [CGP] it is proved using a perturbative analysis that there exist strong solutions of the selection-mutation model, for ε small, near the solutions of the limiting problem. Such limiting problem may have multiple solutions leading to multiple steady solutions of the selection-mutation model. Note however that this result does not imply that all the solutions of the selection-mutation model are well-approximated by the solutions of the limiting problem.

A second problem that I plan to study is a model with asexual reproduction but taking into account horizontal gene transfer:

$$\begin{cases} \varepsilon \partial_t n_\varepsilon - \varepsilon^2 \Delta n_\varepsilon = n_\varepsilon (R(z) - \rho_\varepsilon) + \tau n_\varepsilon \int_{\mathbb{R}} \frac{n_\varepsilon(t,y)}{\rho_\varepsilon(t)} \beta(z-y) dy, \\ \rho_\varepsilon(t) = \int_{\mathbb{R}} n_\varepsilon(t,x) dx. \end{cases}$$

This model is derived from stochastic individual based models $[BCF^{+}18]$ in the limit of large populations. The meanings of the terms in the equations are as above. The new integral term

corresponds to the horizontal gene transfer in form of conjugation, where genetic material is replicated and transferred from a cell to another. This form of genetic exchange is particularly important for instance in the case of plasmids which have a major role in the emergence of bacterial resistance to antibiotics [NHS09]. Horizontal gene transfer is not usually taken into account in the models in evolutionary biology. See however [HLFMW09, MR15a] where some integro-differential models including this phenomenon are studied.

I am interested in the study of the impact of this additional integral term on the asymptotic behavior of the population density n_{ε} , as $\varepsilon \to 0$. In [BCF+18, CFIH+19] numerical simulations reveal that surprising behaviors may arise due to the horizontal gene transfer compared to scenarios where one considers only selection and mutations (see also [CMT] for the theoretical analysis of a stochastic model with a finite number of traits). Depending on the rate or the form of the transfer, in addition to the expected scenarios, one can for instance observe periodic evolution of the dominant trait or evolutionary suicide, i.e. the process of adaptation leads to the extinction of the population. Note that this additional integral term leads to significant difficulties. In particular, it is not easy to identify the limit of n_{ε} , as $\varepsilon \to 0$, which may be concentrated on one or several points, which may go extinct for certain values of transfer rate τ or have a periodic behavior. Moreover, from a technical point of view it is not easy to obtain a priori regularity estimates.

Bibliography

- [ABF⁺19] L. Almeida, P. Bagnerini, G. Fabrini, B.D. Hughes, and T. Lorenzi. Evolution of cancer cell populations under cytotoxic therapy and treatment optimisation: insight from a phenotype-structured model. *ESAIM Math. Model. Numer. Anal.*, 53:1157–1190, 1 2019.
- [ABR17] M. Alfaro, H. Berestycki, and G. Raoul. The effect of climate shift on a species submitted to dispersion, evolution, growth, and nonlocal competition. SIAM J. Math. Anal., 49(1):562–596, 2017.
- [AC14] M. Alfaro and R. Carles. Explicit solutions for replicator-mutator equations: extinction vs. acceleration. *SIAM J. Appl. Math.*, 74:1919–1934, 2014.
- [ACR13] M. Alfaro, J. Coville, and G. Raoul. Traveling waves in a nonlocal equation as a model for a population structured by a space variable and a phenotypical trait. *Comm. Partial Differential Equations (CPDE)*, 38:2126–2154, 2013.
- [ADP12] A. Arnold, L. Desvillettes, and C. Prévost. Existence of nontrivial steady states for populations structured with respect to space and a continuous trait. Comm. on Pure and Applied Analysis, 11(1):83–96, 2012.
- [AW78] D. G. Aronson and H. F. Weinberger. Multidimensional nonlinear diffusion arising in population genetics. *Adv. in Math.*, 30(1):33–76, 1978.
- [B00] R. Bürger. The Mathematical theory of selection, recombination and mutation. Wiley, New-York, 2000.
- [BBC⁺] E. Bouin, T. Bourgeron, V. Calvez, O. Cotto, J. Garnier, T. Lepoutre, and O. Ronce. Equilibria of quantitative genetics models beyond the gaussian approximation i: Maladaptation to a changing environment. *In preparation.*
- [BC13] C. Brändle and E. Chasseigne. Large deviations estimates for some non-local equations. general bounds and applications. *Trans. Amer. Math. Soc.*, pages 3437– 3476, 2013.
- [BCF⁺18] S. Billiard, P. Collet, R. Ferrière, S. Méléard, and V.C. Tran. Stochastic dynamics for adaptation and evolution of microorganisms. In V. Mehrmann and M. Skutella, editors, *European Congress of Mathematics Berlin 2016*, pages 525—550. EMS Publishing House, 2018.
- [BCGL] T. Bourgeron, V. Calvez, J. Garnier, and T. Lepoutre. Existence of recombinationselection equilibria for sexual populations with changing environment. *Preprint*.

[BCM ⁺ 12]	E. Bouin, V. Calvez, N. Meunier, S. Mirrahimi, B. Perthame, G. Raoul, and R. Voi- turiez. Invasion fronts with variable motility: phenotype selection, spatial sorting and wave acceleration. <i>C. R. Math. Acad. Sci. Paris</i> , 350(15-16):761–766, 2012.
[BCV16]	H. Berestycki, J. Coville, and H. H. Vo. Entropy satisfying schemes for computing selection dynamics in competitive interactions. J. Math. Biol., 72:1693–1745, 2016.
[BDD14]	H. Berestycki, L. Desvillettes, and O. Diekmann. Can climate change lead to gap formation? <i>Ecological Complexity</i> , 20:264–270, 2014.
[BDNZ08]	H. Berestycki, O. Diekmann, C. J. Nagelkerke, and P. A. Zegeling. Can a species keep pace with a shifting climate? <i>Bulletin of Mathematical Biology</i> , 71(2):399, Dec 2008.
[BES90]	G. Barles, L. C. Evans, and P. E. Souganidis. Wavefront propagation for reaction- diffusion systems of PDE. <i>Duke Math. J.</i> , 61(3):835–858, 1990.
[BEV17]	N. H. Barton, A. M. Etheridge, and A. Veber. The infinitesimal model: definition, derivation and implications. <i>Theor. Pop. Biol.</i> , 118:50–73, 2017.
[BF18]	H. Berestycki and J. Fang. Forced waves of the Fisher-KPP equation in a shifting environment. <i>Journal of Differential Equations</i> , 264(3):2157–2183, 2018.
[BGHP18]	E., Bouin, J. Garnier, C. Henderson, and F. Patout. Thin front limit of an integro- differential fisher-KPP equation with fat-tailed kernels. <i>SIAM J. Math. Anal.</i> , 50(3):3365–3394, 2018.
[BHR17]	E. Bouin, C. Henderson, and L. Ryzhik. Superlinear spreading in local and nonlocal cane toads equations. <i>J. Math. Pures App.</i> , 108(5):724–750, 2017.
[BJS16]	H. Berestycki, T. Jin, and L. Silvestre. Propagation in a non local reaction diffusion equation with spatial and genetic trait structure. <i>Nonlinearity</i> , 29(4):1434–1466, mar 2016.
[BL95]	R. Burger and M. Lynch. Evolution and extinction in a changing environment: A quantitative-genetic analysis. <i>Evolution; international journal of organic evolution</i> , 49(1):151–163, 1 1995.
[BL16]	J. Bouhours and M. A. Lewis. Climate change and integrodifference equations in a stochastic environment. <i>Bulletin of Mathematical Biology</i> , 78(9):1866–1903, Sep 2016.
[BM15]	E. Bouin and S. Mirrahimi. A hamilton-jacobi limit for a model of population stuctured by space and trait. <i>Comm. Math. Sci.</i> , 13(6):1431–1452, 2015.
[BMP09]	G. Barles, S. Mirrahimi, and B. Perthame. Concentration in Lotka-Volterra parabolic or integral equations: a general convergence result. <i>Methods Appl. Anal.</i> , 16(3):321–340, 2009.
[BMR]	N. Berestycki, C. Mouhot, and G. Raoul. Existence of self-accelerating fronts for a non-local reaction-diffusion equation. <i>Preprint</i> .
[BNPR09]	H. Berestycki, G. Nadin, B. Perthame, and L. Ryzhik. The non-local Fisher-KPP equation : traveling waves and steady states. <i>Nonlinearity</i> , 22:2813–2844, 2009.

[BP88]	G. Barles and B. Perthame. Exit time problems in optimal control and vanishing viscosity method. <i>SIAM J. Control Optim.</i> , 26(5):1133–1148, 1988.
[BP97]	B. Bolker and W. Pacala. Using moment equations to understand stochastically driven spatial pattern formation in ecological systems. <i>Theoretical Population Biology</i> , 52:179–197, 1997.
[BP07]	G. Barles and B. Perthame. Concentrations and constrained Hamilton-Jacobi equations arising in adaptive dynamics. <i>Contemp. Math.</i> , 439:57–68, 2007.
[BPnMGI17]	R. E. Beardmore, R. Peña Miller, F. Gori, and J. Iredell. Antibiotic cycling and antibiotic mixing: Which one best mitigates antibiotic resistance? <i>Mol Biol Evol.</i> , 34(4):802–817, 2017.
[BR08]	H. Berestycki and L. Rossi. Reaction-diffusion equations for population dynamics with forced speed i - the case of the whole space, disc. <i>Discrete Contin. Dyn. Syst.</i> , 21:41–67, 05 2008.
[BR09]	H. Berestycki and L. Rossi. Reaction-diffusion equations for population dynamics with forced speed ii - cylindrical-type domains. <i>Discrete and Continuous Dynamical Systems</i> - A , 25(1):19–61, 2009.
[BRR11]	H. Berestycki, JM. Roquejoffre, and L. Rossi. The periodic patch model for population dynamics with fractional diffusion. <i>Discrete Contin. Dyn. Syst. Ser. S</i> , pages 1–13, 2011.
[CC04]	A. Calsina and S. Cuadrado. Small mutation rate and evolutionarily stable strategies in infinite dimensional adaptive dynamics. J. Math. Biol., $48(2)$:135–159, 2004.
[CCDR13]	A. Calsina, S. Cuadrado, L. Desvillettes, and G. Raoul. Asymptotics of steady states of a selection mutation equation for small mutation rate. <i>Proceedings of the Royal Society of Edinburgh</i> , 143(06):1123–1146, 2013.
[CCR12]	X. Cabré, AC. Coulon, and JM. Roquejoffre. Propagation in Fisher-KPP type equations with fractional diffusion in periodic media. <i>C. R. Math. Acad. Sci. Paris</i> , 350:885–890, 2012.
[CEM]	M. Costa, C. Etchegaray, and S. Mirrahimi. Survival criterion for a population subject to selection and mutations; application to temporally piecewise constant environments. <i>Preprint</i> .
[CFIH ⁺ 19]	V. Calvez, S. Figueroa Iglesias, H. Hivert, S. Méléard, A. Melnykova, and S. Nord- mann. Horizontal gene transfer: numerical comparison between stochastic and deterministic approaches. In <i>ESAIM: Proceedings and Surveys</i> , 2019.
[CFM06]	N. Champagnat, R. Ferrière, and S. Méléard. Unifying evolutionary dynamics: From individual stochastic processes to macroscopic models. <i>Th. Pop. Biol.</i> , 69(3):297–321, 2006.
[CFM08]	N. Champagnat, R. Ferrière, and S. Méléard. <i>Individual-based probabilistic models of adaptive evolution and various scaling approximations</i> , volume 59 of <i>Progress in Probability</i> , pages 75–114. Birkhäuser, 2008.

[CGP] V. Calvez, J. Garnier, and F. Patout. A quantitative genetics model with sexual mode of reproduction in the regime of small variance. *Preprint arXiv:1811.01779*.

[CGP07]	J. Clairambault, S. Gaubert, and B. Perthame. An inequality for the Perron and Floquet eigenvalues of monotone differential systems and age structured equations. C. R. Math. Acad. Sci. Paris, 345(10):549–555, 2007.
[CH19]	N. Champagnat and B. Henri. A probabilistic approach to Dirac concentration in nonlocal models of adaptation with several resources. <i>Ann. Appl. Probab.</i> , 29(4):2175–2216, 2019.
$[\mathrm{CHM^+}]$	V. Calvez, C. Henderson, S. Mirrahimi, O. Turanova, and T. Dumont. Non-local competition slows down front acceleration during dispersal evolution, submitted. <i>Preprint: arXiv:1810.07634</i> .
[CJ11]	N. Champagnat and PE. Jabin. The evolutionary limit for models of populations interacting competitively via several resources. <i>Journal of Differential Equations</i> , 261:179–195, 2011.
[CJL15]	W. Cai, P.E. Jabin, and H. Liu. Time-asymptotic convergence rates towards the discrete evolutionary stable distribution. <i>Math. Methods Appl. Sci</i> , 25:1589–1616, 2015.
[CJM14]	N. Champagnat, PE. Jabin, and S. Méléard. Adaptive dynamics in a stochastic multi-resources chemostat model. J. Math. Pures App., 101(6):755–788, 2014.
[CJR10]	N. Champagnat, PE. Jabin, and G. Raoul. Convergence to equilibrium in competitive lotka-volterra equations. <i>Comptes Rendus de l'Academie des Sciences</i> , 348(23–24):1267–1272, 2010.
[CL]	V. Calvez and KY. Lam. Uniqueness of the viscosity solution of a constrained hamilton-jacobi equation. <i>Preprint arXiv:1809.05317.</i>
[CL07]	N. Champagnat and A. Lambert. Evolution of discrete populations and the canonical diffusion of adaptive dynamics. <i>Ann. Appl. Probab.</i> , 17:102–155, 2007.
[CLM10]	LM. Chevin, R. Lande, and G. M. Mace. Adaptation, plasticity, and extinction in a changing environment: Towards a predictive theory. <i>PLOS Biology</i> , 8(4):1–8, 2010.
[CM07]	N. Champagnat and S. Méléard. Invasion and adaptive evolution for individual- based spatially structured populations. <i>J. Math. Biol.</i> , 55:147–188, 2007.
[CMT]	N. Champagnat, S. Méléard, and V.C. Tran. Stochastic analysis of emergence of evolutionary cyclic behavior in population dynamics with transfer. <i>Preprint</i> .
[Con01]	G. Contreras. Action potential and weak kam solutions. Calc. Var. Partial Differential Equations, 13(4):427–458, 2001.
[Cos16]	M. Costa. A piecewise deterministic model for a prey-predator community. Ann. Appl. Probab., 26(6):3491–3530, 2016.
[CR12]	A.C. Coulon and J.M. Roquejoffre. Transition between linear and exponential propagation in Fisher-KPP type reaction-diffusion equations. <i>Comm. Partial Dif</i> ferential Equations, 37:2029–2049, 2012.
[CR13]	X. Cabré and JM. Roquejoffre. The influence of fractional diffusion on Fisher-KPP equations. <i>Comm. Math. Phys</i> , 320:679–722, 2013.

[CVT15]	L-M. Chevin, M.E. Visser, and J. Tufto. Estimating the variation, autocorrelation, and environmental sensitivity of phenotypic selection. <i>Evolution</i> , 69(9):2319–2332, 2015.
[Day00]	T. Day. Competition and the effect of spatial resource heterogeneity on evolutionary diversification. <i>The American Naturalist</i> , 155(6):790–803, 2000.
[DD03]	M. Doebeli and U. Dieckmann. Speciation along environmental gradients. <i>Nature</i> , 421:259–264, 01 2003.
[Die04]	O. Diekmann. A beginner's guide to adaptive dynamics. In <i>Mathematical modelling</i> of population dynamics, volume 63 of Banach Center Publ., pages 47–86. Polish Acad. Sci., Warsaw, 2004.
[DJMP05]	O. Diekmann, PE. Jabin, S. Mischler, and B. Perthame. The dynamics of adaptation: an illuminating example and a Hamilton-Jacobi approach. <i>Th. Pop. Biol.</i> , 67(4):257–271, 2005.
[DJMR08]	L. Desvillettes, PE. Jabin, S. Mischler, and G. Raoul. On mutation-selection dynamics for continuous structured populations. <i>Commun. Math. Sci.</i> , 6(3):729–747, 2008.
[DL96]	U. Dieckmann and R. Law. The dynamical theory of coevolution: A derivation from stochastic ecological processes. J. Math. Biol., 34:579–612, 1996.
[DL00]	U. Dieckmann and R. Law. Relaxation projections and the method of moments. In <i>The Geometry of Ecological Interactions: Symplifying Spatial Complexity</i> , pages 412–455. Cambridge University Press, 2000.
[DRG13]	F. Débarre, O. Ronce, and S. Gandon. Quantifying the effects of migration and mutation on adaptation and demography in spatially heterogeneous environments. <i>Journal of Evolutionary Biology</i> , 26:1185–1202, 2013.
[DYG15]	F. Débarre, S. Yeaman, and F. Guillaume. Evolution of quantitative traits under a migration-selection balance: when does skew matter? <i>The American Naturalist</i> , 186(37–47), 2015.
[Eng10]	H. Engler. On the speed of spread for fractional reaction-diffusion equations. <i>Int. J. Differ. Equ.</i> , pages Article ID 315421, 16 p., 2010.
[ES89]	L. C. Evans and P. E. Souganidis. A PDE approach to geometric optics for certain semilinear parabolic equations. <i>Indiana Univ. Math. J.</i> , 38(1):141–172, 1989.
[Esh83]	I. Eshel. Evolutionary and continuous stability. <i>Journal of Theoretical Biology</i> , 103(1):99–111, 1983.
[Eva89]	L.C. Evans. The perturbed test function method for viscosity solutions of nonlinear PDE. <i>Proc. R. Soc. Edinb. Sec. A</i> , 111:359–375, 1989.
[Eva92]	L.C. Evans. Periodic homogenisation of certain fully nonlinear partial differential equations. <i>Proc. R. Soc. Edinb. Sec. A</i> , 120(3-4):245–265, 1992.
[Fat16]	A. Fathi. Weak Kam Theorem in Lagrangian Dynamics. Number 88 in Cambridge Studies in Advanced Mathematics. Cambridge Univ Pr, 2016.

[FH15]	G. Faye and M. Holzer. Modulated traveling fronts for a nonlocal Fisher-KPP equation: a dynamical system approach. <i>J. of Differential Equations</i> , 258(7):2257–2289, 2015.
[FI19]	S. Figueroa Iglesias. Integro-differential models for evolutionary dynamics of pop- ulations in time-heterogeneous environments. PhD thesis, University of Paul Sabatier, 2019.
[FIM]	S. Figueroa Iglesias and S. Mirrahimi. Selection and mutation in a shifting and fluctuating environment. <i>Preprint</i> .
[FIM18]	S. Figueroa Iglesias and S. Mirrahimi. Long time evolutionary dynamics of phe- notypically structured populations in fluctuating environments. <i>SIAM J. Math.</i> <i>Anal.</i> , 50(5):5537–5568, 2018.
[Fis19]	R. A. Fisher. Xvthe correlation between relatives on the supposition of mendelian inheritance. <i>Earth and Environmental Science Transactions of the Royal Society of Edinburgh</i> , 52(2):399–433, 1919.
[Fis37]	R. A. Fisher. The wave of advance of advantageous genes. Annals of Eugenics, 7(4):355–369, 1937.
$[\mathrm{FMP}^+]$	C. Fabre, S. Méléard, E. Porcher, C. Teplitsky, and Robert A. Evolution of a structured population in a heterogeneous environment. <i>Preprint</i> .
[Fre85a]	M. Freidlin. Functional integration and partial differential equations, volume 109 of Annals of Mathematics Studies. Princeton University Press, Princeton, NJ, 1985.
[Fre85b]	M. Freidlin. Limit theorems for large deviations and reaction-diffusion equations. <i>The Annals of Probability</i> , 13(3):639–675, 1985.
[Fre86]	M. Freidlin. Geometric optics approach to reaction-diffusion equations. SIAM J. Appl. Math., 46:222–232, 1986.
[Fre87]	M. Freidlin. Tunneling soliton in the equations of reaction-diffusion type. Ann. New York Acad. Sci., 491(1):149–156, 1987.
[FW98]	M. I. Freidlin and A. D. Wentzell. Random Perturbations of Dynamical Systems. Springer, 1998.
[Gar11]	J. Garnier. Accelerating solutions in integro-differential equations. SIAM Journal on Mathematical Analysis, 43(4):1955–1974, 2011.
[GF79]	J. Gärtner and M. Freidlin. The propagation of concentration waves in periodic and random media. <i>Dokl. Akad. Nauk SSSR</i> , 249(3):521–525, 1979.
[GH09]	R. Gomulkiewicz and D. Houle. Demographic and genetic constraints on evolution. <i>The American Naturalist</i> , 174(6):E218–E229, 2009.
[GH12]	JS. Guo and F. Hamel. Dokl. akad. nauk sssr. Archive for Rational Mechanics and Analysis, 204:945–975, 2012.
[GHGR17]	M-E Gil, F. Hamel, Martin G., and L. Roques. Mathematical properties of a class of integro-differential models from population genetics. <i>SIAM J. Appl. Math.</i> , 2017.

- S. A. H. Geritz, E. Kisdi, G. Mészena, and J. A. J. Metz. Evolutionarily singular [GKMM98] strategies and the adaptive growth and branching of the evolutionary tree. Evol. Ecol, 12:35-57, 1998. [GM17] S. Gandon and S. Mirrahimi. A Hamilton-Jacobi method to describe the evolutionary equilibria in heterogeneous environments and with non-vanishing effects of mutations. Comptes rendus-Mathématiques, 355(2):155–160, 2017. [GS16] A. Greco and R. Servadei. Hopf's lemma and constrained radial symmetry for the fractional laplacian. Math. Res. Lett., 23(3):863-885, 2016. [GVA06] S. Génieys, V. Volpert, and P. Auger. Pattern and waves for a model in population dynamics with nonlocal consumption of resources. Math. Model. Nat. Phenom., 1(1):63-80, 2006.[HDT01] A. Hendry, T. Day, and E. B. Taylor. Population mixing and the adaptive divergence of quantitative traits in discrete populations: a theoretical framework for empirical tests. *Evolution*, 55(3):459–466, 2001. [Hel04] B. Helffer. Introduction to semi-classical methods for the Schrödinger operator with magnetic field. Séminaires et Congrés, 05 2004. [Hes91] P. Hess. Periodic-parabolic boundary value problems and positivity. Longman Scientific & Technical, Harlow, Essex, UK, 1991. [HLFMW09] P. Hinow, F. Le Foll, P. Magal, and G. F. Webb. Analysis of a model for transfer phenomena in biological populations. SIAM J. Appl. Math., 70:40-62, 2009. [HP08] J. Húska and P. Polácik. Exponential separation and principal Floquet bundles for linear parabolic equations on \mathbb{R}^N . Discrete Contin. Dynam. Sust. - A, 20(1):81-113, 2008. [HR14] F. Hamel and L. Ryzhik. On the nonlocal fisher-kpp equation: steady states, spreading speed and global bounds. Nonlinearity, 27:2735–2753, 2014. [HS88] J. Hofbauer and K. Sigmund. The theory of evolution and dynamical systems, volume 7 of London Mathematical Society Student Texts. Cambridge University Press, Cambridge, 1988. Mathematical aspects of selection, Translated from the German. B. Jourdain, S. Méléard, and W. A. Woyczynski. Lévy flights in evolutionary [JMW12] ecology. Journal of Mathematical Biology, 65(4):677-707, 2012. P.-E. Jabin and G. Raoul. On selection dynamics for competitive interactions. J. [JR11] Math. Biol., 63(3):493–517, 2011. [Kim65] M. Kimura. A stochastic model concerning the maintenance of genetic variability in quantitative characters. Proc. Natl. Acad. Sci. USA, 54:731-736, 1965.
- [KM14] M. Kopp and S. Matuszewski. Rapid evolution of quantitative traits: theoretical perspectives. *Evolutionary Applications*, 7(1):169–191, 2014.
- [KMZ⁺13] T. Ketola, L. Mikonranta, J. Zhang, K. Saarinen, A. M. Ormala, V. P. Friman, J. Mappes, and J. Laakso. Fluctuating temperature leads to evolution of thermal generalism and preadaptation to novel environments. *Evolution*, 67(10):2936–2944, 2013.

[KPP37]	A. N. Kolmogorov, I. G. Petrovskii, and N. S. Piskunov. Etude de l'équation de diffusion avec accroissement de la quantité de matière, et son application à un problème biologique. <i>Bjul. Moskowskogo Gos. Univ., Série Internationale, Section</i> A, 1:1–26, 1937.
[KRRA ⁺ 12]	A. Kremer, O. Ronce, J. J. Robledo-Arnuncio, F. Guillaume, G. Bohrer, R. Nathan, J. R. Bridle, R. Gomulkiewicz, E. K. Klein, K. Ritland, A. Kuparinen, S. Gerber, and S. Schueler. Long-distance gene flow and adaptation of forest trees to rapid climate change. <i>Ecol Lett.</i> , 15(4):378–92, 2012.
[LÍ9]	A. Léculier. A singular limit in a fractional reaction-diffusion equation with periodic coefficients. Commun. Math. Sci., 17(2):565–586, 2019.
[LA83]	R. Lande and S. Arnold. The measurement of selection on correlated characters. $Evolution,37{:}1210{-}1226,1983.$
[Lan75]	R. Lande. The maintenance of genetic variability by mutation in a polygenic character with linked loci. Genetical Research, $26(3):221-235$, 1975.
[Lan79]	R. Lande. Quantitative genetic analysis of multivariate evolution applied to brain:body size allometry. $Evolution,\ 33:402-416,\ 1979.$
[LCC16]	T. Lorenzi, R. H. Chisholm, and J. Clairambault. Tracking the evolution of cancer cell populations through the mathematical lens of phenotype-structured equations. <i>Biology Direct</i> , 11(1):1–17, 2016.
[LCDH15]	T. Lorenzi, R.H. Chisholmm, L. Desvillettes, and B.D. Hughes. Dissecting the dynamics of epigenetic changes in phenotype-structured populations exposed to fluctuating environments. <i>Journal of Theoretical Biology</i> , 386:166–176, 2015.
[LCS15]	H. Liu, W. Cai, and N. Su. Entropy satisfying schemes for computing selection dynamics in competitive interactions. <i>SIAM Journal on Numerical Analysis</i> , 53(3):1393–1417, 2015.
[LDD08]	O. Leimar, M. Doebeli, and U. Dieckmann. Evolution of phenotypic clusters through competition and local adaptation along an environmental gradient. <i>Evolution</i> , 62(4):807–822, 2008.
[Lev70]	S. A. Levin. Community equilibria and stability, and an extension of the competitive exclusion principle. <i>The American Naturalist</i> , 104:413–423, 1970.
[Lio82a]	P. L. Lions. Generalized solutions of Hamilton-Jacobi equations, volume 69 of Research Notes in Mathematics. Pitman Advanced Publishing Program, Boston, 1982.
[Lio82b]	P. L. Lions. Generalized solutions of Hamilton-Jacobi equations, volume 69 of Research Notes in Mathematics. Pitman Advanced Publishing Program, Boston, 1982.
[LL17]	KY. Lam and Y. Lou. An integro-pde model for evolution of random dispersal. J. Funct. Anal., 272:1755–1790, 2017.
[LMM15]	H. Leman, S. Méléard, and S. Mirrahimi. Influence of a spatial structure on the long time behavior of a competitive Lotka-Volterra type system. <i>Discrete Contin. Dyn. Syst. Ser. B</i> , 20(2):469–493, 2015.

- [LMP11] A. Lorz, S. Mirrahimi, and B. Perthame. Dirac mass dynamics in multidimensional nonlocal parabolic equations. Comm. Partial Differential Equations, 36(6):1071-1098, 2011. [LMR] A. Léculier, S. Mirrahimi, and J.-M. Roquejoffre. Propagation in a fractional reaction-diffusion equation in a periodically hostile environment. Preprint. [LMR10] S. Lavergne, W. Mouquet, N.and Thuiller, and O. Ronce. Biodiversity and climate change: Integrating evolutionary and ecological responses of species and communities. Annual Review of Ecology, Evolution, and Systematics, 41(1):321–350, 2010. [LS96] R. Lande and S. Shannon. The role of genetic variation in adaptation and population persistence in a changing environment. Evolution, 50(1):434–437, 1996. J. L. Lush. Animal Breeding Plans. Ames, Iowa, 1937. [Lus37] [MBPS12] S. Mirrahimi, G. Barles, B. Perthame, and P. E. Souganidis. Singular Hamilton-Jacobi equation for the tail problem. SIAM J. Math. Anal., 44(6):4297–4319, 2012. [MCG97] G. Meszéna, I. Czibula, and S. Geritz. Adaptive dynamics in a 2-patch environment: a toy model for allopatric and parapatric speciation. Journal of Biological Systems, 5(02):265–284, 1997. [MG] S. Mirrahimi and S. Gandon. Evolution of specialization in heterogeneous environments: equilibrium between selection, mutation and migration. Preprint: bioRxiv : https://doi.org/10.1101/353458. [MHM⁺96] J. A. J. Metz, Geritz S. A. H., G. Meszéna, Jacobs F. J. A., and J.S. Van Heerwaarden. Adaptive dynamics, a geometrical study of the consequences of nearly faithful reproduction. In S. J. Van Strien and S. M. Verduyn Lunel, editors, Stochastic and spatial structures of dynamical systems., pages 183–231. North Holland, Elsevier, 1996. [Mir] S. Mirrahimi. Singular limits for models of selection and mutations with heavytailed mutation distribution. To appear in J. Math. Pures App. [Mir11] S. Mirrahimi. Concentration phenomena in PDEs from biology. PhD thesis, Univeristy of Pierre et Marie Curie (Paris 6), 2011. [Mir13] S. Mirrahimi. Migration and adaptation of a population between patches. Discrete and Continuous Dynamical Systems - Series B (DCDS-B), 18(3):753-768, 2013. [Mir17] S. Mirrahimi. A Hamilton-Jacobi approach to characterize the evolutionary equilibria in heterogeneous environments. Mathematical Models and Methods in Applied Sciences, 27, No. (2017)(13):2425–2460, 2017. [MM15] S. Méléard and S. Mirrahimi. Singular limits for reaction-diffusion equations with fractional Laplacian and local or nonlocal nonlinearity. Comm. Partial Differential Equations, 40(5):957-993, 2015.
- [MP15] S. Mirrahimi and B. Perthame. Asymptotic analysis of a selection model with space. J. Math. Pures App., 104(1108–1118), 2015.

- [MPBM11] S. Mirrahimi, B. Perthame, E. Bouin, and P. Millien. Population formulation of adaptative meso-evolution; theory and numerics, pages 159–174. Mathematics and Biosciences in Interaction. Birkhäuser Basel, 2011. [MPS15] S. Mirrahimi, B. Perthame, and P. E. Souganidis. Time fluctuations in a population model of adaptive dynamics. Annales de l'Institut Henri Poincare (C) Analyse Non Linéaire, 32(1):41–58, 2015. [MPW12] S. Mirrahimi, B. Perthame, and J. Y. Wakano. Evolution of species trait through resource competition. Journal of mathematical biology, 64(7):1189-1223, 2012. [MPW14] S. Mirrahimi, B. Perthame, and J. Y. Wakano. Direct competition results from strong competition for limited resource. Journal of mathematical biology, 68(4):931-949, 2014.[MR13] S. Mirrahimi and G. Raoul. Population structured by a space variable and a phenotypical trait. Theoretical Population Biology, 84:87–103, 2013. [MR15a] P. Magal and G. Raoul. Dynamics of a kinetic model describing protein exchanges in a cell population. ArXiv:1511.02665, 2015. [MR15b] S. Mirrahimi and J.-M. Roquejoffre. Uniqueness in a class of Hamilton-Jacobi equations with constraints. C. R. Math. Acad. Sci. Paris, 353:489-494, 2015. [MR16] S. Mirrahimi and J.-M. Roquejoffre. A class of Hamilton-Jacobi equations with constraint: Uniqueness and constructive approach. Journal of Differential Equations, 260(5):4717-4738, 2016. [MS74] J. Maynard Smith. The theory of games and the evolution of animal conflicts. J. theor. Biol., 47:209–221, 1974. [MS13] S. Mirrahimi and P.E. Souganidis. A homogenization approach for the motion of motor proteins. Nonlinear Differential Equations and Applications (NoDEA), 20(1):129-147, 2013.[MSP73] J. Maynard Smith and G. R. Price. The logic of animal conflict. Nature, 246:15–18, 1973. [MW00] P. Magal and G. F. Webb. Mutation, selection and recombination in a model of phenotype evolution. Discrete Contin. Dynam. Syst., 6:221-236, 2000. [NHS09] A. Norman, L.H. Hansen, and S.J. Sørensen. Conjugative plasmids : vessels of the communal gene pool. Philos. Trans. R. Soc. Lond. B. Biol. Sci., 384:2275-2289, 2009. [NRRP13] G. Nadin, L. Rossi, L. Ryzhik, and B. Perthame. Wave-like solutions for nonlocal reaction-diffusion equations: a toy model. Mathematical Modelling of Natural Phenomena, 08:33-41, 2013.
- [NTP10] G. Nadin, M. Tang, and B. Perthame. Can a traveling wave connect two unstable states? the case of the nonlocal fisher equation. C.R. Acad. Sci. Paris, 349:553–557, 2010.
- [PB08] B. Perthame and G. Barles. Dirac concentrations in Lotka-Volterra parabolic PDEs. Indiana Univ. Math. J., 57(7):3275–3301, 2008.

[PMLJ ⁺ 13]	R. Pena-Miller, D. Laehnemann, G. Jansen, A. Fuentes-Hernandez, P. Rosenstiel, H. Schulenburg, and R. Beardmore. When the most potent combination of antibiotics selects for the greatest bacterial load: The smile-frown transition. <i>PLOS Biology</i> , 11(4):1–13, 2013.
[PS16]	B. Perthame and P. E. Souganidis. Rare mutations limit of a steady state dispersal evolution model. <i>Math. Model. Nat. Phenom.</i> , 11:154–166, 2016.
[Rao]	G. Raoul. Macroscopic limit from a structured population model to the kirk patrick-barton model. $\ensuremath{\textit{Preprint}}$
[Rao11]	G. Raoul. Long time evolution of populations under selection and vanishing mutations. <i>Acta Applicandae Mathematica</i> , 114:1–14, 2011.
[Ric04]	S. H. Rice. <i>Evolutionary theory: mathematical and conceptual foundations</i> . Sinauer Associates, Inc., 2004.
[RK01]	O. Ronce and M. Kirkpatrick. When sources become sinks: migration meltdown in heterogeneous habitats. $Evolution,55(8):\!1520\!-\!1531,2001.$
[RT17]	J. M. Roquejoffre and A. Tarfulea. Gradient estimates and symmetrization for Fisher-KPP front propagation with fractional diffusion. <i>Journal de Mathématiques Pures et Appliquées</i> , 108(4):399–424, 2017.
[Sat99]	K. Sato. <i>Lévy processes and Infinitely Divisible Distributions</i> . Number 68 in Cambridge Studies in Advanced Math. Cambridge University Press, 1999.
[Sch74]	T. W. Schoener. Resource partitioning in ecological communities. <i>Science</i> , 13:27–39, 1974.
[Sla78]	M. Slatkin. Spatial patterns in the distributions of polygenic characters. Journal of Theoretical Biology, $70(2)$:213 – 228, 1978.
[SLM13]	O. Savolainen, M. Lascoux, and J. Merilä. Ecological genomics of local adaptation. <i>Nature Reviews Genetics</i> , 14:807–820, 10 2013.
[ST19]	P. E. Souganidis and A. Tarfulea. Front propagation for non-local KPP reaction diffusion equations in periodic media. <i>Nonlinear Differential Equations and Applications</i> , 26:29, 2019.
[Tur15]	O. Turanova. On a model of a population with variable motility. Mathematical Models and Methods in Applied Sciences, $25(10)$:1961–2014, 2015.
[Whi15]	M. C. Whitlock. Modern approaches to local adaptation. <i>The American Naturalist</i> , 186(S1):S1–S4, 2015. PMID: 26098334.
[YG09]	S. Yeaman and F. Guillaume. Predicting adaptation under migration load: the role of genetic skew. $Evolution, 63(11):2926-2938, 2009.$